UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM S-8 Registration Statement

under
The Securities Act of 1933

TPG Inc.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation or organization)

301 Commerce Street, Suite 3300 Fort Worth, TX 76102 (Address of principal executive offices, including zip code)

87-2063362 (I.R.S employer identification no.)

TPG INC. OMNIBUS EQUITY INCENTIVE PLAN (Full title of the plan)

Bradford Berenson, Esq. General Counsel 345 California Street, Suite 3300 San Francisco, CA 94104 (Name and address of agent for service)

(415) 743-1500 (Telephone number, including area code, of agent for service)

With a copy to:

John J. Cannon III, Esq. Gillian Emmett Moldowan, Esq. Shearman & Sterling LLP 599 Lexington Avenue New York, NY 10022 (212) 848-4000

| ndicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company or an |
|---|
| merging growth company. See definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth |
| ompany" in Rule 12b-2 of the Exchange Act. |

| Large accelerated filer | | Accelerated filer | |
|-------------------------|--|-----------------------------------|---|
| Non-accelerated filer | \boxtimes | Smaller reporting company | |
| | | Emerging growth company | |
| 5 5 5 | npany, indicate by check mark if the registrant has elected not to use the extended transit counting standards provided pursuant to Section 7(a)(2)(B) of the Securities Act. \Box | ion period for complying with any | 7 |

CALCULATION OF REGISTRATION FEE

| Title of Securities to be Registered | Amount to be Registered | Proposed Maximum Offering Price Per Share | Proposed Maximum Aggregate Offering Price | Amount of Registration Fee |
|---|-------------------------------|--|--|-------------------------------|
| Class A common stock, \$0.001 par value per share | 30,694,780 (1) | \$29.50(2) | 905,496,010 (2) | \$83,940 (2) |

⁽¹⁾ Pursuant to Rule 416 of the Securities Act of 1933, as amended (the "Securities Act"), this Registration Statement on Form S-8 (this "Registration Statement") shall also cover any additional shares of Class A common stock, par value \$0.001 per share (the "Common Stock"), of TPG Inc. (the "Registrant") that become issuable under the TPG Inc. Omnibus Equity Incentive Plan, as amended (the "Plan"), by reason of any stock dividend, stock split, recapitalization or any other similar transaction that results in an increase in the number of the Registrant's outstanding shares of Common Stock.

|) | Estimated solely for the purpose of calculating the registration fee in accordance with Rule 457(h) under the Securities Act based on the initial public offering price of Class A common stock of \$29.50 per share as set forth on the cover page of the Registrant's Prospectus to be filed on or about January 14, 2022 pursuant to Rule 424(b) under the Securities Act, relating to the Registration Statement on Form S-1, as amended (File No. 333-261681). |
|---|---|
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |

PART I

INFORMATION REQUIRED IN THE SECTION 10(a) PROSPECTUS

The information required by Item 1 and Item 2 of Part I of Form S-8 is omitted from this filing in accordance with Rule 428 under the Securities Act and the introductory note to Part I of Form S-8. The documents containing the information specified in Part I will be delivered to the participants in the Plan covered by this Registration Statement as required by Rule 428(b)(1) under the Securities Act.

PART II

INFORMATION REQUIRED IN THE REGISTRATION STATEMENT

Item 3. Incorporation of Documents by Reference.

The following documents filed by the Registrant with the Securities and Exchange Commission (the "SEC") are incorporated herein by reference:

- (a) <u>Amendment No. 2 to TPG Partners, LLC's Registration Statement on Form S-1</u> filed on January 4, 2022 (File No. 333-261681), which contains audited financial statements for the Registrant's latest fiscal year for which such statements have been filed;
- (b) the description of the Registrant's securities contained in the Registrant's Registration Statement on Form 8-A (File No. 001-41222) filed with the SEC under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), on January 13, 2022, including any other amendment or report filed for the purpose of updating such description; and
- (c) the Registrant's prospectus to be filed on or about January 14, 2022 (the "Prospectus") pursuant to Rule 424(b) promulgated under the Securities Act in connection with the Registration Statement on Form S-1, as amended (File No. 333-261681).

In addition, all documents subsequently filed by the Registrant pursuant to Sections 13(a), 13(c), 14 and 15(d) of the Exchange Act, prior to the filing of a post-effective amendment which indicates that all securities offered hereby have been sold or which deregisters all securities then remaining unsold, shall be deemed to be incorporated by reference herein and to be a part hereof from the date of filing of such documents; *provided*, *however*, that information deemed to have been furnished and not filed shall not be deemed to be incorporated by reference into this Registration Statement. Any statement contained in a document incorporated or deemed to be incorporated by reference herein shall be deemed to be modified or superseded for purposes of this Registration Statement to the extent that a statement contained herein or in any other subsequently filed document that also is or is deemed to be incorporated by reference herein modifies or supersedes such statement. Any such statement so modified or superseded shall not be deemed, except as so modified or superseded, to constitute a part of this Registration Statement.

Item 4. Description of Securities.

See the description of the Registrant's securities contained in the Registration Statement on Form 8-A filed on January 13, 2022 (File No. 001-41222).

Item 5. Interests of Named Experts and Counsel.

Not applicable.

Item 6. Indemnification of Directors and Officers.

Section 145(a) of the Delaware General Corporation Law (the "DGCL") provides that a corporation may indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (other than an action by or in the right of the corporation) by reason of the fact that the person is or was a director, officer, employee or agent of the corporation, or is or was serving at the request of the corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise, against expenses (including attorneys' fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by the person in connection with such action, suit or proceeding if the person acted in good faith and in a manner the person reasonably believed to be in or not opposed to the best interests of the corporation, and, with respect to any criminal action or proceeding by judgment, order, settlement, conviction, or upon a plea of nolo contendere or its equivalent, shall not, of itself, create a presumption that the person did not act in good faith and in a manner which the person reasonably believed to be in or not opposed to the best interests of the corporation, and, with respect to any criminal action or proceeding, had reasonable cause to believe that the person's conduct was unlawful.

Section 145(b) of the DGCL provides that a corporation may indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action or suit by or in the right of the corporation to procure a judgment in its favor by reason of the fact that the person is or was a director, officer, employee or agent of the corporation, or is or was serving at the request of the corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise, against expenses (including attorneys' fees) actually and reasonably incurred by the person in connection with the defense or settlement of such action or suit if the person acted in good faith and in a manner the person reasonably believed to be in or not opposed to the best interests of the corporation and except that no indemnification shall be made in respect of any claim, issue or matter as to which such person shall have been adjudged to be liable to the corporation unless and only to the extent that the Delaware Court of Chancery or the court in which such action or suit was brought shall determine upon application that, despite the adjudication of liability but in view of all of the circumstances of the case, such person is fairly and reasonably entitled to indemnity for such expenses which the Delaware Court of Chancery or such other court shall deem proper.

Section 145(c) of the DGCL provides that to the extent that a present or former director or officer of a corporation has been successful on the merits or otherwise in defense of any action, suit or proceeding referred to in subsections (a) and (b) of Section 145 of the DGCL, or in defense of any claim, issue or matter therein, such person shall be indemnified against expenses (including attorneys' fees) actually and reasonably incurred by such person in connection therewith.

Section 145(e) of the DGCL provides that expenses (including attorneys' fees) incurred by an officer or director of the corporation in defending any civil, criminal, administrative or investigative action, suit or proceeding may be paid by the corporation in advance of the final disposition of such action, suit or proceeding upon receipt of an undertaking by or on behalf of such director or officer to repay such amount if it shall ultimately be determined that such person is not entitled to be indemnified by the corporation as authorized in Section 145 of the DGCL. Such expenses, (including attorneys' fees) incurred by former directors and officers or other persons serving at the request of the corporation as directors, officers, employees or agents of another corporation, partnership, joint venture, trust or other enterprise may be so paid upon such terms and conditions, if any, as the corporation deems appropriate.

Section 145(g) of the DGCL specifically allows a Delaware corporation to purchase liability insurance on behalf of its directors and officers and to insure against potential liability of such directors and officers regardless of whether the corporation would have the power to indemnify such directors and officers under Section 145 of the DGCL.

Section 102(b)(7) of the DGCL permits a Delaware corporation to include a provision in its certificate of incorporation eliminating or limiting the personal liability of directors to the corporation or its stockholders for monetary damages for breach of fiduciary duty as a director. This provision, however, may not eliminate or limit a director's liability (1) for breach of the director's duty of loyalty to the corporation or its stockholders, (2) for acts or omissions not in good faith or involving intentional misconduct or a knowing violation of law, (3) under Section 174 of the DGCL, or (4) for any transaction from which the director derived an improper personal benefit.

Section 145 of the DGCL makes provision for the indemnification of officers and directors in terms sufficiently broad to indemnify the Registrant's officers and directors under certain circumstances from liabilities (including reimbursement for expenses incurred) arising under the Securities Act. The Registrant's bylaws provide that, subject to certain exceptions contained therein, to the fullest extent and under the circumstances permitted by Section 145 of the DGCL, the Registrant will indemnify and advance expenses on behalf of any and all of its officers and directors. The Registrant's certificate of incorporation provides that the Registrant's directors shall not be liable for monetary damages to the Registrant or its stockholders for breach of such director's fiduciary duty as a director to the fullest extent permitted by the DGCL.

The Registrant has purchased insurance policies which, within the limits and subject to the terms and conditions thereof, cover certain expenses and liabilities that may be incurred by directors and officers in connection with proceedings that may be brought against them as a result of an act or omission committed or suffered while acting as the Registrant's director or officer.

The Registrant has entered into indemnification agreements with each of its officers (as defined under Rule 16a-1(f) under the Exchange Act) and directors. The indemnification agreements provide officers and directors subject to Section 16 of the Exchange Act with contractual rights to indemnification, expense advancement and reimbursement, to the fullest extent permitted under the DGCL, subject to certain exceptions contained in those agreements.

Item 7. Exemption from Registration Claimed.

Not applicable.

Item 8. Exhibits.

The following Exhibit Index is filed as part of this Registration Statement. All exhibits are filed herewith.

EXHIBIT INDEX

| Exhibit Number | Description of Document |
|----------------|--|
| 4.1 | Certificate of Incorporation of TPG Inc. |
| 4.2 | Bylaws of TPG Inc. |
| 5.1 | Opinion of Shearman & Sterling LLP |
| 23.1 | Consent of Deloitte & Touche LLP as to TPG Partners, LLC |
| 23.2 | Consent of Deloitte & Touche LLP as to TPG Group Holdings (SBS), L.P. |
| 23.3 | Consent of Shearman & Sterling LLP (contained in the Opinion filed as Exhibit 5.1) |
| 24.1 | Power of Attorney (contained on the signature pages hereto) |
| 99.1 | TPG Inc. Omnibus Equity Incentive Plan |

Item 9. Undertakings.

The undersigned Registrant hereby undertakes:

- (1) To file, during any period in which offers or sales are being made, a post-effective amendment to this registration statement:
 - (i) To include any prospectus required by Section 10(a)(3) of the Securities Act;
- (ii) To reflect in the prospectus any facts or events arising after the effective date of the registration statement (or the most recent post-effective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth in the registration statement; and
- (iii) To include any material information with respect to the plan of distribution not previously disclosed in the registration statement or any material change to such information in the registration statement;

provided, however, that paragraphs (i) and (ii) do not apply if the information required to be included in a post-effective amendment by those paragraphs is contained in periodic reports filed with or furnished to the SEC by the Registrant pursuant to Section 13 or 15(d) of the Exchange Act that are incorporated by reference in the registration statement.

- (2) That, for the purpose of determining any liability under the Securities Act, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial *bona fide* offering thereof.
- (3) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering.

The undersigned Registrant hereby undertakes that, for purposes of determining any liability under the Securities Act, each filing of the Registrant's annual report pursuant to Section 13(a) or Section 15(d) of the Exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the Exchange Act) that is incorporated by reference in the registration statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial *bona fide* offering thereof.

The undersigned Registrant further undertakes that, insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers and controlling persons of the Registrant pursuant to the foregoing provisions, or otherwise, the Registrant has been advised that in the opinion of the SEC such indemnification is against public policy as expressed in the Securities Act and is, therefore, unenforceable.

In the event that a claim for indemnification against such liabilities (other than the payment by the Registrant of expenses incurred or paid by a director, officer or controlling person of the Registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the Registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Securities Act and will be governed by the final adjudication of such issue.

SIGNATURES

Pursuant to the requirements of the Securities Act, the Registrant certifies that it has reasonable grounds to believe that it meets all the requirements for filing on Form S-8 and has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Fort Worth, State of Texas, on January 13, 2022.

TPG INC.

By: /s/ Jon Winkelried

Name: Jon Winkelried

Title: Chief Executive Officer and Director

POWER OF ATTORNEY

Each person whose signature appears below constitutes and appoints each of Jon Winkelried, Jack Weingart and Bradford Berenson, each of them severally, his or her true and lawful attorneys-in-fact and agents, each of whom may act alone, with full power of substitution and resubstitution, for him or her and in his or her name, place and stead, in any and all capacities, to sign any and all amendments (including post-effective amendments) to this Registration Statement on Form S-8, and to sign any related registration statement that is to be effective upon filing pursuant to Rule 462(b) promulgated under the Securities Act of 1933, as amended, and all post-effective amendments thereto, and to file the same, with all exhibits thereto, and other documents in connection therewith, with the Securities and Exchange Commission, granting unto said attorneys-in-fact and agents, and each of them, full power and authority to do and perform each and every act and thing requisite and necessary to be done in and about the premises, as fully to all intents and purposes as he or she might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents or any of them, or his, her or their substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Act of 1933, as amended, this Registration Statement has been signed by the following persons in the capacities indicated on the 13th day of January, 2022.

| Signature | <u>Title</u> |
|---|--|
| /s/ David Bonderman David Bonderman | Founding Partner, Non-Executive Chairman and Director |
| /s/ James G. Coulter James G. Coulter | Founding Partner, Executive Chairman and Director |
| /s/ Jon Winkelried Jon Winkelried | Chief Executive Officer and Director (Principal Executive Officer) |
| /s/ Jack Weingart Jack Weingart | Chief Financial Officer and Director (Principal Financial Officer) |
| /s/ Martin Davidson Martin Davidson | Chief Accounting Officer (Principal Accounting Officer) |
| /s/ Todd Sisitsky Todd Sisitsky | Director |
| /s/ Anilu Vazquez-Ubarri Anilu Vazquez-Ubarri | Director |
| /s/ Maya Chorengel Maya Chorengel | Director |
| /s/ Jonathan Coslet Jonathan Coslet | Director |
| /s/ Kelvin Davis Kelvin Davis | Director |
| /s/ Ganendran Sarvananthan Ganendran Sarvananthan | Director |
| /s/ David Trujillo David Trujillo | Director |
| /s/ Mary Cranston Mary Cranston | Director |
| /s/ Deborah M. Messemer Deborah M. Messemer | Director |

CERTIFICATE OF INCORPORATION OF TPG INC.

ARTICLE I

1.1 Name. The name of the Corporation is: TPG Inc. (the "Corporation").

ARTICLE II

2.1 <u>Address</u>. The registered office of the Corporation in the State of Delaware is Suite 302, 4001 Kennett Pike, County of New Castle, Wilmington, Delaware 19807, and the name of the registered agent whose office address will be the same as the registered office is Maples Fiduciary Services (Delaware) Inc.

ARTICLE III

3.1 <u>Purpose</u>. The purpose of the Corporation is to engage in any lawful act or activity for which corporations may be organized under the Delaware General Corporation Law (the "<u>DGCL</u>"). Without limiting the generality of the foregoing, the Corporation shall have all of the powers conferred on corporations by the DGCL and other applicable law.

ARTICLE IV

- 4.1 <u>Authorized Shares</u>. The total number of shares of all classes of capital stock that the Corporation has authority to issue is 3,115,000,000 shares, consisting of: 2,240,000,000 shares of Class A Common Stock, par value \$0.001 per share ("<u>Class A Common Stock</u>"), 100,000,000 shares of nonvoting Class A Common Stock, par value \$0.001 per share ("<u>Class B Common Stock</u>"), 750,000,000 shares of Class B Common Stock, par value \$0.001 per share ("<u>Class B Common Stock</u>" and together with Class A Common Stock, and nonvoting Class A Common Stock, "<u>Common Stock</u>"), and 25,000,000 shares of Preferred Stock, par value \$0.001 per share ("<u>Preferred Stock</u>"). Notwithstanding anything to the contrary contained herein, the rights and preferences of the Common Stock shall at all times be subject to the rights and preferences of the Preferred Stock as may be set forth in one or more certificates of designations filed with the Secretary of State of the State of Delaware from time to time (each, a "<u>Preferred Designation</u>") in accordance with the DGCL and this Certificate of Incorporation (together with any Preferred Designations, this "<u>Certificate</u>").
- 4.2 <u>Common Stock</u>. The Common Stock shall have the following powers, designations, preferences and rights and qualifications, limitations and restrictions:

(a) Voting.

(i) Except as otherwise expressly provided herein or as required by the DGCL, the holders of shares of Class A Common Stock and Class B Common Stock shall vote together as one class on all matters (including the election of directors) submitted to a vote of the stockholders of the Corporation. Except as otherwise expressly provided herein or required by the DGCL, on any matter submitted to a vote of stockholders generally (i) each share of Class A Common Stock entitles its holder to one vote per share; and (ii) each share of Class B Common

Stock entitles its holder to ten votes per share; provided that from and after the Sunset (as defined below), each share of Class B Common Stock shall entitle its holder to one vote per share. The nonvoting Class A Common Stock will have the same rights and privileges as, and will rank equally and share ratably with, and be identical in all respects as to all matters to, the Class A Common Stock, except that the nonvoting Class A Common Stock will have no voting rights other than such rights as may be required by law. Notwithstanding anything included in this Certificate, to the fullest extent permitted by law, holders of Common Stock, as such, shall have no voting power with respect to, and shall not be entitled to vote on, any amendment to this Certificate that relates solely to the terms of one or more outstanding series of Preferred Stock if the holders of such affected series of Preferred Stock are entitled, either separately or together with the holders of one or more other such series, to vote thereon pursuant to this Certificate. The number of authorized shares of Preferred Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the voting power of all of the then-outstanding shares of capital stock of the Corporation entitled to vote thereon, without a vote of the holders of the Preferred Stock, or of any series thereof, unless a vote of any such holders is required pursuant to the terms of any Preferred Designation.

- (b) <u>Dividends and Distributions</u>. The holders of shares of Class A Common Stock and nonvoting Class A Common Stock shall be entitled to receive dividends and distributions to the extent permitted by law when, as and if declared by the Board of Directors. Except as otherwise provided under this Certificate, dividends and other distributions shall not be declared or paid in respect of Class B Common Stock.
- (c) <u>Liquidation</u>. Upon the dissolution, liquidation or winding up of the Corporation, subject to the rights of the holders of any outstanding series of Preferred Stock, (i) the holders of shares of Class A Common Stock and nonvoting Class A Common Stock shall be entitled to receive the assets of the Corporation available for distribution to its stockholders ratably in proportion to the number of shares held by them and (ii) the holders of Class B Common Stock, as such, shall be entitled to \$0.001 per share.
- (d) <u>Redemption</u>. Shares of Common Stock are not redeemable; provided that (i) shares of Class B Common Stock may be redeemed and cancelled only on and subject to the terms and conditions contemplated by the Partnership Agreements and the Exchange Agreement (each as defined below) and (ii) the Corporation may repurchase shares of Common Stock and the Corporation's subsidiaries may purchase shares of Common Stock.
 - (e) For purposes of this Certificate:
 - (i) The "Exchange Act" shall mean the Securities Exchange Act of 1934, as amended.
- (ii) The "Exchange Agreement" shall mean that certain Exchange Agreement, by and among the Corporation, the TPG OG Partnerships and the other parties identified therein.
 - (iii) "GP LLC" shall mean TPG GP A, LLC, a Delaware limited liability company.

- (iv) The "Initial Public Offering" means the Corporation's initial public offering of Class A Common Stock.
- (v) The "<u>Investor Rights Agreement</u>" shall mean an investor rights agreement by and among the TPG OG Partnerships and certain other investors entered into in connection with the Initial Public Offering.
 - (vi) The "LLCA" shall mean the Limited Liability Company Agreement of GP LLC.
 - (vii) The "Partnership Agreements" shall mean the limited partnership agreements of each of the TPG OG Partnerships.
- (viii) The "Sunset" shall mean 5:00 p.m. in New York City, New York on the date that directors are elected at the first annual meeting of stockholders (or pursuant to a consent of stockholders in lieu thereof) after the earlier of:
- (1) the earliest date specified by the holders of a majority of the outstanding shares of Class B Common Stock in a notice delivered to the Corporation (or, if no date is specified in such notice, the date that such notice is delivered to the Corporation); or
 - (2) the first day of the calendar quarter immediately following the fifth (5th) anniversary of the Initial Public Offering.
- (ix) The "TPG OG Partnerships" shall mean TPG Operating Group I, L.P., TPG Operating Group II, L.P., and TPG Operating Group III, L.P., each a Delaware limited partnership.

The meanings ascribed to defined terms in this Certificate shall apply whether such terms appear capitalized or lowercase, or in singular or plural form. When the terms of this Certificate refer to a specific agreement or other document or a decision by any body, person or entity to determine the meaning or operation of a provision hereof, the Secretary of the Corporation shall maintain a copy of such agreement, document or decision at the principal executive offices of the Corporation, which shall be publicly available with the Corporation's public filings or, to the extent not publicly available, a copy thereof shall be provided free of charge to any stockholder who makes a request therefor. Unless otherwise provided in this Certificate, a reference to any specific agreement or other document shall be deemed a reference to such agreement or document as amended from time to time in accordance with the terms of such agreement or document.

4.3 <u>Preferred Stock</u>. Subject to the terms of the LLCA, the Board of Directors is hereby expressly authorized, to the fullest extent as may now or hereafter be permitted by the DGCL, without any action or vote by the Corporation's stockholders (except as may otherwise be provided by the terms of any class or series of Preferred Stock then outstanding), by resolution or resolutions, at any time and from time to time, to provide for the issuance of a share or shares of Preferred Stock in one or more series and to fix for each such series (i) the number of shares constituting such series and the designation of such series, (ii) the voting powers (if any), whether full or limited, of the shares of such series, (iii) the powers, preferences, and relative, participating, optional or other special rights of the shares of each such series, and (iv) the qualifications,

limitations, and restrictions thereof, and to cause to be filed with the Secretary of State of the State of Delaware a Preferred Designation with respect thereto. Without limiting the generality of the foregoing but subject to the terms of the LLCA, to the fullest extent as may now or hereafter be permitted by the DGCL, the authority of the Board of Directors with respect to the Preferred Stock and any series thereof shall include, but not be limited to, determination of the following:

- (a) the number of shares constituting any series, which number the Board of Directors may thereafter increase or decrease (but not below the number of shares thereof then outstanding) and the distinctive designation of that series;
- (b) the dividend rate or rates on the shares of any series, the terms and conditions upon which and the periods in respect of which dividends shall be payable, whether dividends shall be cumulative and, if so, from which date or dates, and the relative rights of priority, if any, of payment of dividends on shares of that series;
- (c) the voting rights, if any, of such series and, if the shares shall have voting rights, the number of votes per share and the terms and conditions of such voting rights;
- (d) whether any series shall have conversion privileges and, if so, the terms and conditions of conversion, including provision for adjustment of the conversion rate upon such events as the Board of Directors shall determine;
- (e) whether the shares of any series shall be redeemable and, if so, the terms and conditions of such redemption, including the date or dates upon or after which they shall be redeemable and the amount per share payable in case of redemption, which amount may vary under different conditions and at different redemption dates;
- (f) whether any series shall have a sinking fund for the redemption or purchase of shares of that series, and, if so, the terms and amount of such sinking fund;
- (g) the rights of the shares of any series in the event of voluntary or involuntary liquidation, dissolution or winding up of the Corporation, and the relative rights of priority, if any, of payment of shares of that series; and
 - (h) any other powers, preferences, rights, qualifications, limitations, and restrictions of any series.

The powers, preferences and relative, participating, optional and other special rights of the shares of each series of Preferred Stock, and the qualifications, limitations or restrictions thereof, if any, may differ from those of any and all other series at any time outstanding. Unless otherwise provided in the resolution or resolutions providing for the issuance of such series of Preferred Stock, shares of Preferred Stock, regardless of series, which shall be issued and thereafter acquired by the Corporation through purchase, redemption, exchange, conversion or otherwise shall return to the status of authorized but unissued Preferred Stock, without designation as to series of Preferred Stock, and the Corporation shall have the right to reissue such shares.

4.4 Issuance and Retirement of Class B Common Stock.

- (a) Each of the TPG OG Partnerships has issued interests designated as "Common Units" (a Common Unit of each TPG OG Partnership, collectively, a "Unit") pursuant to the terms and subject to the conditions of their respective Partnership Agreements. Under certain circumstances set forth in the Partnership Agreements, additional Units may be issued in connection with certain anti-dilution events set forth in the Partnership Agreements (an "Anti-Dilution Event"). Upon notice from the TPG OG Partnerships of such an Anti-Dilution Event, the Corporation shall issue a number of shares of Class B Common Stock registered in the name of the applicable holder equal to the aggregate number of Common Units issued in connection with such Anti-Dilution Event, for no additional consideration.
- (b) If any outstanding share of Class B Common Stock shall cease to be held by a concurrent holder of a Unit (including a transferee of a Unit), such share shall automatically and without further action on the part of the Corporation or any holder of Class B Common Stock be transferred to the Corporation, shall no longer be outstanding, and all rights with respect to such share shall automatically cease and terminate. The Corporation shall take all necessary action to retire such share and such share shall not be reissued. The Board of Directors shall have the power to construe and apply the provisions of this Section 4.4(b) and accordingly make all determinations and take such actions necessary or desirable to implement such provisions. The Board of Directors shall have the right to demand that a holder of shares of Class B Common Stock supply the Corporation with complete information as to the shares and Units held by such holder.
- 4.5 No Further Issuances of Class B Common Stock. Except as set forth in Section 4.4(a) and except for the issuance of shares of Class B Common Stock in connection with a stock dividend, stock split, reclassification or similar transaction in accordance with the provisions of this Certificate, the Corporation shall not at any time after the date of the filing and effectiveness of this Certificate issue any additional shares of Class B Common Stock.
- 4.6 <u>Reservation of Class A Common Stock</u>. The Corporation will at all times reserve and keep available out of its authorized but unissued shares of Class A Common Stock for the purpose of (i) effecting the exchange of Common Units (and corresponding cancellation of shares of Class B Common Stock) and (ii) issuing shares under the Corporation's equity incentive plan.
- 4.7 <u>Protective Provisions</u>. So long as any shares of Class B Common Stock remain outstanding, the Corporation will not, whether by merger, consolidation or otherwise, amend, alter, repeal or waive this Article IV (or adopt any provision inconsistent therewith), without first obtaining the approval of the holders of a majority of the then-outstanding shares of Class B Common Stock, voting as a separate class, in addition to any other vote required by the DGCL, this Certificate or the Corporation's Bylaws, as the same may be amended or restated from time to time (the "<u>Bylaws</u>").
 - 4.8 Reclassifications, Mergers and Other Transactions.

(a) <u>Proportional Treatment</u>. No subdivision, combination or reclassification (whether by stock split, reverse stock split, stock dividend or otherwise) or dividend declared in connection with a stockholder rights plan (each, a "<u>Stock Adjustment</u>") shall be declared or made on a class of Common Stock without a corresponding Stock Adjustment being declared or made on all other classes of Common Stock in the same proportion and manner, such that the same proportionate economic and voting ownership between the holders of outstanding Class A Common Stock, nonvoting Class A Common Stock, and Class B Common Stock on the record date for such Stock Adjustment is preserved, unless (i) different treatment of the shares of each such class is approved by (A) the holders of a majority of the outstanding Class A Common Stock, (B) the holders of a majority of the outstanding nonvoting Class A Common Stock, and (C) the holders of a majority of the outstanding Class B Common Stock, each of (A), (B), and (C) voting as separate classes, or (ii) such Stock Adjustment is being declared or made in respect of the obligations set forth in Section 4.8(b)(i)-(ii). In the event of any such Stock Adjustment, the Corporation shall cause the TPG OG Partnerships to make corresponding changes to the Common Units to give effect to such Stock Adjustment.

(b) Maintenance.

- (i) The Corporation shall undertake all actions, including, without limitation, a reclassification, dividend, subdivision, combination or recapitalization, with respect to the shares of Class A Common Stock and non-voting Class A Common Stock necessary to maintain at all times a one-to-one ratio between the number of Common Units owned by the Corporation (directly or indirectly, including through its subsidiaries) and the number of outstanding shares of Class A Common Stock and nonvoting Class A Common Stock, disregarding, for purposes of maintaining the one-to-one ratio, (A) shares of Class A Common Stock issued pursuant to any equity incentive plan adopted by the Corporation from time to time, that have not vested thereunder, (B) treasury stock. The shares of Class A Common Stock referred to in clauses (A) and (B) of the foregoing sentence are referred to herein as the "Excluded Class A Common Stock."
- (ii) The Corporation shall not issue, transfer, dispose from its treasury, or repurchase shares of Class A Common Stock unless in connection with any such issuance, transfer, disposition or repurchase the Corporation takes or authorizes all requisite action such that, after giving effect to such issuance, transfer, disposition or repurchase, the number of Common Units owned by the Corporation (directly or indirectly, including through its subsidiaries) will equal on a one-for-one basis the number of outstanding shares of Class A Common Stock and nonvoting Class A Common Stock, disregarding, for purposes of maintaining the one-to-one ratio, the Excluded Class A Common Stock.
- (iii) The Corporation shall not consolidate, merge, combine or consummate any other transaction in which shares of Class A Common Stock are exchanged for or converted into other stock or securities, or the right to receive cash and/or any other property, unless in connection with any such consolidation, merger, combination or other transaction, each share of Class B Common Stock and/or Common Unit shall be entitled to be exchanged for or converted into the same kind and amount of stock or securities, cash and/or any other property, as the case may be, into which or for which each share of Class A Common Stock is exchanged or converted; provided, that the consideration for each share of Class B Common Stock and/or Common Unit shall be deemed the same kind and amount into which or for which each share of Class A Common Stock is exchanged or converted, so long as any differences in the kind and amount of stock or securities, cash and/or any other property are intended (as determined by the

Board of Directors in good faith) to maintain the relative voting power of each share of Class A Common Stock relative to each share of Class B Common Stock; provided, further, that the foregoing provisions of this Section 4.8(b)(iii) shall not apply to any action or transaction (including any consolidation, merger or combination) approved by (A) the holders of a majority of the outstanding shares of Class A Common Stock, and (B) the holders of a majority of the outstanding shares of Class B Common Stock, with each of (A) and (B) voting as separate classes.

4.9 Conversion upon Transfer. Each share of nonvoting Class A Common Stock shall automatically be converted into one fully paid and nonassessable share of Class A Common Stock on a one-for-one basis upon any transfer by any holder of nonvoting Class A Common Stock to a third party as and when permitted by the terms of the Investor Rights Agreement. The conversion of a share of nonvoting Class A Common Stock into Class A Common Stock shall occur automatically without the need for any further action by the holder of such share and whether or not the certificate representing such share (if any) is surrendered to the Corporation or its transfer agent; provided, however, that the Corporation shall not be obligated to issue a certificate evidencing the share of Class A Common Stock issuable upon such conversion unless the certificate evidencing such share of nonvoting Class A Common Stock is either delivered to the Company or its transfer agent as provided below, or the holder notifies the Company or its transfer agent that such certificate has been lost, stolen or destroyed and executes an agreement satisfactory to the Company to indemnify the Company from any loss incurred by it in connection with such certificate. Upon the occurrence of such automatic conversion of a share of nonvoting Class A Common Stock, the holder of such share of nonvoting Class A Common Stock so converted shall surrender the certificate representing such share (if any) at the office of the Company or any transfer agent for the nonvoting Class A Common Stock. From and after the time of such automatic conversion, the certificate (if any) formerly representing shares of nonvoting Class A Common Stock shall be deemed to represent shares of Class A Common Stock.

ARTICLE V

- 5.1 <u>Post-Sunset Governance</u>. From and after the Sunset, the business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors and the provisions of Section 5.8 shall be of no force or effect. In addition to the powers and authority expressly conferred upon them by the DGCL or by this Certificate or the Bylaws, from and after the Sunset, the Board of Directors shall be empowered to exercise all such powers and do all such acts and things as may be exercised or done by the Corporation.
- 5.2 <u>Number of Directors</u>. Subject to any Preferred Designation and to the limitations and requirements set forth in LLCA, the total number of authorized directors constituting the whole Board of Directors shall be determined from time to time exclusively by the Board of Directors; provided that such number shall initially be equal to the number of directors named in Section 15.2.
- 5.3 <u>Elections of Directors</u>. Directors shall be elected annually at the annual meeting of stockholders, which need not be by written ballot except to the extent provided in the Bylaws.
- 5.4 <u>Removal of Directors</u>. Subject to any Preferred Designation, and to the limitations and requirements set forth in the LLCA with respect to the parties thereto, any director may be removed at any time by the affirmative vote of the holders of at least a majority of the voting power of the Corporation's outstanding shares of stock entitled to vote generally in the election of directors, voting together as a single class, with or without cause.

- 5.5 <u>Term</u>. Directors shall hold office until the election and qualification of their respective successors in office or until such director's earlier death, resignation or removal. A director may resign at any time upon notice in writing or by electronic transmission to the Corporation.
- 5.6 <u>Director Elections by Holders of Preferred Stock.</u> Notwithstanding the foregoing, whenever the holders of any one or more series of Preferred Stock shall have the right, voting separately by series, to elect one or more directors at an annual or special meeting of stockholders, the election, filling of vacancies, removal of directors and other features of such one or more directorships shall be governed by the terms of such one or more series of Preferred Stock to the extent permitted by law. During any period when the holders of any series of Preferred Stock, voting separately as a series or together with one or more series, have the right to elect additional directors, then upon commencement and for the duration of the period during which such right continues: (i) the then otherwise total authorized number of directors of the Corporation shall automatically be increased by such specified number of directors, and the holders of such Preferred Stock shall be entitled to elect the additional directors so provided for or fixed pursuant to said provisions, and (ii) each such additional director shall serve until such director's successor shall have been duly elected and qualified, or until such director's right to hold such office terminates pursuant to said provisions, whichever occurs earlier, subject to his or her earlier death, resignation, disqualification or removal. Except as otherwise provided by the Board of Directors in the resolution or resolutions establishing such series, whenever the holders of any series of Preferred Stock having such right to elect additional directors are divested of such right pursuant to the provisions of such stock, the terms of office of all such additional directors elected by the holders of such stock, or elected to fill any vacancies resulting from the death, resignation, disqualification or removal of such additional directors, shall forthwith terminate (in which case each such director thereupon shall cease to be qualified as, and shall cease to be, a director) and t
- 5.7 <u>Vacancies and Newly Created Directorships</u>. Subject to any Preferred Designation, and to the limitations and requirements set forth in the LLCA, any vacancies in the Board of Directors for any reason and any newly created directorships resulting by reason of any increase in the number of directors shall be filled only by the Board of Directors acting by a majority of the directors then in office, even if less than a quorum, or by a sole remaining director, and any directors so appointed shall hold office until the next election of directors and until their successors are duly elected and qualified, provided that, prior to the Sunset, such vacancies may also be filled by a majority of the voting power of the Corporation's outstanding shares of stock entitled to vote generally in the election of directors, voting together as a single class.

5.8 Pre-Sunset Governance.

(a) <u>Generally</u>. Notwithstanding anything to the contrary in this Certificate, but subject to Article XIII, prior to the Sunset, the business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors, an executive committee (the

"Executive Committee") and the Other Committees (as defined below) each with the power and authority conferred pursuant to or in the manner contemplated by, this Section 5.8. Except as otherwise provided in this Section 5.8, the provisions of the DGCL that apply to directors or a board of directors shall apply to the directors, the Board of Directors, the Executive Committee and the Other Committees.

- (b) Executive Committee. Except as set forth in Section 5.8(c) and Section 5.8(d), the Executive Committee shall exclusively have all power and authority conferred or imposed upon a board of directors by the DGCL or upon the Board of Directors by this Certificate or the Bylaws. Except as otherwise provided in this Section 5.8, the provisions of this Certificate and the Bylaws that apply to directors and the Board of Directors shall apply to the Executive Committee and the directors who comprise the Executive Committee. The members of the Executive Committee shall be elected by the holders of Class A Common Stock and Class B Common Stock, voting together as a single class. The Executive Committee shall have the power to delegate any of its power and authority from time to time; provided that any such delegation, or any modification or repeal of any such delegation then in effect, shall require the approval of GP LLC as and when required in accordance with the LLCA.
- (c) <u>Board of Directors</u>. The Board of Directors shall only have exclusive power and authority with respect to: (i) approving or adopting, or recommending to the stockholders, any action or matter (other than the election or removal of directors) expressly required by the DGCL to be submitted to stockholders for approval, (ii) adopting, amending or repealing any provision of the Bylaws in accordance with the LLCA, (iii) adopting the initial Committee Charters (as defined below) and the initial charter of the Executive Committee; (iv) fixing the total number of directors constituting, appointing the initial directors to, and filling vacancies in the Executive Committee in accordance with the LLCA, (v) fixing the total number of directors constituting, appointing directors to, removing directors from, and filling vacancies in the Executive Committee and the Other Committees in accordance with the LLCA, (vi) fixing the total number of directors constituting and filling vacancies in the Board of Directors in accordance with the LLCA and (vii) any decision to bring, commence or initiate any action, suit or proceeding by or on behalf of the Corporation against one or more of its current or former directors, officers or stockholders or involving any other claim arising under the internal affairs doctrine (including any decisions regarding any demand by one or more stockholders to bring or investigate any claims with respect thereto) (clauses (i)-(vii) the "Exclusive Board Matters"). The Board of Directors shall be empowered to exercise all such power and do all such acts and things as may be exercised or done by the Corporation with respect to the Exclusive Board Matters and such matters for which action by the Board of Directors is contemplated under any Committee Charter. The Board of Directors shall also have such power and authority to the extent delegated by the Executive Committee.
- (d) Other Committees. A committee of the Board of Directors may have power and authority with respect to: (i) any action or matter required to be delegated to a committee of independent directors by any applicable rule, regulation, guideline or requirement of any securities exchange on which shares of the Corporation's capital stock are listed for trading, and (ii) any action or matter delegated to such committee by the Board of Directors or the Executive Committee. The Audit Committee, the Compensation Committee and the Conflicts

Committee (the "Other Committees") shall have the power and authority set forth in the charters adopted by the Board of Directors in connection with the Initial Public Offering (each such charter, as amended from time to time in accordance with its terms and with the approval of GP LLC in accordance with the LLCA, a "Committee Charter"). The Other Committees shall have exclusive power and authority to the extent so provided in the Committee Charters (the "Exclusive Committee Matters"). Except as otherwise provided in this Section 5.8, the provisions of this Certificate and the Bylaws that apply to directors and the Board of Directors shall apply to the Other Committees and the directors who comprise the Other Committees. The Other Committees shall be empowered to exercise all such power and do all such acts and things as may be exercised or done by the Corporation with respect to the Exclusive Committee Matters and such other matters within their authority.

ARTICLE VI

- 6.1 <u>Advance Notice</u>. Advance notice of nominations for the election of directors or proposals of other business to be considered by stockholders, other than nominations and proposals submitted (i) by or at the direction of the Board of Directors or (ii) pursuant to the LLCA, must be submitted to the Corporation in the manner provided in the Bylaws. Without limiting the generality of the foregoing, the Bylaws may require that such advance notice include such information as the Board of Directors may deem appropriate or useful.
- 6.2 <u>No Stockholder Action by Consent</u>. Subject to any Preferred Designation, from and after the Sunset, any action required or permitted to be taken by the stockholders of the Corporation must be effected at a duly called annual or special meeting of such stockholders of the Corporation and may not be effected by any consent in lieu of a meeting by such stockholders.
- 6.3 <u>Postponement, Conduct and Adjournment of Meetings</u>. To the fullest extent permitted by law, any meeting of stockholders may be postponed by action of the Board of Directors at any time in advance of such meeting. The Board of Directors shall have the power to adopt such rules and regulations for the conduct of the meetings and management of the affairs of the Corporation as they may deem proper and the power to adjourn any meeting of stockholders without a vote of the stockholders, which powers may be delegated by the Board of Directors to the chairperson of such meeting or other officer in either such rules and regulations or pursuant to the Bylaws.
- 6.4 <u>Special Meetings of Stockholders</u>. Subject to any Preferred Designation, special meetings of the stockholders of the Corporation, for any purpose or purposes, may be called at any time, but only by or at the direction of the Board of Directors, the Executive Chairperson of the Board of Directors, the Non-Executive Chairperson of the Board of Directors or the Chief Executive Officer of the Corporation.

ARTICLE VII

7.1 <u>Limited Liability of Directors</u>. A director of the Corporation shall not be liable to the Corporation or its stockholders for monetary damages for breach of fiduciary duty as a director, except for liability (i) for any breach of the director's duty of loyalty to the Corporation or its stockholders, (ii) for acts or omissions not in good faith or which involve intentional misconduct

or a knowing violation of law, (iii) under Section 174 of the DGCL, or (iv) for any transaction from which the director derived an improper personal benefit. If the DGCL is amended to authorize corporate action further eliminating or limiting the liability of directors, then the liability of a director of the Corporation shall be eliminated or limited to the fullest extent permitted by the DGCL, as so amended. Any alteration, amendment, addition to or repeal of this Section 7.1, adoption of any provision of this Certificate inconsistent with this Section 7.1, or (to the fullest extent permitted by the DGCL) any modification of law, shall not reduce, eliminate or adversely affect any right or protection of a director of the Corporation existing at the time of such alteration, amendment, addition to, repeal, adoption or modification.

7.2 <u>Corporate Opportunities</u>. The Corporation has waived certain corporate opportunities as identified in the LLCA, such that GP LLC and the other persons specified therein shall not be liable to the Corporation, its affiliates or its stockholders for breach of any fiduciary duty as a stockholder or director of the Corporation from pursuit of such opportunities. Any person or entity purchasing or otherwise acquiring or holding any interest in the shares of capital stock of the Corporation shall be deemed to have notice of and consented to the foregoing.

ARTICLE VIII

8.1 <u>Location of Meetings and Books</u>. Meetings of stockholders may be held within or without the State of Delaware, as the Bylaws may provide. The books of the Corporation may be kept (subject to any provision contained in the DGCL) outside the State of Delaware at such place or places as may be designated from time to time by the Board of Directors (or its designee) or in the Bylaws of the Corporation.

ARTICLE IX

9.1 <u>Amendments to Bylaws</u>. Subject to any limitations and requirements set forth in the LLCA, in furtherance and not in limitation of the powers conferred upon it by the laws of the State of Delaware, the Board of Directors is expressly authorized and empowered to make, alter, amend, add to or repeal any and all Bylaws by a resolution of the Board of Directors. In addition to any vote required by this Certificate or applicable law, the affirmative vote of the holders of at least a majority of the voting power of the Corporation's then outstanding shares entitled to vote generally in the election of directors, voting together as a single class, shall be required for the stockholders to make, alter, amend, add to or repeal any or all Bylaws of the Corporation or to adopt any provision inconsistent therewith.

ARTICLE X

10.1 <u>Section 203 of the DGCL</u>. The Corporation shall not be governed by Section 203 of the DGCL ("<u>Section 203</u>"), and the restrictions contained in Section 203 shall not apply to the Corporation.

- 10.2 <u>Limitations on Business Combinations</u>. The Corporation shall not engage in any business combination (as defined below), at any point in time at which the Corporation's Common Stock is registered under Section 12(b) or 12(g) of the Exchange Act, with any interested stockholder (as defined below) for a period of three (3) years following the time that such stockholder became an interested stockholder, unless:
- (a) prior to such time, the Board of Directors approved either the business combination or the transaction which resulted in the stockholder becoming an interested stockholder; or
- (b) upon consummation of the transaction which resulted in the stockholder becoming an interested stockholder, the interested stockholder owned at least eighty-five percent (85%) of the voting stock (as defined below) of the Corporation outstanding at the time the transaction commenced, excluding for purposes of determining the voting stock outstanding (but not the outstanding voting stock owned by the interested stockholder) those shares owned by (1) persons who are directors and also officers and (2) employee stock plans in which employee participants do not have the right to determine confidentially whether shares held subject to the plan will be tendered in a tender or exchange offer; or
- (c) at or subsequent to such time, the business combination is approved by the Board and authorized at an annual or special meeting of stockholders, and not by consent, by the affirmative vote of at least 662/3% of the outstanding voting stock of the Corporation which is not owned by the interested stockholder.
 - 10.3 Exceptions to Prohibition on Interested Stockholder Transactions. The restrictions contained in this Article X shall not apply if:
- (a) a stockholder becomes an interested stockholder inadvertently and (i) as soon as practicable divests itself of ownership of sufficient shares so that the stockholder ceases to be an interested stockholder; and (ii) would not, at any time within the three-year period immediately prior to a business combination between the Corporation and such stockholder, have been an interested stockholder but for the inadvertent acquisition of ownership; or
- (b) the business combination is proposed prior to the consummation or abandonment of and subsequent to the earlier of the public announcement or the notice required hereunder of a proposed transaction which (i) constitutes one of the transactions described in the second sentence of this Section 10.3(b); (ii) is with or by a person who either was not an interested stockholder during the previous three years or who became an interested stockholder with the approval of the Board of Directors; and (iii) is approved or not opposed by a majority of the directors then in office (but not less than one) who were directors prior to any person becoming an interested stockholder during the previous three years or were recommended for election or elected to succeed such directors by a majority of such directors. The proposed transactions referred to in the preceding sentence are limited to (x) a merger or consolidation of the Corporation (except for a merger in respect of which, pursuant to Section 251(f) of the DGCL, no vote of the stockholders of the Corporation is required); (y) a sale, lease, exchange, mortgage, pledge, transfer or other disposition (in one transaction or a series of transactions), whether as part of a dissolution or otherwise, of assets of the Corporation or of any direct or indirect majority owned subsidiary of the Corporation (other than to any direct or indirect wholly-owned subsidiary or to the Corporation determined on a consolidated basis

or the aggregate market value of all the outstanding stock (as defined hereinafter) of the Corporation; or (z) a proposed tender or exchange offer for fifty percent (50%) or more of the outstanding voting stock of the Corporation. The Corporation shall give not less than 20 days' notice to all interested stockholders prior to the consummation of any of the transactions described in clause (x) or (y) of the second sentence of this Section 10.3(b).

- 10.4 <u>Definitions</u>. For purposes of this Article X, references to:
- (a) "affiliate" means a person that directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, another person
- (b) "associate" when used to indicate a relationship with any person, means: (1) any corporation, partnership, unincorporated association or other entity of which such person is a director, officer or partner or is, directly or indirectly, the owner of twenty percent (20%) or more of any class of voting stock; (2) any trust or other estate in which such person has at least a twenty percent (20%) beneficial interest or as to which such person serves as trustee or in a similar fiduciary capacity; and (3) any relative or spouse of such person, or any relative of such spouse, who has the same residence as such person.
 - (c) "business combination" when used in reference to the Corporation and any interested stockholder of the Corporation, means:
- (i) any merger or consolidation of the Corporation or any direct or indirect majority-owned subsidiary of the Corporation (i) with the interested stockholder, or (ii) with any other corporation, partnership, unincorporated association or other entity if the merger or consolidation is caused by the interested stockholder and as a result of such merger or consolidation Section 10.2 is not applicable to the surviving entity;
- (ii) any sale, lease, exchange, mortgage, pledge, transfer or other disposition (in one transaction or a series of transactions), except proportionately as a stockholder of the Corporation, to or with the interested stockholder, whether as part of a dissolution or otherwise, of assets of the Corporation or of any direct or indirect majority-owned subsidiary of the Corporation which assets have an aggregate market value equal to ten percent (10%) or more of either the aggregate market value of all the assets of the Corporation determined on a consolidated basis or the aggregate market value of all the outstanding stock of the Corporation;
- (iii) any transaction which results in the issuance or transfer by the Corporation or by any direct or indirect majority-owned subsidiary of the Corporation of any stock of the Corporation or of such subsidiary to the interested stockholder, except: (a) pursuant to the exercise, exchange or conversion of the securities exercisable for, exchangeable for or convertible into stock of the Corporation or any subsidiary which securities were outstanding prior to the time that the interested stockholder became such; (b) pursuant to a merger under Section 251(g) of the DGCL; (c) pursuant to a dividend or distribution paid or made, or the exercise, exchange or conversion of securities exercisable for, exchangeable for or convertible into stock of the Corporation or any such subsidiary which security is distributed, pro rata to all holders of a class or series of stock of the Corporation subsequent to the time the interested stockholder became such; (d) pursuant to an exchange offer by the Corporation to purchase stock made on the same

terms to all holders of said stock; or (e) any issuance or transfer of stock by the Corporation; provided, however, that in no case under items (c)-(e) of this Section 10.4(c)(iii) shall there be an increase in the interested stockholder's proportionate share of the stock of any class or series of the Corporation or the voting stock of the Corporation (except as a result of immaterial changes due to fractional share adjustments);

(iv) any transaction involving the Corporation or any direct or indirect majority-owned subsidiary of the Corporation which has the effect, directly or indirectly, of increasing the proportionate share of the stock of any class or series, or securities convertible into the stock of any class or series, of the Corporation or of any such subsidiary which is owned by the interested stockholder, except as a result of immaterial changes due to fractional share adjustments or as a result of any purchase or redemption of any shares of stock not caused, directly or indirectly, by the interested stockholder; or

(v) any receipt by the interested stockholder of the benefit, directly or indirectly (except proportionately as a stockholder of the Corporation), of any loans, advances, guarantees, pledges, or other financial benefits (other than those expressly permitted in Section 10.4(c)(i)-(iv)) provided by or through the Corporation or any direct or indirect majority-owned subsidiary.

- (d) "control." including the terms "controlling," "controlled by." and "under common control with" means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a person, whether through the ownership of voting stock, by contract, or otherwise. A person who is the owner of twenty percent (20%) or more of the outstanding voting stock of any corporation, partnership, unincorporated association or other entity shall be presumed to have control of such entity, in the absence of proof by a preponderance of the evidence to the contrary. Notwithstanding the foregoing, a presumption of control shall not apply where such person holds voting stock, in good faith and not for the purpose of circumventing this Article X, as an agent, bank, broker, nominee, custodian or trustee for one or more owners who do not individually or as a group (as such term is used in Rule 13d-5 promulgated under the Exchange Act as such rule is in effect as of the date of this Certificate) have control of such entity.
- (e) "interested stockholder" means any person (other than the Corporation or any direct or indirect majority-owned subsidiary of the Corporation) that (i) is the owner of fifteen percent (15%) or more of the outstanding voting stock of the Corporation, or (ii) is an affiliate or associate of the Corporation and was the owner of fifteen percent (15%) or more of the outstanding voting stock of the Corporation at any time within the three (3) year period immediately prior to the date on which it is sought to be determined whether such person is an interested stockholder, and the affiliates and associates of such person; provided, however, that the term "interested stockholder" shall in no case include or be deemed to include (1) the Principal Holders, the Principal Holder Director Transferees and the Principal Holder Indirect Transferees, or (2) any person whose ownership of share in excess of the fifteen percent (15%) limitation set forth herein is the result of any action taken solely by the Corporation; provided that such person specified in this clause (2) shall be an interested stockholder if thereafter such person acquires additional shares of voting stock of the Corporation, except as a result of further corporate action not caused, directly or indirectly, by such person. For the purpose of determining whether a person is an interested stockholder, the voting stock of the Corporation deemed to be outstanding shall include voting

stock deemed to be owned by the person through application of the definition of "owner" below but shall not include any other unissued stock of the Corporation which may be issuable pursuant to any agreement, arrangement or understanding, or upon exercise of conversion rights, warrants or options, or otherwise.

- (f) "owner," including the terms "own" and "owned," when used with respect to any stock, means a person that individually or with or through any of its affiliates or associates:
 - (i) beneficially owns such stock, directly or indirectly; or
- (ii) has (1) the right to acquire such stock (whether such right is exercisable immediately or only after the passage of time) pursuant to any agreement, arrangement or understanding, or upon the exercise of conversion rights, exchange rights, warrants, options, or otherwise; provided, however, that a person shall not be deemed the owner of stock tendered pursuant to a tender or exchange offer made by such person or any of such person's affiliates or associates until such tendered stock is accepted for purchase or exchange; or (2) the right to vote such stock pursuant to any agreement, arrangement or understanding; provided, however, that a person shall not be deemed the owner of any stock because of such person's right to vote such stock if the agreement, arrangement or understanding to vote such stock arises solely from a revocable proxy or consent given in response to a proxy or consent solicitation made to ten (10) or more persons; or
- (iii) has any agreement, arrangement or understanding for the purpose of acquiring, holding, voting (except voting pursuant to a revocable proxy or consent as described in item (2) of Section 10.4(f)(ii)), or disposing of such stock with any other person that beneficially owns, or whose affiliates or associates beneficially own, directly or indirectly, such stock.
 - (g) "person" means any individual, corporation, partnership, unincorporated association or other entity.
- (h) "Principal Holder Direct Transferee" means any person that acquires (other than in a registered public offering), directly from one or more of the Principal Holders, beneficial ownership of 5% or more of the then-outstanding voting stock of the Corporation.
- (i) "Principal Holders" means GP LLC, affiliates of GP LLC and their respective successors; provided, however, that the term "Principal Holders" shall not include the Corporation or any of the Corporation's direct or indirect subsidiaries.
- (j) "<u>Principal Holder Indirect Transferee</u>" means any person that acquires (other than in a registered public offering) directly from any Principal Holder Direct Transferee or any other Principal Holder Indirect Transferee beneficial ownership of 5% or more of the then-outstanding voting stock of the Corporation.
 - (k) "stock" means, with respect to any corporation, capital stock and, with respect to any other entity, any equity interest.

(l) "voting stock" means stock of any class or series entitled to vote generally in the election of directors. Every reference to a percentage of voting stock in this Article X shall refer to such percentage of votes of such voting stock.

ARTICLE XI

11.1 Exclusive Forum.

(a) Unless the Corporation consents in writing to the selection of an alternative forum, the Court of Chancery of the State of Delaware (or, if the Court of Chancery lacks jurisdiction, a state court located within the State of Delaware or the federal district court for the District of Delaware) shall, to the fullest extent permitted by law, be the sole and exclusive forum for: any (a) derivative action or proceeding brought on behalf of the Corporation; (b) action asserting a claim of breach of a fiduciary duty owed by or other wrongdoing by any current or former director, officer, employee, agent or stockholder of the Corporation to the Corporation or the Corporation's stockholders; (c) action asserting a claim arising under any provision of the DGCL or this Certificate or the Bylaws (as either may be amended from time to time), or as to which the DGCL confers jurisdiction on the Court of Chancery of the State of Delaware; or (d) action asserting a claim governed by the internal affairs doctrine. This Section 11.1(a) shall not apply in any respect to claims or causes of action brought to enforce a duty or liability created by the Securities Act of 1933, as amended (the "Securities Act"), or the Exchange Act, or the rules and regulations promulgated thereunder or any other claim or cause of action for which the federal courts have exclusive jurisdiction.

(b) Unless the Corporation consents in writing to the selection of an alternative forum, the federal district courts of the United States of America shall be the sole and exclusive forum for the resolution of any action asserting a claim arising under the Securities Act or the rules and regulations promulgated thereunder. The Exchange Act provides for sole and exclusive jurisdiction by federal district courts of the United States of America for any actions asserted thereunder.

ARTICLE XII

12.1 <u>Amendment</u>. The Corporation reserves the right, at any time and from time to time, to alter, amend, add to or repeal any provision contained in this Certificate in any manner now or hereafter prescribed by the laws of the State of Delaware, and all rights, preferences, privileges and powers of any nature conferred upon stockholders, directors or any other persons herein are granted subject to this reservation.

ARTICLE XIII

13.1 <u>LLCA</u>. Notwithstanding anything in this Certificate to the contrary (including the provisions of Article V), (i) the Corporation is not authorized to engage in any act or activity that would constitute a breach by the Corporation of the LLCA, including by any amendment to this Certificate, and (ii) the Corporation shall lack the power to engage in any such act or activity, unless (in the case of either of clauses (i) or (ii)) such act or activity is approved, or ratified after such act or activity occurs, by the parties to the LLCA. For the avoidance of doubt, a breach of the LLCA shall not occur if an act or activity would constitute a breach of a contractual right of one or more of the parties to the LLCA and such right has been waived (either by a limited waiver or otherwise) by such parties.

ARTICLE XIV

14.1 <u>Severability</u>. If any provision (or any part thereof) of this Certificate shall be held to be invalid, illegal or unenforceable as applied to any circumstance for any reason whatsoever, the validity, legality and enforceability of such provisions in any other circumstance and of the remaining provisions of this Certificate including, without limitation, each portion of any section of this Certificate containing any such provision held to be invalid, illegal or unenforceable that is not itself held to be invalid, illegal or unenforceable) shall not in any way be affected or impaired thereby.

ARTICLE XV

15.1 <u>Incorporator</u>. The name of the incorporator is Bradford Berenson. The address of the incorporator is 345 California Street, Suite 3300, San Francisco, CA 94104.

15.2 <u>Initial Board of Directors</u>. The initial directors of the Corporation shall be David Bonderman, James G. Coulter, Maya Chorengel, Jonathan Coslet, Kelvin Davis, Ganen Sarvananthen, Todd Sisitsky, David Trujillo, Anilu Vazquez-Ubarri, Jack Weingart and Jon Winkelried. The address of each such director is 345 California Street, Suite 3300, San Francisco, CA 94104.

[The remainder of this page is intentionally left blank.]

I, THE UNDERSIGNED, being the incorporator, for the purpose of forming a corporation under the laws of the State of Delaware do make, file and record this Certificate of Incorporation, do certify that the facts herein stated are true, and, accordingly, have hereto set my hand this 12th day of the month of January in the year 2022.

/s/ Bradford Berenson Bradford Berenson Incorporator

[CERTIFICATE OF INCORPORATION OF TPG INC.]

BYLAWS

OF

TPG INC. (a Delaware corporation)

ARTICLE I STOCKHOLDERS

Section 1.01. <u>Annual Meetings</u>. The annual meeting of the stockholders of TPG Inc. (the "<u>Corporation</u>") for the election of directors and for the transaction of such other business as properly may come before such meeting shall be held at such place, either within or without the State of Delaware, or, within the sole discretion of the Board of Directors of the Corporation (the "<u>Board of Directors</u>" or "<u>Board</u>"), and subject to such guidelines and procedures as the Board of Directors may authorize, by means of remote communication as authorized by the General Corporation Law of the State of Delaware (the "<u>DGCL</u>"), and at such date and at such time as may be fixed from time to time by resolution of the Board of Directors (or its designee) and set forth in the notice of the meeting.

Section 1.02. Special Meetings. Special meetings of the stockholders of the Corporation may be called as set forth in the Certificate of Incorporation of the Corporation (as it may be amended from time to time, and together with any certificate of designations relating to any series of preferred stock (each, a "Preferred Designation") in effect from time to time, the "Certificate of Incorporation"). Any such special meetings of the stockholders shall be held at such places, within or without the State of Delaware, or, within the sole discretion of the Board of Directors, and subject to such guidelines and procedures as the Board of Directors may authorize, by means of remote communication as authorized by the DGCL, as shall be specified in the respective notice thereof.

Section 1.03. <u>Stockholder Action by Consent</u>. Any action required or permitted to be taken at any annual or special meeting of stockholders of the Corporation may be taken without a meeting, without prior notice and without a vote only to the extent permitted by and in the manner provided in the Certificate of Incorporation and in accordance with applicable law.

Section 1.04. Notice of Meetings; Waiver.

(a) Unless otherwise required by the DGCL or the Certificate of Incorporation, the Secretary of the Corporation or any Assistant Secretary shall give notice of the place, if any, date and time of each meeting of the stockholders, and, in the case of a special meeting, the purpose or purposes for which such meeting is called, the record date for determining the stockholders entitled to vote at the meeting (if such date is different from the record date for determining the stockholders entitled to notice of the meeting) and the means of remote communication, if any, by which stockholders and proxyholders may be deemed to be present in person and vote at such meeting. The notice shall be given not fewer than ten (10) nor more than sixty (60) days before the date on which the meeting is to be held, to each stockholder entitled to vote at such meeting as of the record date for determining the stockholders entitled to notice of the meeting, except as otherwise required by law, the Certificate of Incorporation or these Bylaws.

(b) All such notices shall be delivered in writing or in any other manner permitted by the DGCL. If mailed, such notice shall be deemed given when deposited in the United States mail, postage prepaid, addressed to the stockholder at his, her or its mailing address as it appears on the records of the Corporation. If delivered by courier service, notice shall be deemed given at the earlier of when the notice is received or left at such stockholder's address as it appears on the records of the Corporation. If given by electronic mail, notice shall be deemed given when directed to such stockholder's electronic mail address unless the stockholder has notified the Corporation in writing or by electronic transmission of an objection to receiving notice by electronic mail or such notice is prohibited by the DGCL or Section 1.04(c). Notice to stockholders may also be given by other forms of electronic transmission consented to by the stockholder. If given by a posting on an electronic network together with separate notice to the stockholder of such specific posting, such notice shall be deemed given upon the later of (x) such posting and (y) the giving of such separate notice. If notice is given by any other form of electronic transmission, such notice shall be deemed given when directed to the stockholder.

(c) Notwithstanding Section 1.04(b), a notice may not be given by electronic transmission (including email) from and after the time that: (i) the Corporation is unable to deliver by electronic transmission two (2) consecutive notices given by the Corporation; and (ii) such inability becomes known to the Secretary or an Assistant Secretary of the Corporation or to the transfer agent or other person responsible for the giving of notice. However, the inadvertent failure to treat such inability as a revocation shall not invalidate any meeting or other action. For purposes of these Bylaws, except as otherwise limited by applicable law, the term "electronic transmission" means any form of communication not directly involving the physical transmission of paper, including the use of, or participation in, one or more electronic networks or databases (including one or more distributed electronic networks or databases), that creates a record that may be retained, retrieved and reviewed by a recipient thereof, and that may be directly reproduced in paper form by such recipient through an automated process. A notice by electronic mail must include a prominent legend that the communication is an important notice regarding the Corporation. A notice by electronic mail will be deemed to include any files attached thereto and any information hyperlinked to a website if such electronic mail includes the contact information of an officer or agent of the Corporation who is available to assist with accessing such files or information. "Electronic mail" and "electronic mail address" are defined in accordance with Sections 232(d)(2) and 232(d)(3) of the DGCL, respectively.

(d) A written waiver of any notice signed by the person entitled to notice, or a waiver by electronic transmission by the person entitled to notice, whether before or after the time stated therein, shall be deemed equivalent to notice. Neither the business to be transacted at, nor the purpose of, any annual or special meeting of the stockholders need be specified in a waiver of notice or any waiver by electronic transmission unless so required by the Certificate of Incorporation or the Bylaws. Attendance of a stockholder at a meeting of stockholders shall constitute a waiver of notice of such meeting, except when the stockholder attends a meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business on the ground that the meeting is not lawfully called or convened.

(e) If a stockholder meeting is to be held by means of remote communication and stockholders will take action at such meeting, the notice of such meeting must: (i) specify the means of remote communication, if any, by which stockholders and proxyholders may be deemed to be present and vote at such meeting; and (ii) if such meeting is to be held solely by means of remote communication, provide, or be accompanied by, the information required to access the stockholder list. A waiver of notice may be given by electronic transmission.

Section 1.05. Quorum. Except as otherwise required by law or by the Certificate of Incorporation or these Bylaws, at each meeting of stockholders the presence in person or by proxy of the holders of record of a majority in voting power of the shares of capital stock of the Corporation issued and outstanding and entitled to vote at the meeting of stockholders shall constitute a quorum for the transaction of business at such meeting. Where a separate vote by one or more classes or series is required, the presence in person or by proxy of the holders of record of a majority in voting power of the shares entitled to vote shall constitute a quorum entitled to take action with respect to that vote on that matter. Shares of the Corporation's capital stock shall neither be entitled to vote nor counted for quorum purposes if such shares belong to (i) the Corporation, (ii) another corporation, if a majority of the shares entitled to vote in the election of directors of such other corporation is held, directly or indirectly, by the Corporation or (iii) any other entity, if a majority of the voting power of such other entity is held or otherwise controlled, directly or indirectly, by the Corporation; provided, however, that the foregoing shall not limit the right of the Corporation to vote stock, including but not limited to its own stock, held by it in a fiduciary capacity.

Section 1.06. Voting.

(a) Each stockholder entitled to vote at any meeting of stockholders shall be entitled to such number of votes as set forth in the Certificate of Incorporation.

(b) Except as otherwise required by law, the Certificate of Incorporation or these Bylaws, and subject to the limitations and requirements set forth in the GP LLCA with respect to the parties thereto, directors shall be elected as set forth in Section 2.02 of these Bylaws. All other matters presented to the stockholders at a meeting at which a quorum is present shall, unless a different or minimum vote is required by the Certificate of Incorporation, these Bylaws, the rules or regulations of any stock exchange applicable to the Corporation, or any law or regulation applicable to the Corporation or its securities, in which case such different or minimum vote shall be the applicable vote on the matter, be decided by the affirmative vote of at least a majority of the voting power of the stock present in person or represented by proxy and entitled to vote on the subject matter, voting as a single class, and where a separate vote by a class or series or classes or series is required, if a quorum of such class or series or classes or series is present, such act shall be authorized by the affirmative vote of at least a majority of the voting power of the stock of such class or series or classes or series present in person or represented by proxy and entitled to vote on the subject matter.

Section 1.07. <u>Voting by Ballot</u>. No vote of the stockholders on an election of directors or any other matter need be taken by written ballot or by electronic transmission unless otherwise provided in the Certificate of Incorporation or required by law.

Section 1.08. <u>Postponement and Adjournment</u>. To the fullest extent permitted by law, any meeting of stockholders may be postponed, rescheduled or cancelled by action of the Board of Directors at any time in advance of such meeting. Any meeting of stockholders may be adjourned by the Chair of the meeting, whether or not a quorum is present. In addition, in the absence of a quorum, the stockholders so present may, by the affirmative vote of the holders of a majority in voting power of the shares of the Corporation which are present in person or by proxy and entitled to vote thereon, adjourn the meeting from time to time until a quorum shall attend. Notice of any adjourned meeting of the stockholders of the Corporation need not be given if the time, place, if any, thereof, and the means of remote communications, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such adjourned meeting are announced at the meeting at which the adjournment is taken; provided that if the adjournment is for more than thirty (30) days, a notice of the adjourned meeting shall be given to each stockholder of record entitled to vote at the meeting. If after the adjournment a new record date for determination of stockholders entitled to vote is fixed for the adjourned meeting, the Board of Directors shall fix as the record date for determining stockholders entitled to notice of such adjourned meeting the same or an earlier date as that fixed for determination of stockholders entitled to vote at the adjourned meeting, and shall give notice of the adjourned meeting to each stockholder of record as of the record date so fixed for notice of such adjourned meeting. At the adjourned meeting, the Corporation may transact any business which might have been transacted at the original meeting.

Section 1.09. Proxies. Any stockholder entitled to vote at any meeting of the stockholders may authorize another person or persons to vote at any such meeting and express such vote on behalf of such stockholder by a proxy authorized by an instrument in writing or by a transmission permitted by law and filed in accordance with the procedure established for the meeting. No such proxy shall be voted or acted upon after the expiration of three (3) years from the date of such proxy, unless such proxy provides for a longer period. A proxy shall be irrevocable if it states that it is irrevocable and if, and only as long as, it is coupled with an interest. A proxy may be made irrevocable regardless of whether the interest with which it is coupled is an interest in the stock itself or an interest in the Corporation generally. A stockholder may revoke any proxy which is not irrevocable by attending the meeting and voting in person or by filing with the Secretary of the Corporation either an instrument in writing revoking the proxy or another duly executed proxy bearing a later date.

Section 1.10. <u>Organization; Procedure</u>. At every meeting of stockholders, the Chair of such meeting shall be the Executive Chairman (or Chair, as applicable), or in the event of such person's absence or disability, a Chair chosen by the Board of Directors. The Secretary of the Corporation, or in the event of his or her absence or disability, an Assistant Secretary, if any, or if there be no Assistant Secretary, in the absence of the Secretary of the Corporation, an appointee of the Chair of the meeting, shall act as Secretary of the meeting. The order of business and all other matters of procedure at every meeting of stockholders may be determined by the Chair of such meeting. Except to the extent inconsistent with such rules and regulations as adopted by the Board of Directors, the Chair of the meeting shall have the authority to adopt and enforce such rules and regulations for the conduct of any meeting of stockholders and the safety of those in attendance as, in the judgment of the Chair, are necessary, appropriate or convenient for the conduct of the meeting. Rules and regulations for the conduct of meetings of stockholders, whether adopted by the Board of Directors or by the Chair of the meeting, may include, without limitation,

establishing: (i) an agenda or order of business for the meeting; (ii) rules and procedures for maintaining order at the meeting and the safety of those present; (iii) limitations on attendance at or participation in the meeting to stockholders entitled to vote at the meeting, their duly authorized and constituted proxies and such other persons as the chairman of the meeting shall permit; (iv) restrictions on entry to the meeting after the time fixed for the commencement thereof; (v) limitations on the time allotted for consideration of each agenda item and for questions and comments by participants; (vi) regulations for the opening and closing of the polls for balloting and matters which are to be voted on by ballot (if any); and (vii) procedures (if any) requiring attendees to provide the Corporation advance notice of their intent to attend the meeting.

Section 1.11. <u>Business at Annual and Special Meetings</u>. No business (including director nominations) may be transacted at an annual or special meeting of stockholders other than business that is:

- (a) specified in a notice of meeting (or any supplement thereto) given by or at the direction of the Board of Directors or a duly authorized committee thereof,
- (b) otherwise brought before the meeting by or at the direction of the Board of Directors or a duly authorized committee thereof or any authorized officer of the Corporation to whom the Board of Directors or such committee shall have delegated such authority, or
- (c) otherwise brought before the meeting by a "Record Holder" who complies with the notice procedures set forth in Section 1.12 of these Bylaws.

A "<u>Record Holder</u>" is a stockholder of record at the time of the giving of the notice required in Section 1.12 and who is entitled to vote at the meeting. Clause (c) of this Section 1.11 shall be the exclusive means for a Record Holder to make director nominations or submit other business before a meeting of stockholders (other than proposals brought under Rule 14a-8 under the Securities Exchange Act of 1934, as amended (the "<u>Exchange Act</u>") and included in the Corporation's notice of meeting). Notwithstanding anything in these Bylaws to the contrary, no business shall be conducted at a stockholders' meeting except in accordance with the procedures set forth in Section 1.11 and Section 1.12 of these Bylaws.

Section 1.12. Notice of Stockholder Business and Nominations. In order for a Record Holder to properly bring any nominations or any item of business before a meeting of stockholders, the Record Holder must give timely notice thereof in writing to the Secretary of the Corporation in compliance with the requirements of this Section 1.12. This Section 1.12 shall constitute an "advance notice provision" for annual meetings for purposes of Rule 14a-4(c)(1) under the Exchange Act.

- (a) To be timely, a Record Holder's notice shall be delivered to the Secretary at the principal executive offices of the Corporation:
- (i) in the case of an annual meeting of stockholders, not earlier than the open of business on the one-hundred twentieth (120th) day and not later than the close of business on the ninetieth (90th) day prior to the first anniversary of the preceding year's annual meeting; provided, however, that in the event the date of the annual meeting is more than thirty (30) days before or more than sixty (60) days after such anniversary date, notice

by the stockholder to be timely must be so delivered not earlier than the open of business on the one-hundred twentieth (120th) day prior to the date of such annual meeting and not later than the close of business on the later of the ninetieth (90th) day prior to the date of such annual meeting or, if the first public announcement by the Corporation of the date of such annual meeting is less than one hundred (100) days prior to the date of such annual meeting, the tenth (10th) day following the day on which public announcement of the date of such meeting is first made by the Corporation;

- (ii) in the case of a special meeting of stockholders called for the purpose of electing directors, not earlier than the open of business on the one-hundred twentieth (120th) day prior to such special meeting and not later than the close of business on the later of the ninetieth (90th) day prior to such special meeting or the tenth (10th) day following the date on which notice of the date of the special meeting was made by the Corporation, mailed or public disclosure of the date of the special meeting was made, whichever first occurs; and
- (iii) in no event shall any adjournment or postponement of an annual or special meeting, or the announcement thereof, commence a new time period (or extend the time period) for the giving of a stockholder's notice as described above.
- (b) To be in proper form, whether in regard to a nominee for election to the Board of Directors or other business, a Record Holder's notice to the Secretary must:
 - (i) set forth, as to the Record Holder, the following information together with a representation as to the accuracy of the information:
 - (A) the name and address of the Record Holder as they appear on the Corporation's books and the names and addresses of any Stockholder Related Persons of such Record Holder;
 - (B) the class or series and number of shares of the Corporation that are, directly or indirectly, owned of record or beneficially owned (within the meaning of Rule 13d-3 under the Exchange Act) by the Record Holder or any Stockholder Associated Person of the Record Holder (except that such Record Holder or Stockholder Associated Person of the Record Holder shall in all events be deemed to beneficially own any shares of any class or series of the Corporation as to which such Record Holder or Stockholder Associated Person of the Record Holder has a right to acquire beneficial ownership at any time in the future) and the date such ownership was acquired;
 - (C) any option, warrant, convertible security, stock appreciation right, or similar right with an exercise or conversion privilege or a settlement payment or mechanism at a price related to any class or series of shares of the Corporation or with a value derived in whole or in part from the price, value or volatility of any class or series of shares of the Corporation, whether or not the instrument or right shall be subject to settlement in the underlying class or series of capital stock of the Corporation or otherwise (a "Derivative Instrument") that is directly or indirectly owned beneficially by the Record Holder or any Stockholder Associated Person of the Record Holder and any other direct or indirect opportunity to profit or share in any profit derived from any increase or decrease in the price, value or volatility of shares of the Corporation;

(D) any proxy, contract, arrangement, understanding or relationship pursuant to which the Record Holder or Stockholder Associated Person of the Record Holder has a right to vote or has granted a right to vote any shares of any security of the Corporation;

- (E) any short interest in any security of the Corporation (for purposes of these Bylaws a person shall be deemed to have a short interest in a security if the Record Holder or any Stockholder Associated Person of the Record Holder directly or indirectly, through any contract, arrangement, understanding, relationship or otherwise, has the opportunity to profit or share in any profit derived from any decrease in the value of the subject security);
- (F) any rights to dividends on the shares of any security of the Corporation owned beneficially by the Record Holder or any Stockholder Associated Person of the Record Holder that are separated or separable from the underlying shares of the Corporation;
- (G) any proportionate interest in shares of the Corporation or Derivative Instruments held, directly or indirectly, by a general or limited partnership or limited liability company or similar entity in which the Record Holder or any Stockholder Associated Person of the Record Holder is a general partner or, directly or indirectly, beneficially owns an interest in a general partner, is the manager, managing member or directly or indirectly beneficially owns an interest in the manager or managing member of a limited liability company or similar entity;
- (H) any performance-related fees (other than an asset-based fee) that the Record Holder or any Stockholder Associated Person of the Record Holder is entitled to based on any increase or decrease in the value of shares of the Corporation or Derivative Instruments or short interests, if any;
- (I) any arrangements, rights, or other interests described in Sections 1.12(b)(i)(C)-(H) held by members of such Record Holder's immediate family sharing the same household;
- (J) a representation that the Record Holder intends to appear in person or by proxy at the meeting to nominate the person(s) named or propose the business specified in the notice and whether or not such stockholder intends to deliver a proxy statement and/or form of proxy to holders of at least the percentage of the Corporation's outstanding shares required to approve the nomination(s) or the business proposed and/or otherwise to solicit proxies from stockholders in support of the nomination(s) or the business proposed;

- (K) a certification regarding whether or not such Record Holder and any Stockholder Associated Person of the Record Holder have complied with all applicable federal, state and other legal requirements in connection with such Record Holder's and/or Stockholder Associated Persons' acquisition of shares or other securities of the Corporation and/or such Record Holder's and/or Stockholder Associated Persons' acts or omissions as a stockholder of the Corporation;
- (L) any other information relating to the Record Holder and/or Stockholder Associated Person of the Record Holder that would be required to be disclosed in a proxy statement or other filings required to be made in connection with solicitations of proxies for, as applicable, the proposal and/or for the election of directors in a contested election pursuant to Section 14 of the Exchange Act and the rules and regulations thereunder; and
 - (M) any other information as reasonably requested by the Corporation.

Such information shall be provided as of the date of the notice and shall be supplemented by the Record Holder not later than ten (10) days after the record date for the meeting to disclose such ownership as of the record date.

- (ii) if the notice relates to any business other than a nomination of a director or directors that the stockholder proposes to bring before the meeting, the notice must set forth:
- (A) a reasonably detailed description of the business desired to be brought before the meeting (including the text of any resolutions proposed for consideration), the reasons for conducting such business at the meeting, and any material direct or indirect interest of the Record Holder or any Stockholder Associated Persons in such business; and
- (B) a reasonably detailed description of all agreements, arrangements and understandings, direct and indirect, between the Record Holder, and any other person or persons (including their names) in connection with the proposal of such business by the Record Holder.
- (iii) set forth, as to each person, if any, whom the Record Holder proposes to nominate for election or reelection to the Board of Directors:
- (A) all information with respect to such proposed nominee that would be required to be set forth in a Record Holder's notice pursuant to this Section 1.12 if such proposed nominee were a Record Holder;
- (B) all information relating to the nominee (including, without limitation, the nominee's name, age, business and residence address and principal occupation or employment and the class or series and number of shares of capital stock of the Corporation that are owned beneficially or of record by the nominee) that would be required to be disclosed in a proxy statement or other filings required to be made in connection with solicitations of proxies for election of directors in a contested election pursuant to Section 14 of the Exchange Act and the rules and regulations thereunder (including such person's written consent to being named in a proxy statement as a nominee and to serving as a director if elected);

(C) a description of any agreements, arrangements and understandings between or among such stockholder or any Stockholder Associated Person, on the one hand, and any other persons (including any Stockholder Associated Person), on the other hand, in connection with the nomination of such person for election as a director; and

(D) a description of all direct and indirect compensation and other material monetary agreements, arrangements, and understandings during the past three years, and any other material relationships, between or among the Record Holder and respective affiliates and associates, on the one hand, and each proposed nominee, and his or her respective affiliates and associates, on the other hand, including, without limitation all information that would be required to be disclosed pursuant to Item 404 of Regulation S-K if the Record Holder making the nomination or on whose behalf the nomination is made, if any, or any affiliate or associate thereof or person acting in concert therewith, were the "registrant" for purposes of Item 404 and the nominee were a director or executive officer of such registrant.

(iv) with respect to each nominee for election or reelection to the Board of Directors, the Record Holder shall include a completed and signed questionnaire, representation, and agreement required by Section 1.13 of these Bylaws. The Corporation may require any proposed nominee to furnish such other information as may reasonably be required by the Corporation to determine the eligibility of the proposed nominee to serve as an independent director of the Corporation or that could be material to a reasonable stockholder's understanding of the independence, or lack thereof, of the nominee. The number of nominees a Record Holder may nominate for election at an annual or special meeting (or in the case of Record Holder giving the notice on behalf of a beneficial owner, the number of nominees a Record Holder may nominate for election at the annual or special meeting on behalf of such beneficial owner) shall not exceed the number of directors to be elected at such annual or special meeting.

(c) For purposes of these Bylaws:

- (i) "Affiliate" and "Associate" are defined by reference to Rule 12b-2 under the Exchange Act. An "affiliate" is any "person that directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, the person specified."
- (ii) "Control" is defined as the "possession, direct or indirect, of the power to direct or cause the direction of the management policies of a person, whether through the ownership of voting securities, by contract, or otherwise."

- (iii) The term "<u>associate</u>" of a person means: (i) any corporation or organization (other than the registrant or a majority-owned subsidiary of the registrant) of which such person is an officer or partner or is, directly or indirectly, the beneficial owner of ten (10) percent or more of any class of equity securities, (ii) any trust or other estate in which such person has a substantial beneficial interest or as to which such person serves as trustee or in a similar fiduciary capacity, and (iii) any relative or spouse of such person, or any relative of such spouse, who has the same home as such person or who is a director or officer of the registrant or any of its parents or subsidiaries.
 - (iv) "GP LLCA" means the limited liability company agreement of TPG GP A, LLC, a Delaware limited liability company.
- (v) "public announcement" means disclosure in a press release reported by a national news service or in a document publicly filed by the Corporation with the Securities and Exchange Commission pursuant to Section 13, 14, or 15(d) of the Exchange Act and the rules and regulations thereunder.
- (vi) "<u>Stockholder Associated Person</u>" means, with respect to any stockholder, (A) any beneficial owner of shares of stock of the Corporation owned of record or beneficially by such stockholder (other than a stockholder that is a depositary) and (B) any person controlling, controlled by or under common control with any stockholder, or any Stockholder Associated Person identified in clause (A) above.
- (vii) The "<u>TPG OG Partnerships</u>" means TPG Operating Group I, L.P., TPG Operating Group II, L.P., and TPG Operating Group III, L.P., each a Delaware limited partnership.
- (d) Only those persons who are nominated in accordance with the procedures set forth in these Bylaws shall be eligible to serve as directors. Only such business shall be conducted at a meeting of stockholders as shall have been brought before the meeting in accordance with the procedures set forth in these Bylaws, provided, however, that, once business has been properly brought before the meeting in accordance with Section 1.12, nothing in this Section 1.12(d) shall be deemed to preclude discussion by any stockholder of such business. If any information submitted pursuant to this Section 1.12 by any stockholder, such information shall be deemed not to have been provided in accordance with Section 1.12. Notwithstanding the foregoing provisions of this Section 1.12, unless otherwise required by law, or otherwise determined by law or the Chair of the meeting, if the stockholder (or a qualified representative of the stockholder) does not appear at the annual or special meeting of stockholders to present a nomination or other business, such nomination shall be disregarded and such other business shall not be transacted, notwithstanding that proxies in respect of such vote may have been received by the Corporation. Except as otherwise provided by law, the Certificate of Incorporation, or these Bylaws, the Chair of the meeting shall have the power and duty to determine whether a nomination or any business proposed to be brought before the meeting was made or proposed, as the case may be, in compliance with the procedures set forth in these Bylaws and, if he or she should determine that any proposed nomination or business is not in compliance with these Bylaws, he or she shall so declare to the meeting and any such nomination or business not properly brought before the meeting shall be disregarded or not be transacted.

- (e) Notwithstanding the foregoing provisions of these Bylaws, a Record Holder also shall comply with all applicable requirements of the Exchange Act and the rules and regulations thereunder with respect to the matters set forth in these Bylaws; provided, however, that any references in these Bylaws to the Exchange Act or the rules thereunder are not intended to and shall not limit the requirements applicable to nominations or proposals as to any other business to be considered pursuant to Section 1.11 or Section 1.12 of these Bylaws.
- (f) Nothing in these Bylaws shall be deemed to (i) affect any rights of (A) stockholders to request inclusion of proposals in the Corporation's proxy statement pursuant to Rule 14a-8 under the Exchange Act or (B) the holders of any series of preferred stock, if any, if so provided under the Certificate of Incorporation, or (ii) affect any rights of a party to the GP LLCA.
- Section 1.13. <u>Submission of Questionnaire, Representation and Agreement</u>. To be eligible to be a nominee for election or reelection as a director of the Corporation by a Record Holder, a person must complete and deliver (in accordance with the time periods prescribed for delivery of notice under Section 1.12 of these Bylaws) to the Secretary at the principal executive offices of the Corporation a written questionnaire providing the information requested about the background and qualifications of such person and the background of any other person or entity on whose behalf the nomination is being made and a written representation and agreement (the questionnaire, representation, and agreement to be in the form provided by the Secretary upon written request) that such person:
 - (a) is not and will not become a party to:
 - (i) any agreement, arrangement or understanding with, and has not given any commitment or assurance to, any person or entity as to how the person, if elected as a director of the Corporation, will act or vote on any issue or question (a "<u>Voting Commitment</u>") that has not been disclosed to the Corporation, or
 - (ii) any Voting Commitment that could limit or interfere with the person's ability to comply, if elected as a director of the Corporation, with the person's fiduciary duties under applicable law,
- (b) is not and will not become a party to any agreement, arrangement or understanding with any person or entity other than the Corporation with respect to any direct or indirect compensation, reimbursement or indemnification in connection with service or action as a director that has not been disclosed to the Corporation, and
- (c) in the person's individual capacity and on behalf of any person or entity on whose behalf the nomination is being made, would be in compliance, if elected as a director of the Corporation, and will comply with all applicable publicly disclosed corporate governance, conflict of interest, confidentiality and stock ownership and trading policies and guidelines of the Corporation.
- Section 1.14. <u>Inspectors of Elections</u>. Preceding any meeting of the stockholders, if required by law, the Board of Directors shall appoint one (1) or more persons to act as "inspectors" of elections, and may designate one (1) or more alternate inspectors. In the event no inspector or alternate is able to act, the Chair of such meeting shall appoint one (1) or more inspectors to act at the meeting. Each inspector, before entering upon the discharge of the

duties of an inspector, shall take and sign an oath faithfully to execute the duties of inspector with strict impartiality and according to the best of his or her ability. The inspector shall: ascertain the number of shares outstanding and the voting power of each; determine the shares represented at a meeting and the validity of proxies and ballots; count all votes and ballots; determine and retain for a reasonable period a record of the disposition of any challenges made to any determination by the inspectors; and certify his or her determination of the number of shares represented at the meeting, and his or her count of all votes and ballots.

Section 1.15. List of Stockholders Entitled to Vote. The Corporation shall prepare, at least ten (10) days before every meeting of the stockholders, a complete list of the stockholders entitled to vote at the meeting; provided, however, that if the record date for determining the stockholders entitled to vote is less than 10 days before the date of the meeting, the list shall reflect the stockholders entitled to vote as of the 10th day before the meeting date. Such list shall be arranged in alphabetical order and shall show the address of each stockholder and the number of shares registered in the name of each stockholder. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting, during ordinary business hours, for a period of at least ten (10) days prior to the meeting either (i) on a reasonably accessible electronic network, provided that the information required to gain access to such list is provided with the notice of the meeting, or (ii) during ordinary business hours, at the Corporation's principal executive office. In the event that the Corporation determines to make the list available on an electronic network, the Corporation may take reasonable steps to ensure that such information is available only to stockholders of the Corporation. If the meeting is to be held at a place, then the list shall be produced and kept at the time and place of the meeting during the whole time thereof, and may be inspected by any stockholder who is present. If the meeting is to be held solely by means of remote communication, then the list shall also be open to the examination of any stockholder during the whole time of the meeting on a reasonably accessible electronic network, and the information required to access such list shall be provided with the notice of the meeting. Except as otherwise provided by law, such list shall be the only evidence as to who are the stockholders entitled to examine the list of stockholders required by this Section 1.16 or to vote in person or by

Section 1.16. Meetings by Remote Communications. The Board of Directors may, in its sole discretion, determine that a meeting of stockholders shall not be held at any place, but may instead be held solely by means of remote communication in accordance with Section 211(a)(2) of the DGCL. If authorized by the Board of Directors in its sole discretion, and subject to such guidelines and procedures as the Board of Directors may authorize, stockholders and proxyholders not physically present at a meeting of stockholders may, by means of remote communication: (a) participate in a meeting of stockholders; and (b) be deemed present in person and vote at a meeting of stockholders whether such meeting is to be held at a designated place or solely by means of remote communication, provided that: (i) the Corporation shall implement reasonable measures to verify that each person deemed present and permitted to vote at the meeting by means of remote communication is a stockholder or proxyholder; (ii) the Corporation shall implement reasonable measures to provide such stockholders and proxyholders a reasonable opportunity to participate in the meeting and to vote on matters submitted to the stockholders, including an opportunity to read or hear the proceedings of the meeting substantially concurrently with such proceedings; and (iii) if any stockholder or proxyholder votes or takes other action at the meeting by means of remote communication, a record of such vote or other action shall be maintained by the Corporation.

ARTICLE II BOARD OF DIRECTORS

Section 2.01. General Powers. From and after the Sunset (as defined in the Certificate of Incorporation), the business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors and the provisions of Section 5.8 of the Certificate of Incorporation shall be of no force or effect. Prior to the Sunset, the business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors, the Executive Committee (as defined in the Certificate of Incorporation) and the Other Committees (as defined in the Certificate of Incorporation), each with the power and authority conferred pursuant to, or in the manner contemplated by, Section 5.8 of the Certificate of Incorporation. Prior to the Sunset, except as otherwise provided in Section 5.8 of the Certificate of Incorporation (including with respect to Exclusive Board Matters and Exclusive Committee Matters (as such terms are defined in the Certificate of Incorporation)), the Executive Committee shall exclusively have all power and authority conferred or imposed upon a board of directors by the DGCL or upon the Board of Directors by the Certificate of Incorporation or these Bylaws. Except as otherwise provided in Section 5.8 of the Certificate of Incorporation, the provisions of these Bylaws that apply to directors and the Board of Directors shall apply to the Executive Committee (and the directors who comprise the Executive Committee), and Other Committees (and the directors who comprise the Other Committees).

Section 2.02. Number, Election and Qualification. Subject to any Preferred Designation, and to the limitations and requirements set forth in the GP LLCA, the total number of directors constituting the Board of Directors shall be at least one, or such larger number as may be determined from time to time exclusively by resolution adopted by the Board of Directors. No reduction of the authorized number of directors shall have the effect of removing any director before that director's term of office expires. At any meeting of stockholders at which directors are to be elected, directors shall be elected by the plurality of the votes cast by the holders of shares present in person or represented by proxy at the meeting and entitled to vote thereon. Directors need not be stockholders of the Corporation.

Section 2.03. Chair of the Board. Unless the Board of Directors otherwise provides, subject to the control vested in the Board of Directors (or its delegates) by statute, by the Certificate of Incorporation, or by these Bylaws, the Executive Chairman (or the Chair, as applicable) shall, if present, preside over all meetings of the stockholders and of the Board of Directors and shall have such other duties and powers as from time to time may be assigned to him or her by the Board of Directors, the Certificate of Incorporation or these Bylaws. Unless the Board of Directors otherwise provides, in the absence (or inability or refusal to act) of the Chair of the Board, the Chief Executive Officer (if such person shall be a director) or such other director or officer of the Corporation designated by the Chair of the Board, shall preside when present at all meetings of the stockholders and the Board of Directors.

Section 2.04. <u>Founders</u>. David Bonderman and James Coulter are the Company Founders. For so long as Mr. Coulter is a member of the Board of Directors, Mr. Coulter shall be the Chair of the Board; provided, that, for so long as Mr. Coulter remains employed by or otherwise provides services to the Corporation on a substantially full-time basis, he shall be the Executive Chairman. For so long as Mr. Bonderman is a member of the Board of Directors, Mr. Bonderman shall be Non-Executive Chairman of the Board. Each of the Founders shall continue to hold the title of Founder after they cease to be members of the Board of Directors.

Section 2.05. <u>Annual and Regular Meetings</u>. The annual meeting of the Board of Directors for the purpose of electing officers and for the transaction of such other business as may come before the meeting shall be held after the annual meeting of the stockholders and may be held at such places within or without the State of Delaware and at such times as the Board may from time to time determine, and if so determined notice thereof need not be given. Notice of such annual meeting of the Board of Directors need not be given. The Board of Directors from time to time may by resolution provide for the holding of regular meetings and fix the place (which may be within or without the State of Delaware) and the date and time of such meetings. Notice of regular meetings need not be given, provided, however, that if the Board of Directors shall fix or change the time or place of any regular meeting, notice of such action shall be mailed promptly, or sent by telephone, including a voice messaging system or other system or technology designed to record and communicate messages, telegraph, electronic mail or other electronic means, to each director.

Section 2.06. Special Meetings; Notice. Special meetings of the Board of Directors for any purpose or purposes shall be held whenever called by the Executive Chairman (or Chair, as applicable), the Non-Executive Chairman, the Chief Executive Officer or by any two (2) directors, at such place (within or without the State of Delaware), date and time as may be specified in the notices of such meetings. Notice shall be duly given to each director (a) in person or by telephone at least twenty-four (24) hours in advance of the meeting, (b) by sending written notice by reputable overnight courier, or electronic transmission, or delivering written notice by hand, to such director's address at least twenty-four (24) hours in advance of the meeting, or (c) by sending written notice by first-class mail to such director's business or to such other address as any director may request by notice to the Secretary at least seventy-two (72) hours in advance of the meeting. Attendance of director at a meeting shall constitute a waiver of notice of such meeting, except when the person attends a meeting for the express purpose of objecting at the beginning of the meeting, to the transaction of any business because the meeting is not lawfully called or convened.

Section 2.07. <u>Quorum; Voting</u>. At all meetings of the Board of Directors, the presence of at least a majority of the total number of directors shall constitute a quorum for the transaction of business. Except as otherwise required by law, the Certificate of Incorporation or these Bylaws, the vote of at least a majority of the directors present at any meeting at which a quorum is present shall be the act of the Board of Directors.

Section 2.08. <u>Adjournment</u>. A majority of the directors present, whether or not a quorum is present, may adjourn any meeting of the Board of Directors to another time or place.

Section 2.09. <u>Action Without a Meeting</u>. Any action required or permitted to be taken at any meeting of the Board of Directors, or any committee thereof, may be taken without a meeting if all members of the Board of Directors or committee, as the case may be, consent thereto in writing or by electronic transmission. After the action is taken, the consent or consents relating thereto shall be filed with the minutes of proceedings of the Board of Directors or committee.

Section 2.10. <u>Regulations</u>; <u>Manner of Acting</u>. To the extent consistent with applicable law, the Certificate of Incorporation and these Bylaws, the Board of Directors may adopt by resolution such rules and regulations for the conduct of meetings of the Board of Directors and for the management of the property, affairs and business of the Corporation as the Board of Directors may deem appropriate. The directors shall act only as a Board of Directors and the individual directors shall have no power in their individual capacities unless expressly authorized by the Board of Directors.

Section 2.11. <u>Action by Telephonic Communications</u>. Members of the Board of Directors, or any committee thereof, may participate in a meeting of the Board or committee by means of conference telephone or other communications equipment by means of which all persons participating in the meeting can hear and communicate with each other, and participation in a meeting pursuant to this provision shall constitute presence in person at such meeting.

Section 2.12. <u>Resignations</u>. Any director may resign at any time by submitting an electronic transmission or by delivering a written notice of resignation, signed by such director, to the Corporation. Unless otherwise specified therein, such resignation shall take effect upon delivery.

Section 2.13. <u>Removal of Directors</u>. Subject to any Preferred Designation, and to the limitations and requirements set forth in the GP LLCA with respect to the parties thereto, any director may be removed at any time by the affirmative vote of the holders of at least a majority of the voting power of the Corporation's outstanding shares of stock entitled to vote generally in the election of directors, voting together as a single class, with or without cause.

Section 2.14. <u>Vacancies and Newly Created Directorships</u>. Subject to any Preferred Designation, and to the limitations and requirements set forth in the GP LLCA, any vacancies in the Board of Directors for any reason and any newly created directorships resulting by reason of any increase in the number of directors shall be filled only by the Board of Directors acting by a majority of the directors then in office, even if less than a quorum, or by a sole remaining director, and any directors so appointed shall hold office until the next election of directors and until their successors are duly elected and qualified, provided that, prior to the Sunset (as defined in the Certificate of Incorporation), such vacancies may also be filled by a majority of the voting power of the Corporation's outstanding shares of stock entitled to vote generally in the election of directors, voting together as a single class.

Section 2.15. <u>Compensation</u>. The amount, if any, which each director shall be entitled to receive as compensation for such director's services, shall be fixed from time to time by resolution of the Board of Directors or any committee thereof or as an agreement between the Corporation and any director. The directors may be reimbursed their out-of-pocket expenses, if any, of attendance at each meeting of the Board of Directors in accordance with the Corporation's policies in effect from time to time and may be paid a fixed sum for attendance at each meeting of the Board of Directors or a stated salary for service as director, payable in cash or securities. No such payment shall preclude any director from serving the Corporation in any other capacity and receiving compensation therefor. Members of special or standing committees may be allowed compensation and reimbursement for service as committee members as determined by the Board or a committee thereof.

Section 2.16. Reliance on Accounts and Reports, Etc. A director, or a member of any committee designated by the Board of Directors, shall, in the performance of such director's or member's duties, be fully protected in relying in good faith upon the records of the Corporation and upon information, opinions, reports or statements presented to the Corporation by any of the Corporation's officers or employees, or committees designated by the Board of Directors, or by any other person as to the matters the director or the member reasonably believes are within such other person's professional or expert competence and who the director or member reasonably believes or determines has been selected with reasonable care by or on behalf of the Corporation.

ARTICLE III COMMITTEES

Section 3.01. <u>Committees</u>. The Board of Directors, by resolution, may designate from among its members one (1) or more committees of the Board of Directors, each consisting of one or more directors as from time to time may be fixed by the Board of Directors. Any such committee shall serve at the pleasure of the Board of Directors. The Board of Directors or the committee may appoint a Chair of any committee, who shall preside at meetings of any such committee.

Section 3.02. <u>Powers</u>. Subject to any limitation imposed by applicable law, each committee shall have and may exercise such powers as may be provided by the Board of Directors or provided in the charter of such committee. No committee shall have the power or authority to approve or adopt, or recommend to the stockholders, any action or matter (other than the election or removal of directors) expressly required by the DGCL to be submitted by the Board of Directors to the stockholders for approval; or to adopt, amend or repeal the Bylaws of the Corporation.

Section 3.03. <u>Proceedings</u>. Except as otherwise provided herein, in charter of such committee or required by law, each committee may fix its own rules of procedure and may meet at such place (within or without the State of Delaware), at such time and upon such notice, if any, as it shall determine from time to time. Each committee shall keep minutes of its proceedings and shall report such proceedings to the Board of Directors if requested by the Board of Directors.

Section 3.04. Quorum and Manner of Acting. Except as may be otherwise provided in the resolution creating such committee or in the rules or charter of such committee, at all meetings of any committee, the presence of members constituting a majority of the total number of committee members serving shall constitute a quorum for the transaction of business, except that, in the case of one-member committees, the presence of one member shall constitute a quorum and in the case of two-member committees, the presence of two members shall constitute a quorum. The act of the majority of the members present at any meeting at which a quorum is present shall be the act of such committee. Any action required or permitted to be taken at any meeting of any committee may be taken without a meeting, in accordance with Section 2.09 of these Bylaws. The members of any committee shall act only as a committee, and the individual members of such committee shall have no power in their individual capacities unless expressly authorized by the Board of Directors or the committee.

Section 3.05. <u>Action by Telephonic Communications</u>. Unless otherwise provided by the Board of Directors, members of any committee may participate in a meeting of such committee by means of conference telephone or other communications equipment by means of which all persons participating in the meeting can hear and communicate with each other, and participation in a meeting pursuant to this provision shall constitute presence in person at such meeting.

Section 3.06. <u>Resignations</u>. Any member of any committee may resign at any time by submitting an electronic transmission or by delivering a written notice of resignation, signed by such member, to the Board of Directors or the Executive Chairman (or Chair, as applicable). Unless otherwise specified therein, such resignation shall take effect upon delivery.

Section 3.07. <u>Removal</u>. Any member of any committee may be removed at any time, either for or without cause, by resolution adopted by the Board of Directors.

Section 3.08. <u>Vacancies</u>. If any vacancy shall occur in any committee, by reason of disqualification, death, resignation, removal or otherwise, the remaining members shall continue to act, and any such vacancy may be filled by the Board of Directors.

ARTICLE IV OFFICERS

Section 4.01. <u>Appointment, Selection and Designation of Officers Other than the Chief Executive Officer</u>. The Chief Executive Officer may, from time to time as they deem advisable, select natural persons and designate them as officers of the Corporation (together with the chief executive officer, the "Officers") and assign titles (including, without limitation, "chief operating officer," "chief financial officer," "general counsel," "chief administrative officer," "chief compliance officer," "principal accounting officer," "chairman," "senior chairman," "executive vice chairman," "vice chairman," "president," "treasurer," "assistant treasurer," "secretary," "assistant secretary," "general manager," "senior managing director," "managing director" and "director") to any such persons. An Officer may be removed with or without cause by the Board of Directors. Any vacancies occurring in any office other than the office of Chief Executive Officer may be filled by the Chief Executive Officer in the same manner as such officers are appointed and selected pursuant to this Section 4.01.

Section 4.02. <u>Delegation of Duties</u>. Unless the Board of Directors determines otherwise, if a title is one commonly used for officers of a corporation formed under the DGCL, the assignment of such title shall constitute the delegation to such Person of the authorities and duties that are normally associated with that office. The Board of Directors may delegate to any officer any of the Board of Director's powers to the extent permitted by applicable law, including the power to bind the Corporation. Any delegation pursuant to this Section 4.02 may be revoked at any time by the Board of Directors.

Section 4.03. <u>Chief Executive Officer</u>. The Chief Executive Officer shall have such authority and responsibilities as set forth in the Corporation's employment agreement of the Chief Executive Officer in effect at such time and shall possess such other powers and perform such other duties as may be assigned to the Chief Executive Officer by these Bylaws, as may from time to time be assigned by the Board of Directors and as may be incident to the office of Chief Executive Officer of the Corporation. The Chief Executive Officer shall have general authority to execute bonds, deeds and contracts in the name of the Corporation, except where required or permitted by law to be otherwise signed and executed and except that the other officers of the Corporation may sign and execute documents when so authorized by these Bylaws, the Board of Directors or the Chief Executive Officer.

Section 4.04. <u>Secretary</u>. The Secretary shall issue all authorized notices for, and shall keep minutes of, all meetings of the stockholders and the Board of Directors. He or she shall have charge of the corporate books and shall perform such other duties as the Board of Directors may from time to time prescribe.

Section 4.05. <u>Term of Office</u>. Each officer shall hold office until his or her successor shall have been duly elected and shall have qualified or until his or her death or until he or she shall resign, but, subject to the requirements of the Certificate of Incorporation, any officer may be removed pursuant to the provisions set forth in Section 4.01.

Section 4.06. <u>Authority and Duties of Officers</u>. The officers of the Corporation shall have such authority and shall exercise such powers and perform such duties as may be specified in these Bylaws or pursuant to authorization of the Board of Directors, or which generally pertain to such officer's title and each officer shall exercise such powers and perform such duties as may be required by law.

ARTICLE V CAPITAL STOCK

Section 5.01. <u>Certificates of Stock</u>. The Board of Directors may authorize that some or all of the shares of any or all of the Corporation's classes or series of stock be evidenced by a certificate or certificates of stock. The Board of Directors may also authorize the issue of some or all of the shares of any or all of the Corporation's classes or series of stock without certificates. The rights and obligations of stockholders with the same class and/or series of stock shall be identical whether or not their shares are represented by certificates.

(a) Shares with Certificates. If the Board of Directors chooses to issue shares of stock evidenced by a certificate or certificates, each individual certificate shall include the following on its face: (i) the Corporation's name, (ii) the fact that the Corporation is organized under the laws of Delaware, (iii) the name of the person to whom the certificate is issued, (iv) the number of shares represented thereby, (v) the class of shares and the designation of the series, if any, which the certificate represents, and (vi) such other information as applicable law may require or as may be lawful. If the Corporation is authorized to issue different classes of shares or different

series within a class, the designations, relative rights, preferences and limitations determined for each class or series (and the authority of the Board of Directors to determine variations for future series) shall be summarized on the front or back of each certificate. Alternatively, each certificate shall state on its front or back that the Corporation will furnish the stockholder this information in writing, without charge, upon request. Each certificate of stock issued by the Corporation shall be signed by any two authorized officers of the Corporation, including the Chief Executive Officer, the Secretary and the Chief Financial Officer. If the person who signed a certificate no longer holds office when the certificate is issued, the certificate is nonetheless valid.

(b) Shares without Certificates. If the Board of Directors chooses to issue shares of stock without certificates, the Corporation, shall, within a reasonable time after the issue or transfer of shares without certificates, send the stockholder a written notice containing the information required to be set forth or stated on certificates pursuant to the laws of the DGCL. The Corporation may adopt a system of issuance, recordation and transfer of its shares of stock by electronic or other means not involving the issuance of certificates, provided the use of such system by the Corporation is permitted in accordance with applicable law.

Section 5.02. <u>Signatures</u>. All signatures on the certificate referred to in Section 5.01 of these Bylaws may be a facsimile, to the extent permitted by law. In case any officer, transfer agent or registrar who has signed, or whose facsimile signature has been placed upon a certificate, shall have ceased to be such officer, transfer agent or registrar before such certificate is issued, it may be issued by the Corporation with the same effect as if he or she were such officer, transfer agent or registrar at the date of issue.

Section 5.03. <u>Lost, Stolen or Destroyed Certificates</u>. Except as provided in this Section 5.03, no new share certificates for shares shall be issued to replace a previously issued certificate unless the latter is surrendered to the Corporation and cancelled at the same time. The Corporation may direct that a new certificate be issued in place of any certificate theretofore issued by the Corporation alleged to have been lost, stolen or destroyed, upon delivery to the Corporation of an affidavit (or other document acceptable to the Corporation) of the owner or owners of such certificate, setting forth such allegation. The Corporation may require the owner of such lost, stolen or destroyed certificate, or his or her legal representative, to give the Corporation a bond (or other security, including an indemnification agreement) sufficient to indemnify it against any claim that may be made against it on account of the alleged loss, theft or destruction of any such certificate or the issuance of any such new certificate.

Section 5.04. <u>Transfer of Stock</u>. Upon surrender to the Corporation or the transfer agent of the Corporation of a certificate for shares, duly endorsed or accompanied by appropriate evidence of succession, assignment or authority to transfer, the Corporation shall issue a new certificate to the person entitled thereto, cancel the old certificate and record the transaction upon its books. Within a reasonable time after the transfer of uncertificated stock, the Corporation shall send to the registered owner thereof a written notice containing the information required to be set forth or stated on certificates pursuant to the laws of the DGCL. Subject to the provisions of the Certificate of Incorporation and these Bylaws, the Board of Directors may prescribe such additional rules and regulations as it may deem appropriate relating to the issue, transfer and registration of shares of the Corporation. Except as otherwise required by law, no transfer of stock shall be valid against the Corporation for any purpose until it shall have been entered in the stock records of the Corporation by an entry showing the names of the persons from and to whom it was transferred.

Section 5.05. Record Date.

(a) In order to determine the stockholders entitled to notice of any meeting of stockholders or any adjournment thereof, the Board of Directors may fix, in advance, a record date, which record date shall not precede the date on which the resolution fixing the record date is adopted by the Board of Directors, and which shall not be more than sixty (60) nor fewer than ten (10) days before the date of such meeting. If the Board of Directors so fixes a date, such date shall also be the record date for determining the stockholders entitled to vote at such meeting unless the Board of Directors determines, at the time it fixes such record date, that a later date on or before the date of the meeting shall be the date for making such determination. If no record date is fixed by the Board of Directors, the record date for determining stockholders entitled to notice of and to vote at a meeting of stockholders shall be at the close of business on the day next preceding the day on which the meeting is held. A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting, provided, however, that the Board of Directors may fix a new record date for the determination of stockholders entitled to vote at a adjourned meeting, and in such case shall also fix as the record date for stockholders entitled to notice of such adjourned meeting the same or an earlier date as that fixed for determination of stockholders entitled to vote in accordance herewith at the adjourned meeting.

(b) In order that the Corporation may determine the stockholders entitled to receive payment of any dividend or other distribution or allotment of any rights of the stockholders entitled to exercise any rights in respect of any change, conversion or exchange of stock, or for the purpose of any other lawful action, the Board of Directors may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted, and which record date shall be not more than sixty (60) days prior to such action. If no record date is fixed, the record date for determining stockholders for any such purpose shall be at the close of business on the day on which the Board of Directors adopts the resolution relating thereto.

(c) Unless otherwise restricted by the Certificate of Incorporation, in order that the Corporation may determine the stockholders entitled to consent to corporate action without a meeting, the Board of Directors may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board of Directors, and which record date shall not be more than ten (10) days after the date upon which the resolution fixing the record date is adopted by the Board of Directors. If no record date for determining stockholders entitled to consent to corporate action without a meeting is fixed by the Board of Directors, (i) when no prior action of the Board of Directors is required by law, the record date for such purpose shall be the first date on which a signed consent setting forth the action taken or proposed to be taken is delivered to the Corporation in accordance with applicable law, and (ii) if prior action by the Board of Directors is required by law, the record date for such purpose shall be at the close of business on the day on which the Board of Directors adopts the resolution taking such prior action.

Section 5.06. <u>Registered Stockholders</u>. The Corporation may treat the registered owner as the person exclusively entitled to receive dividends and other distributions, to vote, to receive notice and otherwise to exercise all the rights and powers of the owner of the shares represented by such certificate.

Section 5.07. <u>Transfer Agent and Registrar</u>. The Board of Directors may appoint one (1) or more transfer agents and one (1) or more registrars, and may require all certificates representing shares to bear the signature of any such transfer agents or registrars.

ARTICLE VI INDEMNIFICATION

Section 6.01. <u>Indemnification and Advancement of Expenses</u>. The Corporation shall indemnify and provide advancement to any Indemnitee (as defined below) to the fullest extent permitted by law (as such may be amended from time to time), subject only to the limitations set forth in Section 6.07. The rights to indemnification and advancement conferred in this Section shall be contract rights. In furtherance of the foregoing indemnification and advancement obligations, and without limiting the generality thereof:

(a) <u>Proceedings Other Than Proceedings by or in the Right of the Corporation</u>. Any Indemnitee shall be entitled to the rights of indemnification and advancement provided in this Section 6.01(a) if, by reason of his or her Corporate Status (as defined below), Indemnitee was, is, will or might be involved as a party, potential party, non-party witness or otherwise in any Proceeding other than a Proceeding by or in the right of the Corporation. Pursuant to this Section 6.01(a), any Indemnitee shall be indemnified against all Expenses, judgments, penalties, fines and amounts paid in settlement actually and reasonably incurred by Indemnitee (or on Indemnitee's behalf) in connection with such Proceeding if Indemnitee acted in good faith and in a manner Indemnitee reasonably believed to be in or not opposed to the best interests of the Corporation, and with respect to any criminal Proceeding, had no reasonable cause to believe Indemnitee's conduct was unlawful. The termination of any Proceeding by judgment, order, settlement, conviction, or upon a plea of *nolo contendere* or its equivalent, shall not, of itself, create a presumption that Indemnitee did not act in good faith and in a manner which Indemnitee reasonably believed to be in or not opposed to the best interests of the Corporation, and, with respect to any criminal action or proceeding, had reasonable cause to believe that Indemnitee's conduct was unlawful.

(b) Proceedings by or in the Right of the Corporation. Any Indemnitee shall be entitled to the rights of indemnification and advancement provided in this Section 6.01(b) if, by reason of his or her Corporate Status, Indemnitee was, is, will or might be involved as a party, potential party, non-party witness or otherwise in any Proceeding other than a Proceeding by or in the right of the Corporation. Pursuant to this Section 6.01(b), any Indemnitee shall be indemnified against all Expenses actually and reasonably incurred by Indemnitee (or on Indemnitee's behalf) in connection with such Proceeding if Indemnitee acted in good faith and in a manner Indemnitee reasonably believed to be in or not opposed to the best interests of the Corporation; provided, however, if applicable law so provides, no indemnification against such Expenses shall be made in respect of any claim, issue or matter in such Proceeding as to which Indemnitee shall have been finally adjudged to be liable to the Corporation unless and to the extent that the Court of Chancery of the State of Delaware or the court in which such Proceeding was brought shall determine that such indemnification may be made.

(c) Other Sources. The Corporation hereby acknowledges that Indemnitees may have certain rights to indemnification, advancement of expenses or insurance provided by sources other than the Corporation ("Third Party Indemnitors"). The Corporation hereby agrees (i) that it is the indemnitor of first resort (i.e., its obligations to the Indemnitees are primary and any obligation of the Third Party Indemnitors to advance expenses or to provide indemnification for the same expenses or liabilities incurred by the Indemnitees are secondary), (ii) that it shall be required to advance the full amount of Expenses incurred by the Indemnitees (or on Indemnitees' behalf) and shall be liable for the full amount of all Expenses, judgments, penalties, fines and amounts paid in settlement by reason of such Indemnitee's Corporate Status to the extent legally permitted and as required by the terms of this paragraph and the Bylaws of the Corporation from time to time (or any other agreement between the Corporation and the Indemnitees), without regard to any rights the Indemnitees may have against the Third Party Indemnitors and (iii) that it irrevocably waives, relinquishes and releases the Third Party Indemnitors from any and all claims against the Third Party Indemnitors for contribution, subrogation or any other recovery of any kind in respect thereof. The Corporation further agrees that no advancement or payment by the Third Party Indemnitors on behalf of the Indemnitees with respect to any claim for which the Indemnitees have sought indemnification from the Corporation shall affect the foregoing and the Third Party Indemnitors shall have a right of contribution or to be subrogated to the extent of such advancement or payment to all of the rights of recovery of the Indemnitees against the Corporation. The Third Party Indemnitors are express third party beneficiaries of the terms of this paragraph.

Section 6.02. <u>Indemnification for Expenses of a Party Who is Wholly or Partly Successful</u>. Notwithstanding any other provision of this Article VI, to the extent that any Indemnitee is, by reason of his or her Corporate Status, a party to (or participant in) and is successful, on the merits or otherwise, in any Proceeding, or in defense of any claim, issue or matter therein, in whole or in part, Indemnitee shall be indemnified to the maximum extent permitted by law, as such may be amended from time to time, against all Expenses actually and reasonably incurred by Indemnitee or on Indemnitee's behalf in connection therewith. If such Indemnitee is not wholly successful in such Proceeding but is successful, on the merits or otherwise, as to one or more but less than all claims, issues or matters in such Proceeding, the Corporation shall, to the fullest extent permitted by applicable law, indemnify Indemnitee against all Expenses actually and reasonably incurred by him or her or on his or her behalf in connection with each successfully resolved claim, issue or matter. For purposes of this Section 6.02 and without limitation, the termination of any claim, issue or matter in such a Proceeding by dismissal, with or without prejudice, shall be deemed to be a successful result as to such claim, issue or matter.

Section 6.03. Employees and Agents. This Article VI shall not limit the right or ability of the Corporation, to the extent and in the manner permitted by law, to indemnify and to advance expenses to persons other than Indemnitees when and as authorized by appropriate corporate action. Moreover, nothing in this Article VI shall limit the right or ability of the Corporation, to the extent and in the manner permitted by law, to indemnify and to advance expenses to officers of the Corporation. Without limiting the generality of the foregoing, the Corporation may, to the extent authorized from time to time by the Board of Directors, provide rights to indemnification and advancement of expenses to employees and agents of the Corporation.

Section 6.04. <u>Advancement of Expenses</u>. Notwithstanding any other provision of this Article VI, the Corporation shall advance all Expenses incurred by or on behalf of any Indemnitee in connection with any Proceeding by reason of Indemnitee's Corporate Status within thirty (30) days after the receipt by the Corporation of a statement or statements from Indemnitee requesting such advance or advances from time to time, whether prior to or after final disposition of such Proceeding, and regardless of such Indemnitee's ability to repay any such amounts in the event of an ultimate determination that Indemnitee is not entitled thereto. Such statement or statements shall reasonably evidence the Expenses incurred by Indemnitee and shall, to the extent required by law, include or be preceded or accompanied by a written undertaking by or on behalf of Indemnitee to repay any Expenses advanced if it shall ultimately be determined that Indemnitee is not entitled to be indemnified against such Expenses. Any advances and undertakings to repay pursuant to this Section 6.04 shall be unsecured and interest free.

Section 6.05. Non-Exclusivity. The rights to indemnification and to the payment of Expenses incurred in defending a Proceeding in advance of the final disposition of such Proceeding conferred in this Article VI shall not be exclusive of any other rights which any person may have or hereafter acquire under applicable law, the Certificate of Incorporation, these Bylaws, any agreement, vote of stockholders, resolution of directors or otherwise. The assertion or employment of any right or remedy in this Article VI, or otherwise, shall not prevent the concurrent assertion or employment of any other right or remedy.

Section 6.06. <u>Insurance</u>. The Corporation shall have the power to purchase and maintain insurance, at its expense, to the fullest extent permitted by law, as such may be amended from time to time. Without limiting the generality of the foregoing, the Corporation shall have the power to purchase and maintain insurance on behalf of any person who is or was or has agreed to become a director, officer, employee or agent of the Corporation, or who is serving, was serving, or has agreed to serve at the request of the Corporation as a director, officer, trustee, general partner, managing member, fiduciary, employee or agent of any other Enterprise, against any liability asserted against him or her and incurred by him or her or on his or her behalf in such capacity, or arising out of his or her status as such, whether or not the Corporation would have the power to indemnify him or her against such liability.

Section 6.07. Exception to Rights of Indemnification and Advancement. Notwithstanding any provision in this Article VI, the Corporation shall not be obligated by this Article VI to make any indemnity or advancement in connection with any claim made against an Indemnitee:

(a) subject to Section 6.01(c), for which payment has actually been made to or on behalf of such Indemnitee under any insurance policy or other indemnity provision, except with respect to any excess beyond the amount paid under any insurance policy or other indemnity provision;

(b) for an accounting of profits made from the purchase and sale (or sale and purchase) by such Indemnitee of securities of the Corporation within the meaning of Section 16(b) of the Exchange Act or similar provisions of state statutory law or common law;

- (c) for reimbursement to the Corporation of any bonus or other incentive-based or equity based compensation or of any profits realized by Indemnitee from the sale of securities of the Corporation in each case as required under the Exchange Act; or
- (d) in connection with any Proceeding (or any part of any Proceeding) initiated by such Indemnitee, including any Proceeding (or any part of any Proceeding) initiated by such Indemnitee against the Corporation or its directors, officers, employees or other Indemnitees, unless (i) the Corporation has joined in or, prior to such Proceeding's initiation, the Board of Directors authorized such Proceeding (or any part of such Proceeding), (ii) the Corporation provides the indemnification or advancement, in its sole discretion, pursuant to the powers vested in the Corporation under applicable law, or (iii) the Proceeding is one to enforce such Indemnitee's rights under this Article VI or any other indemnification, advancement or exculpation rights to which Indemnitee may at any time be entitled under applicable law or any agreement.

Section 6.08. <u>Definitions</u>. For purposes of this Article VI:

- (a) "Corporate Status" describes the status of an individual who is or was or has agreed to become a director or officer of the Corporation or while an officer or director of the Corporation who is serving, was serving, or has agreed to serve at the request of the Corporation as a director, officer, trustee, general partner, managing member, fiduciary, employee or agent of any other Enterprise.
- (b) "Enterprise" shall mean the Corporation and any other corporation, constituent corporation (including any constituent of a constituent) absorbed in a consolidation or merger to which the Corporation (or any of their wholly owned subsidiaries) is a party, limited liability company, partnership, joint venture, trust, employee benefit plan or other enterprise, of which Indemnitee is or was serving at the request of the Corporation as a director, officer, trustee, general partner, managing member, fiduciary, employee or agent.
- (c) "Expenses" shall include all direct and indirect costs, fees and expenses of any type or nature whatsoever, including, without limitation, all attorneys' fees and costs, retainers, court costs, transcript costs, fees of experts, witness fees, travel expenses, fees of private investigators and professional advisors, duplicating costs, printing and binding costs, telephone charges, postage, delivery service fees, fax transmission charges, secretarial services, any federal, state, local or foreign taxes imposed on Indemnitee as a result of the actual or deemed receipt of any payments under this Article VI, ERISA excise taxes and penalties, and all other disbursements, obligations or expenses in connection with prosecuting, defending, preparing to prosecute or defend, investigating, being or preparing to be a witness in, settlement or appeal of, or otherwise participating in, a Proceeding, including, without limitation, reasonable compensation for time spent by the Indemnitee for which he or she is not otherwise compensated by the Corporation or any third party. Expenses also shall include Expenses incurred in connection with any appeal resulting from any Proceeding, including without limitation the principal, premium, security for, and other costs relating to any cost bond, supersedes bond, or other appeal bond or its equivalent.
- (d) "<u>Indemnitee</u>" means any current or former director or officer of the Corporation in such person's capacity as such, and any other person the Corporation in its sole discretion designates as an "Indemnitee" as permitted by applicable law.

(e) "Proceeding" shall include any threatened, pending or completed action, suit, arbitration, mediation, alternate dispute resolution mechanism, investigation, inquiry, administrative hearing or any other actual, threatened or completed proceeding, whether brought by or in the right of the Corporation or otherwise and whether of a civil (including intentional or unintentional tort claims), criminal, administrative or investigative (formal or informal) nature, including appeal therefrom, in which Indemnitee was, is, will or might be involved as a party, potential party, non-party witness or otherwise by reason of the fact of Indemnitee's Corporate Status, by reason of any action (or failure to act) taken by him or her or of any action (or failure to act) on his or her part while acting pursuant to his or her Corporate Status, whether or not serving in such capacity at the time any liability or expense is incurred for which indemnification, reimbursement, or advancement of expenses can be provided under this Article VI. If the Indemnitee believes in good faith that a given situation may lead to or culminate in the institution of a Proceeding, this shall be considered a Proceeding under this Article VI. When the context so requires, "Proceeding" also refers to any claim, issue or matter within a Proceeding.

Section 6.09. Right of Indemnitee to Bring Suit. If a request for indemnification under Sections 6.01 or 6.02 of these Bylaws is not paid in full by the Corporation within ninety (90) days, or if a request for an advancement of expenses under Section 6.04 of these Bylaws is not paid in full by the Corporation within thirty (30) days, in each case, after a written request has been received by the Corporation, Indemnitee may at any time thereafter bring suit against the Corporation in the Court of Chancery of the State of Delaware or any other court of competent jurisdiction in the State of Delaware to recover the unpaid amount of the claim. In any such action, the Corporation shall have the burden of proving that Indemnitee was not entitled to the requested indemnification, advancement or payment of Expenses. It shall be a defense to any such action (other than an action brought to enforce a claim for Expenses incurred in defending any proceeding in advance of its final disposition where the required undertaking, if any is required, has been tendered to the Corporation) that Indemnitee has not met the standards of conduct which make it permissible under these Bylaws, the Certificate of Incorporation or the DGCL for the Corporation to indemnify Indemnitee for the amount claimed. Neither the failure of the Corporation (including its Board of Directors, independent legal counsel, or its stockholders) to have made a determination prior to the commencement of such action that indemnification or advancement is proper in the circumstances because Indemnitee has met the applicable standard of conduct set forth in these Bylaws, the Certificate of Incorporation or the DGCL, nor an actual determination by the Corporation (including its Board of Directors, independent legal counsel, or its stockholders) that Indemnitee has not met such applicable standard of conduct, shall be a defense to the action or create a presumption that Indemnitee has not met any applicable standard of conduct. If successful, in whole or in part, In

Section 6.10. <u>Survival of Indemnification and Advancement of Expenses</u>. The indemnification and advancement of expenses provided by, or granted pursuant to, this Article VI shall continue as to a person who has ceased to be a director or officer and shall inure to the benefit of the heirs, executors and administrators of such a person.

Section 6.11. <u>Change in Rights</u>. Neither any amendment nor repeal of this Article VI, nor the adoption of any provision in these Bylaws inconsistent with this Article VI, shall eliminate or reduce the effect of this Article VI in respect of any acts or omissions occurring prior to such alteration, amendment, addition to, repeal or adoption.

ARTICLE VII GENERAL PROVISIONS

Section 7.01. <u>Dividends</u>. Subject to any applicable provisions of law or the Certificate of Incorporation, dividends upon the shares of capital stock of the Corporation may be declared by the Board of Directors at any regular or special meeting of the Board of Directors and any such dividend may be paid in cash, property or shares of the Corporation's capital stock. A member of the Board of Directors, or a member of any committee designated by the Board of Directors, shall be fully protected in relying in good faith upon the records of the Corporation and upon such information, opinions, reports or statements presented to the Corporation by any of its officers or employees, or committees of the Board of Directors, or by any other person as to matters the director reasonably believes are within such other person's professional or expert competence and who the director or member reasonably believes or determines has been selected with reasonable care by or on behalf of the Corporation, as to the value and amount of the assets, liabilities and/or net profits of the Corporation, or any other facts pertinent to the existence and amount of surplus or other funds from which dividends might properly be declared and paid.

Section 7.02. Execution of Instruments. The Board of Directors may authorize, or provide for the authorization of, officers, employees or agents to enter into any contract or execute and deliver any instrument in the name and on behalf of the Corporation. Any such authorization must be in writing or by electronic transmission and may be general or limited to specific contracts or instruments. The officers of the Corporation may also execute and deliver such contracts or instruments which generally pertain to the duties associated with such officer's title. Any person who is authorized to execute a contract, instrument or other document on behalf of the Corporation may execute a power of attorney allowing another person to execute such document on behalf of the Corporation.

Section 7.03. <u>Voting as Stockholder</u>. Unless otherwise determined by resolution of the Board of Directors, the Chief Executive Officer, the President, if any, the Chief Financial Officer, Treasurer, Secretary or any other person authorized by the Board of Directors shall have full power and authority on behalf of the Corporation to attend any meeting of stockholders or equity holders of any corporation or other entity in which the Corporation may hold stock or equity interests, and to act, vote (or execute proxies to vote) and exercise in person or by proxy all other rights, powers and privileges incident to the ownership of such stock or equity interests. Such officers acting on behalf of the Corporation shall have full power and authority to execute any instrument expressing consent to or dissent from any action of any such corporation or entity without a meeting. The Board of Directors may by resolution from time to time confer such power and authority upon any other person or persons.

Section 7.04. Fiscal Year. The fiscal year of the Corporation shall be fixed, and shall be subject to change, by the Board of Directors.

Section 7.05. <u>Notices</u>. An affidavit of the Secretary or an Assistant Secretary or of the transfer agent or other agent of the Corporation that the notice has been given in writing or by a form of electronic transmission shall, in the absence of fraud, be *prima facie* evidence of the facts stated therein.

Section 7.06. Form of Records. Any records administered by or on behalf of the Corporation in the regular course of its business, including its stock ledger, books of account and minute books, may be kept on or by means of, or be in the form of, any information storage device or method, or one or more electronic networks or databases (including one or more distributed electronic networks or databases), provided that the records so kept can be converted into clearly legible paper form within a reasonable time, and, with respect to the stock ledger, that the records so kept (i) can be used to prepare the list of stockholders specified in the DGCL, (ii) record the information specified in the DGCL, and record transfers as specified in the DGCL. The Corporation shall so convert any records so kept upon the request of any person entitled to inspect such records pursuant to any provision of the DGCL.

Section 7.07. Severability. If any provision (or any part thereof) of these Bylaws shall be held to be invalid, illegal or unenforceable as applied to any circumstance for any reason whatsoever the validity, legality and enforceability of such provisions in any other circumstance and of the remaining provisions of these Bylaws (including, without limitation, each portion of any section of these Bylaws containing any such provision held to be invalid, illegal or unenforceable that is not itself held to be invalid, illegal or unenforceable) shall not in any way be affected or impaired thereby.

ARTICLE VIII AMENDMENT OF BYLAWS

Section 8.01. <u>By the Board</u>. Subject to the provisions of the Certificate of Incorporation, and to limitations and requirements set forth in the GP LLCA, the Board of Directors may make, alter, amend, add to or repeal any and all of these Bylaws.

Section 8.02. <u>By the Stockholders</u>. Subject to the provisions of the Certificate of Incorporation, the affirmative vote of the holders of at least a majority of the voting power of the Corporation's then outstanding shares entitled to vote generally in the election of directors, voting together as a single class, shall be required for the stockholders to make, alter, amend, add to or repeal any or all Bylaws of the Corporation or to adopt any provision inconsistent therewith.

ARTICLE IX CONSTRUCTION

Notwithstanding anything to the contrary in these Bylaws, in the event of any conflict between the provisions of these Bylaws as in effect from time to time and the provisions of the Certificate of Incorporation as in effect from time to time, the provisions of the Certificate of Incorporation (including the provisions of Section 5.8 thereof prior to the Sunset) shall be controlling. References in the Certificate of Incorporation and these Bylaws to the Board of Directors shall apply to the Board of Directors, the Executive Committee and the Other Committees, as the context requires consistent with the authority granted pursuant to Section 5.8 of the Certificate of Incorporation. The provisions of these Bylaws shall be subject to the Corporation's obligations under the GP LLCA and the Corporation shall not take any action

inconsistent with such obligations. Unless the context requires otherwise, the general provisions, rules of construction and definitions in the DGCL shall govern the construction of these Bylaws. Without limiting the generality of this provision, the singular number includes the plural, the plural number includes the singular, and the term "person" includes corporations, other business entities, and natural persons. Unless otherwise provided in these Bylaws, a reference to any specific agreement or other document shall be deemed a reference to such agreement or document as amended from time to time in accordance with the terms of such agreement or document. When the terms of these Bylaws refer to a specific agreement or other document or a decision by any body, person or entity to determine the meaning or operation of a provision hereof, the Secretary of the Corporation shall maintain a copy of such agreement, document or decision at the principal executive offices of the Corporation, which shall be publicly available with the Corporation's public filings or, to the extent not publicly available, a copy thereof shall be provided free of charge to any stockholder who makes a request therefor.

Adopted January 12, 2022

SHEARMAN & STERLING LLP

599 Lexington Avenue New York, NY 10022-6069 +1.212.848.4000

January 13, 2022

TPG Inc. 301 Commerce Street, Suite 3300 Fort Worth, TX 76102

Ladies and Gentlemen:

We are acting as counsel for TPG Inc., a Delaware corporation (the "<u>Company</u>"), in connection with preparation and filing by the Company of a registration statement on Form S-8 (the "<u>Registration Statement</u>") with the Securities and Exchange Commission under the Securities Act of 1933, as amended (the "<u>Securities Act</u>"), with respect to 30,694,780 shares of Class A common stock, par value \$0.001, of the Company (the "<u>Shares</u>") that may be delivered from time to time pursuant to the TPG Inc. Omnibus Equity Incentive Plan (the "<u>Plan</u>"). In connection with the foregoing, we have reviewed originals or copies identified to our satisfaction of the following documents:

- (a) The Registration Statement;
- (b) The certificate of incorporation and bylaws of the Company, in each case as amended to date; and
- (c) Originals or copies of such other corporate records of the Company, certificates of public officials and of officers of the Company, and agreements and other documents as we have deemed necessary as a basis for the opinions expressed below.

In our examination, we have assumed the genuineness of all signatures, the authenticity of all documents, certificates and instruments submitted to us as originals and the conformity with originals of all documents submitted to us as copies.

Our opinion set forth below is based on the text of the Plan as referenced in the Exhibit Index to the Registration Statement.

Our opinion expressed below is limited to the General Corporation Law of the State of Delaware, and we do not express any opinion herein concerning any other law.

Based upon and subject to the foregoing and having regard for such legal considerations as we have deemed relevant, we are of the opinion that authorized but not previously issued Shares that may be delivered under the Plan have been duly authorized by the Company and, when (a) issued and delivered by the Company in accordance with the terms of the Plan and (b) paid for in full in accordance with the terms of the Plan, will be validly issued, fully paid and non-assessable.

This opinion letter speaks only as of the date hereof. We expressly disclaim any responsibility to advise you of any development or circumstance of any kind, including any change of law or fact that may occur after the date of this opinion letter that might affect the opinions expressed herein.

We hereby consent to the filing of this opinion as an exhibit to the Registration Statement. In giving this consent, we do not thereby concede that we come within the category of persons whose consent is required by the Securities Act or the General Rules and Regulations of the Securities and Exchange Commission promulgated thereunder.

/s/ Shearman & Sterling LLP Shearman & Sterling LLP

Very truly yours,

CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We consent to the incorporation by reference in this Registration Statement on Form S-8 of our report dated September 1, 2021, relating to the financial statements of TPG Partners, LLC appearing in the Registration Statement No. 333-261681 on Form S-1 of TPG Partners, LLC. We also consent to the reference to us under the heading "Experts" in such Registration Statement.

/s/ Deloitte & Touche LLP

Fort Worth, Texas January 13, 2022

CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We consent to the incorporation by reference in this Registration Statement on Form S-8 of our report dated September 1, 2021, relating to the financial statements of TPG Group Holdings (SBS), L.P. appearing in the Registration Statement No. 333-261681 on Form S-1 of TPG Partners, LLC. We also consent to the reference to us under the heading "Experts" in such Registration Statement.

/s/ Deloitte & Touche LLP

Fort Worth, Texas January 13, 2022

TPG Inc. Omnibus Equity Incentive Plan

ARTICLE 1. ESTABLISHMENT, PURPOSE AND DURATION

- **1.1 Establishment.** TPG Inc., a Delaware corporation, establishes an incentive compensation plan to be known as the TPG Inc. Omnibus Equity Incentive Plan, as set forth in this document. This Plan permits the grant of Nonqualified Stock Options, Incentive Stock Options, Stock Appreciation Rights, Restricted Stock, Restricted Stock Units, Performance Shares, Performance Units, Cash-Based Awards and Other Stock-Based Awards. This Plan shall become effective upon the effectiveness of the initial public offering of the Company (the "**Effective Date**") and shall remain in effect as provided in Section 1.3. This Plan was approved by the Board on December 7, 2021 and the Company's shareholders on December 20, 2021.
- **1.2 Purpose of this Plan**. The purpose of this Plan is to attract, retain and motivate TPG Personnel, to compensate them for their contributions to the long-term growth and profits of the Company and to align the interests of TPG Personnel with those of Company shareholders by encouraging TPG Personnel to acquire a proprietary interest in the success of the Company.
- **1.3 Duration of this Plan**. Unless sooner terminated as provided herein, this Plan shall terminate on the day before the tenth anniversary of the Effective Date. After this Plan is terminated, no Awards may be granted under this Plan, but Awards previously granted will remain outstanding in accordance with their applicable terms and conditions and this Plan's terms and conditions.

ARTICLE 2. DEFINITIONS

Whenever used in this Plan, the following terms shall have the meanings set forth below, and when the meaning is intended, the initial letter of the word shall be capitalized.

- **2.1** "**10% Owner**" has the meaning set forth in Section 6.6(a)(iii).
- **2.2** "AAA" has the meaning set forth in Section 23.26(a).
- **2.3** "Affiliate" means any Subsidiary and any Person or entity that directly or indirectly, through one or more intermediaries, controls, is controlled by or is under common control with the Company.
- **2.4** "Award" means, individually or collectively, a grant under this Plan of Nonqualified Stock Options, Incentive Stock Options, Stock Appreciation Rights, Restricted Stock, Restricted Stock Units, Performance Shares, Performance Units, Cash-Based Awards or Other Stock-Based Awards, in each case subject to the terms of this Plan.
- 2.5 "Award Agreement" means either (a) a written or electronic agreement entered into by the Company and a Participant setting forth the terms and provisions applicable to an Award granted under this Plan, including any amendment or modification thereof or (b) a written or electronic statement issued by the Company to a Participant describing the terms and provisions of such Award, including any amendment or modification thereof. The Committee may provide for the use of electronic, internet or other non-paper Award Agreements, and the use of electronic, internet or other non-paper means for the acceptance thereof and actions thereunder by a Participant. The Committee shall have the exclusive authority to determine the terms of an Award Agreement evidencing an Award granted under this Plan, subject to the provisions herein. The terms of an Award Agreement need not be uniform among all Participants or among similar types of Awards.

- **2.6** "Beneficial Owner" or "Beneficial Ownership" shall have the meaning ascribed to such term in Rule 13d-3 of the General Rules and Regulations under the Exchange Act.
 - **2.7** "Board" or "Board of Directors" means the Board of Directors of the Company.
 - 2.8 "Cash-Based Award" means an Award, denominated in cash, granted to a Participant as described in Article 12.
- 2.9 "Cause" means, unless otherwise provided in the applicable Award Agreement or in a separation policy of the Company or its Affiliates applicable to the Participant, (a) indictment for a felony or other crime involving moral turpitude; or (b) in the reasonable determination of the Committee: (i) a material breach of any employment agreement, offer letter, or similar agreement; (ii) a material breach of a policy of the Company Group governing Participant's service; (iii) Participant's material and sustained dereliction of duties assigned to Participant in the course of Participant's services or other egregious conduct of Participant that would customarily result in termination of an employee or partner (other than as a result of death or Disability), in each case as reasonably determined by the Committee; (iv) any fraud, embezzlement, theft, bad faith or similar misconduct (whether or not in connection with Participant's services); (v) gross negligence in connection with Participant's services; (vi) any order or decree from a court or regulatory agency involving Participant's violation of federal or state securities laws, regulations or rules; or (vii) other material misconduct that is or could reasonably be expected to be harmful to the business interests or reputation of the Company Group. If following Participant's termination other than for Cause, it is discovered that Participant's services could have been terminated for Cause, Participant's services will, at the election of the Committee and without notice to Participant, be deemed to have been terminated for Cause retroactively to the date the events giving rise to Cause occurred.
- **2.10** "Change in Control" means, except as otherwise provided in the applicable Award Agreement, the occurrence of any of the following events:
- (a) any Person or any group of Persons acting together that would constitute a "group" for purposes of Section 13(d) of the Exchange Act (excluding (i) a corporation or other entity owned, directly or indirectly, by the shareholders of the Company in substantially the same proportions as their ownership of stock of the Company, (ii) Permitted Investors or (iii) a Person or group of Persons in which one or more Affiliates of Permitted Investors, directly or indirectly, hold Beneficial Ownership of securities representing more than 50% of the total voting power in such Person or held by such group (including TPG GP A, LLC, collectively, "**Permitted Acquirers**")) is or becomes the Beneficial Owner, directly or indirectly, of securities of the Company representing more than 50% of the combined voting power of the Company's then outstanding voting securities; or
- (b) the following individuals cease for any reason to constitute a majority of the number of Directors then serving: (i) on or before the Sunset, individuals who, on the Effective Date, constitute the Board and any new Director elected by the Company's shareholders and (ii) following the Sunset, individuals who, as of the Sunset, constitute the Board and any new director whose appointment or election by the Board or nomination for election by the Company's shareholders was approved or recommended by a vote of at least a majority of the Directors then still in office who either were Directors on the Effective Date or whose appointment, election or nomination for election was previously so approved or recommended by the Directors referred to in this subclause (ii); or
- (c) there is consummated a merger or consolidation of the Company with any other corporation or other entity and, immediately after the consummation of such merger or consolidation, either (i) the Board immediately prior to the merger or consolidation does not constitute at least a majority of the board of directors of the company surviving the merger or, if the surviving company is a Subsidiary, the ultimate parent thereof, or (ii) the voting securities of the Company (including Common Units exchangeable

to any voting securities of the Company pursuant to the Exchange Agreement on an as-exchanged basis) immediately prior to such merger or consolidation do not continue to represent or are not converted into more than 50% of the combined voting power of the then outstanding voting securities of the Person resulting from such merger or consolidation or, if the surviving company is a Subsidiary, the ultimate parent thereof; or

- (d) except as may otherwise be determined by the Committee, any Person or any group of Persons acting together that would constitute a "group" for purposes of Section 13(d) of the Exchange Act, other than a Permitted Acquirer (i) becomes the Beneficial Owner of the Company's securities, directly or indirectly, having more than 30% of the total voting power of the then outstanding securities of the Company that may be cast for the election of Directors of the Company and (ii) Beneficially Owns more of such total voting power than is Beneficially Owned by any Permitted Acquirer; or
- (e) the shareholders of the Company approve a plan of complete liquidation or dissolution of the Company or there is consummated an agreement or series of related agreements for the sale, lease or other disposition, directly or indirectly, by the Company of all or substantially all of the Company's assets, other than such sale or other disposition by the Company of all or substantially all of the Company's assets to an entity at least 50% of the combined voting power of the voting securities of which are owned by shareholders of the Company in substantially the same proportions as their ownership of the Company immediately prior to such sale.

Notwithstanding the foregoing, (A) the Sunset in and of itself shall not constitute a "Change in Control" and (B) except with respect to clause (b) and clause (c)(i) above, a "Change in Control" shall not be deemed to have occurred by virtue of the consummation of any transaction or series of integrated transactions immediately following which the holders of the shares of the Company (including Common Units exchangeable to any shares of the Company pursuant to the Exchange Agreement on an as-exchanged basis) immediately prior to such transaction or series of transactions continue to have substantially the same proportionate ownership in, and voting control over, and own substantially all of the shares of, an entity which owns, directly or indirectly, all or substantially all of the assets of the Company immediately following such transaction or series of transactions. For purposes of this definition of Change in Control, references to "the Company's stock/securities" or "stock/securities of the Company" shall include, for the avoidance of doubt, Common Units or other securities or rights of any entity that are convertible into, or exercisable or exchangeable for, the Company's stock or other securities. To the extent that any payment or benefit granted under this Plan constitutes "non-qualified deferred compensation" subject to Code Section 409A, and to the extent that such payment or benefit is payable upon a Participant's termination of employment, then such payments or benefits shall be payable only upon a "change in control" as defined in Code Section 409A.

- **2.11** "Code" means the U.S. Internal Revenue Code of 1986, as amended from time to time and the applicable regulations and guidance promulgated thereunder and any successor or similar provision.
- 2.12 "Committee" means the Board or any committees or individuals (any of whom may be one or more Employees or Partners) designated by the Board or governance guidelines of the Company to have authority under this Plan. Authority under this Plan may be designated to one or more committees or individuals simultaneously, and each such committee or individual may be designated to have broad or limited authority with respect to this Plan. References to the "Committee" shall be deemed references to any such designated committee or individual, but only to the extent of the authority designated to such committee or individual. If the Committee does not exist or cannot function for any reason, the Board may take any action under this Plan that would otherwise be the responsibility of the Committee, in which case references to the "Committee" shall be deemed references to the Board. The Committee (or a subcommittee thereof) shall be constituted to comply with the requirements of Rule 16(b) of the Exchange Act and any applicable listing or governance requirements of any securities exchange on which the Shares are listed; provided, however, that, if any Committee member is found not to have met the qualification requirements of Section 16(b) of the Exchange Act, any actions taken or Awards granted by the Committee shall not be invalidated by such failure to so qualify.

- **2.13** "Common Stock" means the Class A common stock of the Company, par value \$0.001 per share, or such other class of share or other securities as may be applicable under Section 4.3 of this Plan.
 - **2.14 "Common Unit"** has the meaning set forth in the Exchange Agreement.
 - **2.15** "Company" means TPG Inc., a Delaware corporation, and any successor thereto as provided in Section 23.24.
- **2.16** "Company Group" means, collectively, the Company and its direct and indirect Subsidiaries (to the extent of its economic ownership interest in such subsidiaries) and its Affiliates.
 - **2.17** "Control" has the meaning set forth in Rule 405 under the Securities Act.
 - **2.18** "Corporate Transactions" has the meaning set forth in Section 4.3(a).
 - **2.19** "Data" has the meaning set forth in Section 23.4.
 - **2.20** "Director" means any individual who is a member of the Board of Directors of the Company.
- **2.21** "Disability" means a determination by the Committee in good faith or by a physician designated by or otherwise approved by the Company of any physical or mental incapacity that prevents Participant from carrying out all or substantially all of Participant's duties for any period of 180 consecutive days or any aggregate period of six months in any 12-month period. For the avoidance of doubt, Disability shall not include Retirement or other termination of employment by Participant.
 - 2.22 "Dividend Equivalents" has the meaning set forth in Article 18.
 - **2.23** "Effective Date" has the meaning set forth in Section 1.1.
- 2.24 "Employee" means any individual (or any wholly-owned corporate alter ego of the applicable individual) performing Services for the Company or an Affiliate and designated as an employee of the Company or an Affiliate on its payroll records. An Employee shall not include any individual during any period he or she is classified or treated by the Company or Affiliate as an independent contractor, a consultant or an employee of an employment, consulting or temporary agency or any other entity other than the Company or Affiliate, without regard to whether such individual is subsequently determined to have been, or is subsequently retroactively reclassified, as a common-law employee of the Company or Affiliate during such period. An individual shall not cease to be an Employee in the case of (a) any leave of absence approved by the Company or (b) transfers between locations of the Company or between the Company or any Affiliate (unless determined otherwise by the Committee). For purposes of Incentive Stock Options, no such leave may exceed 90 days, unless reemployment upon expiration of such leave is guaranteed by statute or contract. If reemployment upon expiration of a leave of absence approved by the Company is not so guaranteed, then three months following the 91st day of such leave, any Incentive Stock Option held by a Participant shall cease to be treated as an Incentive Stock Option and shall be treated for tax purposes as a Nonqualified Stock Option. Neither Service as a Director nor payment of a director's fee by the Company shall alone be sufficient to constitute "employment" by the Company.

- **2.25** "Exchange Act" means the Securities Exchange Act of 1934, as amended from time to time, or any successor act thereto and the regulations and guidance promulgated thereunder.
- **2.26 "Exchange Agreement"** means the Exchange Agreement, dated on or about the date hereof, by and among the Company, TPG Operating Group I, L.P., TPG Operating Group II, L.P., and each "TPG OG Limited Partner" (as defined therein), as amended from time to time.
- **2.27** "Fair Market Value" or "FMV" means, with respect to a Share, the fair market value thereof as of the relevant date of determination, as determined in accordance with the valuation methodology approved by the Committee (based on objective criteria) from time to time. In the absence of any alternative valuation methodology approved by the Committee, Fair Market Value shall be deemed to be equal to the volume weighted average trading price of a Share on NASDAQ as reported on by NASDAQ (or such established national securities exchange as may be designated by the Committee or, in the event that the Common Stock is not listed for trading on NASDAQ or such other national securities exchange as may be designated by the Committee but is quoted on an automated system), for the ten-day period immediately preceding the valuation date (or, if there were no sales on the valuation date, the average of the highest and lowest quoted selling prices as reported on said composite tape or automated system for the most recent day during which a sale occurred). The definition of FMV may differ depending on whether FMV is in reference to the grant, exercise, vesting, settlement or payout of an Award.
 - **2.28 "Grant Date"** means the date an Award is granted to a Participant pursuant to this Plan.
 - **2.29** "Grant Price" means the price established at the time of grant of an SAR pursuant to Article 7.
- **2.30** "Incentive Stock Option" or "ISO" means an Award granted pursuant to Article 6 that is designated as an Incentive Stock Option and that is intended to meet the requirements of Code Section 422 or any successor provision.
- **2.31 "Insider"** shall mean an individual who is, on the relevant date, an officer (as defined in Rule 16a-1(f) of the Exchange Act (or any successor provision)) or Director of the Company, or a more than 10% Beneficial Owner of any class of the Company's equity securities that is registered pursuant to Section 12 of the Exchange Act, as determined by the Board in accordance with Section 16 of the Exchange Act.
 - **2.32** "ISO Subsidiary" has the meaning set forth in Section 6.6(a)(ii).
 - 2.33 "NASDAQ" means the NASDAQ Global Select Market.
 - **2.34** "Non-Employee Director" means a Director who is not an Employee or a Partner.
- **2.35** "Nonqualified Stock Option" or "NQSO" means an Award granted pursuant to Article 6 that is not intended to meet the requirements of Code Section 422, or that otherwise does not meet such requirements.
- **2.36 "Option"** means an Award granted to a Participant pursuant to Article 6, which Award may be an Incentive Stock Option or a Nonqualified Stock Option.
 - **2.37 "Option Price"** means the price at which a Share may be purchased by a Participant pursuant to an Option.

- **2.38 "Other Stock-Based Award"** means an equity-based or equity-related Award not otherwise described by the terms of this Plan that is granted pursuant to Article 12.
 - **2.39 "Parent Corporation"** has the meaning set forth in Section 6.6(a)(i).
 - **2.40** "Participant" means any eligible individual as set forth in Article 5 to whom an Award is granted.
- **2.41 "Partner**" means any individual (or any wholly-owned corporate alter ego of the applicable individual) performing Services for the Company or an Affiliate and designated as a partner of an Affiliate of the Company. An individual shall not cease to be a Partner in the case of (a) any leave of absence approved by the Company or (b) transfers between locations of the Company or between the Company or any Affiliate (unless determined otherwise by the Committee).
 - 2.42 "Performance Measures" means measures, as described in Article 14, upon which performance goals are based.
- **2.43** "**Performance Period**" means the period during which performance goals must be met in order to determine the degree of payout or vesting with respect to an Award.
 - 2.44 "Performance Share" means an Award granted pursuant to Article 10.
 - **2.45 "Performance Unit"** means an Award granted pursuant to Article 11.
- **2.46** "**Period of Restriction**" means the period when an Award is subject to a substantial risk of forfeiture (based on the passage of time, the achievement of performance goals or upon the occurrence of other events as determined by the Committee, in its discretion).
- **2.47 "Permitted Investors"** means (a) each TPG Partner, (b) each other Person that directly or indirectly owns Common Units on the Effective Date and (c) with respect to any Person referred to in (a) or (b), any member of such Person's immediate family or any estate planning entity of such Person (other than a charitable organization or foundation or any Person wholly owned and controlled, directly or indirectly, by such charitable organization or foundation).
 - **2.48 "Permitted Transferees"** has the meaning set forth in Section 13.2.
 - **2.49** "Person" shall have the meaning ascribed to such term in Section 3(a)(9) of the Exchange Act and used in Sections 13(d) and 14(d) thereof.
 - 2.50 "Plan" means this TPG Inc. Omnibus Incentive Plan, as may be amended from time to time.
 - 2.51 "Restricted Stock" means an Award granted pursuant to Article 8.
 - **2.52** "Restricted Stock Unit" means an Award granted pursuant to Article 9.
- **2.53 "Retirement"** shall have the meaning set forth in the applicable Award Agreement, or such other definition as the Committee may determine from time to time, and, for the avoidance of doubt, may have an equivalent meaning as set forth in an applicable Company policy.
 - **2.54** "Service" means the performance of services by a TPG Person.
 - 2.55 "Share" means a share of Common Stock.

- **2.56** "Share Payment" has the meaning set forth in Section 22.2.
- **2.57** "**Share Reserve**" has the meaning set forth in Section 4.1.
- **2.58** "Stock Appreciation Right" or "SAR" means an Award granted pursuant to Article 7.
- **2.59** "Subsidiary" means (a) TPG Operating Group, (b) any other corporation or other entity (domestic or foreign) with respect to which the Company, directly or indirectly, has the power, whether through the ownership of voting securities, by contract or otherwise, to elect at least a majority of the members of such corporation's board of directors or analogous governing body, (c) any other corporation or entity in which the Company, directly or indirectly, has an equity or similar interest and which the Committee designates as a Subsidiary for purposes of this Plan or (d) any partnership (i) the sole general partner or the managing general partner of which is the Company or a Subsidiary or (ii) the only general partners of which are the Company or one or more Subsidiaries (or any combination thereof). Notwithstanding the foregoing, portfolio companies and pooled investment entities of the Company shall not constitute Subsidiaries.
 - 2.60 "Successor" has the meaning set forth in Section 23.24.
 - 2.61 "Sunset" has the meaning set forth in the Company's Certificate of Incorporation.
 - **2.62** "Tax Laws" has the meaning set forth in Section 23.22.
- **2.63** "**Termination Date**" means (a) with respect to a Termination of Services for any reason other than a Retirement or death, the date on which either TPG or the Participant provides notice of Termination of Services to the other party, and (b) with respect to a Termination of Services due to a Retirement or death, the last date on which the Participant is providing Service.
- **2.64** "**Termination of Directorship**" means the time when a Non-Employee Director ceases to be a Non-Employee Director for any reason, including, but not by way of limitation, a termination by resignation, failure to be elected or death.
- **2.65** "**Termination of Services**" means termination or cessation of the Participant's Services, regardless of the reason for the termination of Services and who initiates the termination of Services (including, for the avoidance of doubt, Termination of Directorship or termination of a consultant or third-party service provider). With respect to termination of employment, Termination of Services occurs on the date immediately following only the minimum notice of termination period, if any, prescribed by applicable employment or labor standards legislation and does not include any additional notice period to which a Participant might be entitled under contract or the common law.
 - **2.66** "TPG" means, collectively, the Company and its Affiliates.
- **2.67** "**TPG Operating Group**" means, collectively, TPG Operating Group I, L.P., a Delaware limited partnership, TPG Operating Group II, L.P., a Delaware limited partnership, and each of their respective consolidated subsidiaries, provided that, TPG Operating Group shall not include Tarrant RemainCo I, L.P., a Delaware limited partnership, Tarrant RemainCo II, L.P., a Delaware limited partnership, Tarrant RemainCo III, L.P., a Delaware limited partnership, and L.P., a Delaware limited partnership, and L.P., a Delaware limited partnership, and L.P., a Delaware limited partnership,
- **2.68 "TPG Partner"** means, as of any date, any Active Partner or Former Partner (each as defined in the Seventh Amended and Restated Limited Partnership Agreement of TPG Partner Holdings, L.P., dated on or about the date hereof).

2.69 "**TPG Personnel**" means Employees (including prospective employees), Partners (including prospective partners), Directors and any consultant, agent, advisor or independent contractor (or, in each case, any wholly-owned corporate alter ego of the applicable individual) who renders bona fide Services to the Company or an Affiliate that (a) are not in connection with the offer and sale of the Company's securities in a capital-raising transaction, (b) do not directly or indirectly promote or maintain a market for the Company's securities, and (c) are provided by a natural person who has contracted directly with the Company or an Affiliate to render such Services. "**TPG Person**" shall have a correlative, singular meaning.

ARTICLE 3. ADMINISTRATION

- **3.1 General**. The Committee shall be responsible for administering this Plan, subject to this Article 3 and the other provisions of this Plan. In consultation with management of the Company, the Committee may employ attorneys, consultants, accountants, agents and other individuals as may reasonably be necessary to assist it in the administration of this Plan, any of whom may be an Employee or Partner, and the Committee, the Company and its officers and Directors shall be entitled to rely upon the advice, opinions or valuations of any such individuals. No member of the Committee shall be liable for any action taken or not taken in reliance upon any such information or advice. All actions taken and all interpretations and determinations made by the Committee under this Plan shall be made in its sole discretion and shall be final, binding and conclusive upon the Participants, the Company or any Affiliate, and all other interested individuals.
- **3.2 Authority of the Committee.** Subject to any express limitations set forth in this Plan and the limitations of any delegation of authority to the Committee, the Committee shall have full and exclusive discretionary power and authority to take such actions as it deems necessary and advisable with respect to the administration, interpretation and implementation of this Plan including, but not limited to, the following:
- (a) To determine from time to time which of the persons eligible under this Plan will be granted Awards, when and how each Award will be granted, what type or combination of types of Awards will be granted, the provisions of each Award granted (which need not be identical), including the time or times when an eligible individual will be permitted to receive Shares pursuant to an Award and the number of Shares subject to an Award;
- (b) To construe and interpret this Plan and Awards granted under it, and to establish, amend, and revoke rules and regulations for its administration. The Committee, in the exercise of this power, may correct any defect, omission or inconsistency in this Plan or in an Award Agreement, in a manner and to the extent it shall deem necessary or expedient to make this Plan fully effective;
 - (c) To approve forms of Award Agreements for use under this Plan;
 - (d) To determine Fair Market Value of a Share;
- (e) To employ attorneys, consultants, accountants, agents and other individuals as may reasonably be necessary to assist it in the administration of this Plan, any of whom may be an Employee or Partner;
 - (f) To amend this Plan, an Award or any Award Agreement after the date of grant subject to the terms of this Plan;
 - (g) To apply alternative definitions than are reflected in this Plan;

- (h) To adopt sub-plans and special provisions applicable to Awards regulated by the laws of a jurisdiction other than and outside of the United States;
- (i) To authorize any Person to execute on behalf of the Company any instrument required to effect the grant of an Award previously granted by the Board;
 - (i) To determine whether Awards shall be settled in Shares, cash or in any combination thereof;
 - (k) To determine whether Awards shall provide for Dividend Equivalents;
- (l) To establish a program whereby Participants designated by the Committee may elect to receive Awards under this Plan in lieu of compensation otherwise payable in cash; and
- (m) To impose such restrictions, conditions or limitations as it determines appropriate as to the timing and manner of any resales by a Participant or other subsequent transfers by a Participant of any Shares, including, without limitation, restrictions under an insider trading policy and restrictions as to the use of a specified brokerage firm for such resales or other transfers.
- **3.3 Delegation**. To the extent not prohibited by applicable laws, rules and regulations, the Committee may delegate to (a) one or more of its members, (b) one or more officers of the Company or any Affiliate, or (c) one or more agents or advisors such administrative duties or powers as it may deem appropriate or advisable under such conditions and limitations as the Committee may set at the time of such delegation or thereafter, including but not limited to the authority to grant Awards to a TPG Person and the authority to manage the day-to-day administrative operations of the Plan. The Committee or any individuals to whom it has delegated duties or powers as aforesaid may employ one or more individuals to render advice with respect to any responsibility the Committee or such individuals may have under this Plan. For purposes of this Plan, reference to the Committee shall be deemed to refer to any subcommittee, subcommittees or other persons or groups of persons to whom the Committee delegates authority pursuant to this Section 3.3.

ARTICLE 4. SHARES SUBJECT TO THIS PLAN AND MAXIMUM AWARDS

4.1 Number of Shares Authorized and Available for Awards. Subject to adjustment as provided under this Plan, the maximum number of Shares that are available for Awards under this Plan shall be equal to 30,694,780 (the "**Share Reserve**"), subject to adjustment as provided in Section 4.3 hereof. Notwithstanding the foregoing, the Share Reserve shall automatically increase on January 1st of each year beginning in 2023 and ending with a final increase on January 1, 2032, by a number of Shares equal to the amount (if any) by which (a) 10% of the aggregate number of outstanding Shares on a fully converted and diluted basis (assuming the redemption of all then outstanding common units and promote units of the TPG Operating Group) on the last day of the immediately preceding fiscal year exceeds (b) the aggregate number of Shares available for issuance under this Plan, unless the Committee should decide to increase the number of Shares available under this Plan by a lesser amount. Shares authorized under this Plan may be authorized and unissued Shares, Shares that have been reacquired by the Company, treasury Shares or any combination of the foregoing, as may be determined from time to time by the Board or by the Committee. Any of the authorized Shares may be used for any type of Award under this Plan, and any or all of the Shares may be allocated to Incentive Stock Options; provided that the maximum number of Shares that may be issued as Incentive Stock Options under this Plan may not exceed 10,000,000.

- **4.2 Share Usage**. The number of Shares remaining available for issuance shall be reduced by the number of Shares subject to outstanding Awards and, for Awards that are not denominated by Shares, by the number of Shares actually delivered upon settlement or payment of the Award. For purposes of determining the number of Shares that remain available for issuance under this Plan, the number of Shares related to an Award granted under this Plan that terminates by expiration, forfeiture, cancellation or otherwise without the issuance of the Shares or is settled through the issuance of consideration other than Shares (including cash), shall be available again for grant under this Plan. Shares tendered by a Participant, repurchased by the Company using proceeds from the exercise of Options or withheld by the Company in payment of the exercise price of an Option or to satisfy any tax withholding obligation for an Award shall not again be available for Awards.
- **4.3 Adjustments in Authorized Shares**. Adjustment in authorized Shares available for issuance under this Plan or under an outstanding Award shall be subject to the following provisions:
- (a) In the event of any corporate event or transaction such as a merger, consolidation, reorganization, recapitalization, separation, reclassification, partial or complete liquidation, stock dividend, stock split, reverse stock split, split up, spin-off, distribution of stock or property of the Company, combination of Shares, exchange of Shares, dividend in kind, extraordinary cash dividend, rights offering to purchase Shares at a price that is substantially below FMV or any other similar corporate event or transaction ("Corporate Transactions"), the Committee, in order to preserve, but not increase, Participants' rights under this Plan, shall substitute or adjust, as applicable, (i) the number and kind of Shares that may be issued under this Plan or under particular forms of Award Agreements, (ii) the number and kind of Shares subject to outstanding Awards (including by payment of cash to a Participant), and (iii) the Option Price or Grant Price applicable to outstanding Awards and other value determinations applicable to outstanding Awards. The Committee, in its discretion, shall determine the methodology or manner of making such substitution or adjustment subject to applicable laws, rules and regulations.
- (b) In addition to the adjustments permitted under Section 4.3(a) above, the Committee, in its sole discretion, may make such other adjustments or modifications in the terms of any Award that it deems appropriate to reflect any Corporate Transaction, including, but not limited to, modifications of performance goals and changes in the length of Performance Periods, subject to the limitations set forth in Section 14.2.
- (c) The determination of the Committee as to the foregoing adjustments, if any, shall be conclusive and binding on Participants under this Plan. Unless otherwise determined by the Committee, such adjusted Awards shall be subject to the same restrictions and vesting or settlement schedule to which the underlying Award is subject.

ARTICLE 5. ELIGIBILITY AND PARTICIPATION

- **5.1 Eligibility to Receive Awards**. Individuals eligible to participate in this Plan include all TPG Personnel (including, in each case, any whollyowned corporate alter ego of the applicable TPG Personnel), as determined by the Committee. No TPG Personnel has a right to participate in this Plan.
- **5.2 Participation in this Plan.** Subject to the provisions of this Plan, the Committee may, from time to time, select from all individuals eligible to participate in this Plan, those individuals to whom Awards shall be granted and shall determine, in its sole discretion, the nature of any and all terms permissible by law and the amount of each Award.

ARTICLE 6. STOCK OPTIONS

6.1 Grant of Options. Options may be granted to Participants in such number, and upon such terms, and at any time and from time to time as shall be determined by the Committee, in its sole discretion. Each grant of an Option shall be evidenced by an Award Agreement which shall specify whether the Option is in the form of a Nonqualified Stock Option or an Incentive Stock Option.

- **6.2 Option Price**. The Option Price for each grant of an Option shall be determined by the Committee in its sole discretion and shall be specified in the Award Agreement evidencing such Option; <u>provided</u>, <u>however</u>, the Option Price must be at least equal to 100% of the Fair Market Value of a Share as of the Option's Grant Date, subject to adjustment as provided for under Section 4.3.
- **6.3 Term of Option**. The term of an Option granted to a Participant shall be determined by the Committee, in its sole discretion; <u>provided</u>, <u>however</u>, no Option shall be exercisable later than the tenth-anniversary date of its grant. If upon the expiration of the term of an Option (other than an Incentive Stock Option), a Participant is prohibited from trading in the Shares by applicable laws, rules or regulations or the Company's insider trading plan as in effect from time to time, the term of the Option shall be automatically extended to the 30th day following the expiration of such prohibition; <u>provided</u>, <u>however</u>, that this provision shall not apply if prohibited by applicable laws, rules and regulations in effect from time to time.
- **6.4 Exercise of Option**. An Option shall be exercisable at such times and be subject to such restrictions and conditions as the Committee shall in each instance approve, which terms and restrictions need not be the same for each grant or for each Participant.
- **6.5 Payment of Option Price.** An Option shall be exercised by the delivery of a notice of exercise to the Company or an agent designated by the Company in a form specified or accepted by the Committee, or by complying with any alternative procedures that may be authorized by the Committee, setting forth the number of Shares with respect to which the Option is to be exercised, accompanied by full payment for the Shares. A condition of the issuance of the Shares as to which an Option shall be exercised shall be the payment of the Option Price. The Option Price of any exercised Option shall be payable to the Company in accordance with one of the following methods:
 - (a) in cash or its equivalent;
- (b) by tendering (either by actual delivery or attestation) previously acquired Shares having an aggregate Fair Market Value at the time of exercise equal to the Option Price;
 - (c) by a cashless (broker-assisted) exercise in accordance with procedures authorized by the Committee from time to time;
- (d) through net Share settlement or similar procedure involving the withholding of Shares subject to the Option with a value equal to the Option Price;
 - (e) by any combination of (a), (b), (c) and (d); or
 - (f) any other method approved or accepted by the Committee in its sole discretion.

Unless otherwise determined by the Committee, all payments under all of the methods indicated above shall be paid in United States dollars or Shares, as applicable.

6.6 Special Rules Regarding ISOs. The terms of any Incentive Stock Option granted under this Plan shall comply in all respects with the provisions of Code Section 422, or any successor provision thereto, as amended from time to time. Notwithstanding any provision of this Plan to the contrary, an Option granted in the form of an ISO to a Participant shall be subject to the following rules:

(a) Special ISO definitions:

- (i) "**Parent Corporation**" shall mean as of any applicable date a corporation in respect of the Company that is a parent corporation within the meaning of Code Section 424(e).
- (ii) "**ISO Subsidiary**" shall mean as of any applicable date any corporation in respect of the Company that is a subsidiary corporation within the meaning of Code Section 424(f).
- (iii) A "10% Owner" is an individual who owns stock possessing more than ten percent of the total combined voting power of all classes of stock of the Company or its Parent Corporation or any ISO Subsidiary.
 - (b) Eligible Employees. An ISO may be granted solely to eligible Employees of the Company, Parent Corporation or ISO Subsidiary.
 - (c) Specified as an ISO. An Award Agreement evidencing the grant of an ISO shall specify that such grant is intended to be an ISO.
- (d) **Option Price**. The Option Price for each grant of an ISO shall be determined by the Committee in its sole discretion and shall be specified in the Award Agreement; <u>provided</u>, <u>however</u>, the Option Price must be at least equal to 100% of the Fair Market Value of a Share as of the ISO's Grant Date (in the case of 10% Owners, the Option Price may not be less than 110% of such Fair Market Value), subject to adjustment provided for under Section 4.3.
 - (e) Right to Exercise. Any ISO granted to a Participant shall be exercisable during his or her lifetime solely by such Participant.
- (f) **Exercise Period**. The period during which a Participant may exercise an ISO shall not exceed ten years (five years in the case of a Participant who is a 10% Owner) from the date on which the ISO was granted.
- (g) **Termination of Services**. In the event a Participant terminates employment due to death or Disability (as defined in Code Section 22(e)(3)), the Participant (or, in the case of death, the person(s) to whom the Option is transferred by will or the laws of descent and distribution) shall have the right to exercise the Participant's ISO award during the period specified in the applicable Award Agreement solely to the extent the Participant had the right to exercise the ISO on the date of his death or Disability; as applicable; <u>provided</u>, <u>however</u>, that such period may not exceed one year from the date of such Termination of Services due to death or Disability or, if shorter, the remaining term of the ISO. In the event a Participant terminates employment for reasons other than death or Disability, the Participant shall have the right to exercise the Participant's ISO during the period specified in the applicable Award Agreement solely to the extent the Participant had the right to exercise the ISO on the date of such Termination of Services; <u>provided</u>, <u>however</u>, that such period may not exceed three months from the date of such Termination of Services or if shorter, the remaining term of the ISO.
- (h) **Dollar Limitation**. To the extent that the aggregate Fair Market Value of (i) the Shares with respect to which Options designated as Incentive Stock Options plus (ii) the shares of stock of the Company, Parent Corporation and any ISO Subsidiary with respect to which other Incentive Stock Options are exercisable for the first time by a holder of such Incentive Stock Options during any calendar year under all plans of the Company and ISO Subsidiary exceeds \$100,000, such Options shall be treated as Nonqualified Stock Options. For purposes of the preceding sentence, Options shall be taken into account in the order in which they were granted, and the Fair Market Value of the Shares shall be determined as of the time the Option or other incentive stock option is granted.

- (i) **Duration of Plan**. No ISO may be granted more than ten years after the earlier of (a) adoption of this Plan by the Board and (b) the Effective Date.
- (j) **Notification of Disqualifying Disposition**. If any Participant shall make any disposition of Shares issued pursuant to the exercise of an ISO, such Participant shall notify the Company of such disposition within 30 days thereof. The Company shall use such information to determine whether a disqualifying disposition as described in Code Section 421(b) has occurred.
- (k) **Transferability**. No ISO may be sold, transferred, pledged, assigned or otherwise alienated or hypothecated, other than by will or by the laws of descent and distribution; <u>provided</u>, <u>however</u>, that at the discretion of the Committee, an ISO may be transferred to a grantor trust under which Participant making the transfer is the sole beneficiary.

ARTICLE 7. STOCK APPRECIATION RIGHTS

- **7.1 Grant of SARs**. SARs may be granted to Participants in such number, and upon such terms, and at any time and from time to time as shall be determined by the Committee, in its sole discretion. Each grant of SARs shall be evidenced by an Award Agreement.
- **7.2 Grant Price**. The Grant Price for each grant of an SAR shall be determined by the Committee and shall be specified in the Award Agreement evidencing the SAR; <u>provided</u>, <u>however</u>, the Grant Price must be at least equal to 100% of the FMV of a Share as of the Grant Date, subject to adjustment as provided for under Section 4.3.
- **7.3 Term of SAR**. The term of an SAR granted to a Participant shall be determined by the Committee, in its sole discretion; <u>provided</u>, <u>however</u>, no SAR shall be exercisable later than the tenth-anniversary date of its grant.
- **7.4** Exercise of SAR. An SAR shall be exercisable at such times and be subject to such restrictions and conditions as the Committee shall in each instance approve, which terms and restrictions need not be the same for each grant or for each Participant.
- **7.5 Notice of Exercise.** An SAR shall be exercised by the delivery of a notice of exercise to the Company or an agent designated by the Company in a form specified or accepted by the Committee, or by complying with any alternative procedures that may be authorized by the Committee, setting forth the number of Shares with respect to which the SAR is to be exercised.
- **7.6 Settlement of SARs**. Upon the exercise of an SAR, pursuant to a notice of exercise properly completed and submitted to the Company in accordance with Section 7.5, a Participant shall be entitled to receive payment from the Company in an amount equal to the product of:
 - (a) The excess of the Fair Market Value of a Share on the date of exercise over the Grant Price; and
 - (b) The number of Shares with respect to which the SAR is exercised.

Payment shall be made in cash, Shares or a combination thereof as specified in the Award Agreement.

ARTICLE 8. RESTRICTED STOCK

8.1 Grant of Restricted Stock. Restricted Stock may be granted to Participants in such number, and upon such terms, and at any time and from time to time as shall be determined by the Committee, in its sole discretion. Each grant of Restricted Stock shall be evidenced by an Award Agreement.

- **8.2 Nature of Restrictions**. Each grant of Restricted Stock shall be subject to a Period of Restriction that shall lapse upon the satisfaction of such conditions and restrictions as are determined by the Committee in its sole discretion and set forth in an applicable Award Agreement. Such conditions or restrictions may include, without limitation, one or more of the following:
 - (a) A requirement that a Participant pay a stipulated purchase price for each Share of Restricted Stock;
 - (b) Restrictions based upon the achievement of specific performance goals;
 - (c) Time-based restrictions on vesting following the attainment of the performance goals;
 - (d) Time-based restrictions; and
- (e) Restrictions under applicable laws and restrictions under the requirements of any stock exchange or market on which such Shares are listed or traded.
- **8.3 Issuance of Shares**. To the extent deemed appropriate by the Committee, the Company may retain the certificates representing Shares of Restricted Stock in the Company's possession until such time as all conditions or restrictions applicable to such Shares have been satisfied or lapse. Shares of Restricted Stock covered by each Restricted Stock grant shall become freely transferable by the Participant after all conditions and restrictions applicable to such Shares have been satisfied or lapsed (including satisfaction of any applicable tax withholding obligations).
- **8.4 Shareholder Rights**. Unless otherwise determined by the Committee and set forth in a Participant's applicable Award Agreement, to the extent permitted or required by law a Participant holding Shares of Restricted Stock granted hereunder shall be granted full rights as a shareholder of the Company (including voting rights) with respect to those Shares during the Period of Restriction.

ARTICLE 9. RESTRICTED STOCK UNITS

- **9.1 Grant of Restricted Stock Units**. Restricted Stock Units may be granted to Participants in such number, and upon such terms, and at any time and from time to time as shall be determined by the Committee, in its sole discretion. A grant of Restricted Stock Units shall not represent the grant of Shares but shall represent a promise to deliver a corresponding number of Shares or the value of each Share based upon the completion of Service, performance conditions or such other terms and conditions as specified in the applicable Award Agreement. Each grant of Restricted Stock Units shall be evidenced by an Award Agreement.
- **9.2 Nature of Restrictions**. Each grant of Restricted Stock Units that is subject to a Period of Restriction, such Period of Restriction shall lapse upon the satisfaction of such conditions and restrictions as are determined by the Committee in its sole discretion and set forth in an applicable Award Agreement. Such conditions or restrictions may include, without limitation, one or more of the following:
 - (a) A requirement that a Participant pay a stipulated purchase price for each Restricted Stock Unit;
 - (b) Restrictions based upon the achievement of specific performance goals;

- (c) Time-based restrictions on vesting following the attainment of the performance goals;
- (d) Time-based restrictions; and
- (e) Restrictions under applicable laws or under the requirements of any stock exchange on which Shares are listed or traded.
- **9.3 Settlement and Payment Restricted Stock Units**. Unless otherwise elected by the Participant or otherwise provided for in the Award Agreement, Restricted Stock Units shall be settled upon the date such Restricted Stock Units vest. Such settlement may be made in Shares, cash or a combination thereof, as specified in the Award Agreement.

ARTICLE 10. PERFORMANCE SHARES

- **10.1 Grant of Performance Shares**. Performance Shares may be granted to Participants in such number, and upon such terms, and at any time and from time to time as shall be determined by the Committee, in its sole discretion. Each grant of Performance Shares shall be evidenced by an Award Agreement.
- **10.2 Value of Performance Shares**. Each Performance Share shall have an initial value equal to the Fair Market Value of a Share on the Grant Date. The Committee shall set performance goals in its discretion that, depending on the extent to which they are met over the specified Performance Period, shall determine the number of Performance Shares that shall be paid to a Participant.
- **10.3 Earning of Performance Shares**. After the applicable Performance Period has ended, the number of Performance Shares earned by the Participant over the Performance Period shall be determined as a function of the extent to which the applicable corresponding performance goals have been achieved. This determination shall be made solely by the Committee.
- **10.4 Form and Timing of Payment of Performance Shares**. At the close of the applicable Performance Period, or as soon as practicable thereafter or otherwise provided for in the Award Agreement, the Company shall pay any earned Performance Shares to the applicable Participant in the form of cash or Shares or a combination thereof, as specified in a Participant's applicable Award Agreement. Any Shares paid to a Participant under this Section 10.4 may be subject to any restrictions deemed appropriate by the Committee.

ARTICLE 11. PERFORMANCE UNITS

- **11.1 Grant of Performance Units**. Subject to the terms and provisions of this Plan, Performance Units may be granted to a Participant in such number, and upon such terms, and at any time and from time to time as shall be determined by the Committee, in its sole discretion. Each grant of Performance Units shall be evidenced by an Award Agreement.
- **11.2 Value of Performance Units**. Each Performance Unit shall have an initial notional value equal to a dollar amount determined by the Committee, in its sole discretion. The Committee shall set performance goals in its discretion that, depending on the extent to which they are met over the specified Performance Period, shall determine the number of Performance Units that shall be settled and paid to the Participant.
- 11.3 Earning of Performance Units. After the applicable Performance Period has ended, the number of Performance Units earned by the Participant over the Performance Period shall be determined as a function of the extent to which the applicable corresponding performance goals have been achieved. This determination shall be made solely by the Committee.

11.4 Form and Timing of Payment of Performance Units. At the close of the applicable Performance Period, or as soon as practicable thereafter or otherwise provided for in the Award Agreement, the Company shall pay any earned Performance Units to the applicable Participant in the form of cash or Shares or a combination thereof, as specified in a Participant's applicable Award Agreement. Any Shares paid to a Participant under this Section 11.4 may be subject to any restrictions deemed appropriate by the Committee.

ARTICLE 12. OTHER STOCK-BASED AWARDS AND CASH-BASED AWARDS

12.1 Grant of Other Stock-Based Awards and Cash-Based Awards

- (a) The Committee may grant Other Stock-Based Awards not otherwise described by the terms of this Plan, including, but not limited to, the grant or offer for sale of unrestricted Shares and the grant of deferred Shares or deferred Share units, in such amounts and subject to such terms and conditions, as the Committee shall determine, in its sole discretion. Such Awards may involve the transfer of actual Shares to Participants, or payment in cash or otherwise of amounts based on the value of Shares.
- (b) The Committee, at any time and from time to time, may grant Cash-Based Awards to a Participant in such amounts and upon such terms as the Committee shall determine, in its sole discretion.

12.2 Value of Other Stock-Based Awards and Cash-Based Awards.

- (a) Each Other Stock-Based Award shall be expressed in terms of Shares or units based on Shares, as determined by the Committee, in its sole discretion.
- (b) Each Cash-Based Award shall specify a payment amount or payment range as determined by the Committee, in its sole discretion. If the Committee exercises its discretion to establish performance goals, the value of Cash-Based Awards paid to the Participant shall depend on the extent to which such performance goals are met.
- 12.3 Payment of Other Stock-Based Awards and Cash-Based Awards. Payment, if any, with respect to Cash-Based Awards and Other Stock-Based Award shall be made in accordance with the terms of the applicable Award Agreement, in cash, Shares or a combination of both as determined by the Committee in its sole discretion.

ARTICLE 13. TRANSFERABILITY OF AWARDS AND SHARES

- **13.1 Transferability of Awards**. Except as provided in Section 13.2, during a Participant's lifetime, Options and SARs shall be exercisable only by the Participant. Awards shall not be transferable other than by will or the laws of descent and distribution or pursuant to a domestic relations order entered into by a court of competent jurisdiction. No Awards shall be subject, in whole or in part, to attachment, execution or levy of any kind. Any purported transfer in violation of this Section 13.1 shall be null and void.
- **13.2 Committee Action**. Notwithstanding Section 13.1, the Committee may, subject to applicable laws, rules and regulations and such terms and conditions as it shall specify, determine that any or all Awards shall be transferable, for no consideration to a Permitted Transferee. Any Award transferred to a Permitted Transferee shall be further transferable only by last will and testament or the laws of descent

and distribution or, for no consideration, to another Permitted Transferee of the Participant. "**Permitted Transferees**" include (a) a Participant's family member, (b) one or more trusts established in whole or in part for the benefit of one or more of such family members, (c) one or more entities which are beneficially owned in whole or in part by the Participant or one or more such family members, or (d) a charitable or not-for-profit organization.

13.3 Restrictions on Share Transferability. The Committee may impose such restrictions on any Shares acquired by a Participant under this Plan as it may deem advisable, including, without limitation, minimum holding period requirements, restrictions under applicable federal securities laws, under the requirements of any stock exchange or market upon which such Shares are then listed or traded or under any blue sky or state securities laws applicable to such Shares.

ARTICLE 14. PERFORMANCE MEASURES

14.1 Performance Measures. Any Award to a Participant may be subject to performance goals as determined at the discretion of the Committee, which may include, but are not limited to, any of the following: (a) book value or earnings per share; (b) cash flow, free cash flow or operating cash flow; (c) earnings before or after any of, or any combination of, interest, taxes, depreciation, or amortization; (d) expenses/costs; (e) gross, net or pre-tax income (aggregate or on a per-Share basis); (f) net income as a percentage of sales; (g) gross or net operating margins or income, including operating income; (h) gross or net sales or revenues; (i) gross profit or gross margin; (j) improvements in capital structure, cost of capital or debt reduction; (k) market share or market share penetration; (l) growth in managed assets; (m) reduction of losses, loss ratios and expense ratios; (n) asset turns, inventory turns or fixed asset turns; (o) operational performance measures; (p) profitability ratios (pre or post tax); (q) profitability of an identifiable business unit or product; (r) return measures (including return on assets, return on equity, return on investment, return on capital, return on invested capital, gross profit return on investment, gross margin return on investment, economic value added or similar metric); (s) Share price (including growth or appreciation in Share price and total shareholder return); (t) strategic business objectives (including objective project milestones); (u) transactions relating to acquisitions or divestitures; or (v) working capital. Any Performance Measure(s) may, as the Committee in its sole discretion deems appropriate, (i) relate to the performance of the Company or any Affiliate as a whole or any business unit or division of the Company or any Affiliate or any combination thereof, (ii) be compared to the performance of a group of comparator companies, or published or special index, (iii) be based on a change in the Performance Measure over a specified period of time and such change may be measured based on an arithmetic change over the specified period (e.g., cumulative change or average change), or percentage change over the specified period (e.g., cumulative percentage change, average percentage change or compounded percentage change), (iv) relate to or be compared to one or more other Performance Measures or (v) any combination of the foregoing. Subject to Section 23.1, the Committee also has the authority to provide for accelerated vesting of any Award based on the achievement of performance goals pursuant to any Performance Measures.

14.2 Evaluation of Performance. The Performance Measures shall, to the extent possible, be determined in accordance with generally accepted accounting principles consistently applied on a business unit, divisional, Subsidiary or consolidated basis or any combination thereof. The Committee may provide in any Award that any evaluation of performance may include or exclude the impact, if any, on reported financial results of any events that occur during a Performance Period including, but not limited to: (a) asset write-downs, (b) litigation or claim judgments or settlements, (c) changes in tax laws, accounting principles or other laws or provisions, (d) reorganization or restructuring programs, (e) acquisitions or divestitures, (f) foreign exchange gains and losses and (g) gains and losses that are treated as unusual or infrequently occurring items within the meaning of the accounting standards of the Financial Accounting Standard Board or such comparable successor term.

14.3 Adjustment of Awards. The Committee shall retain the discretion to adjust any Awards, either on a formula or discretionary basis or any combination, as the Committee determines, in its sole discretion.

ARTICLE 15. TERMINATION OF SERVICES

- **15.1 Termination of Services.** The Committee shall specify in the applicable Award Agreement, at or after the time of grant of an Award the provisions governing the terms of an Award in the event of a Participant's Termination of Services. Subject to applicable laws, rules and regulations, in connection with a Participant's termination, as well as Section 23.1, the Committee shall have the discretion to accelerate the vesting, exercisability or settlement of, eliminate the restrictions and conditions applicable to or extend the post-termination exercise period of an outstanding Award. Such provisions shall be determined by the Committee in its sole discretion and may be specified in the applicable Award Agreement or determined at a subsequent time. The Committee's decisions need not be uniform among all Award Agreements and Participants and may reflect distinctions based on the reasons for termination.
- **15.2 Post-Termination Findings**. If subsequent to the Participant's Termination of Services other than for Cause, it is discovered that the Participant's Services could have been terminated for Cause, the Participant's Services may, at the election of the Committee, be deemed to have been terminated for Cause retroactively to the date the events giving rise to Cause occurred.

ARTICLE 16. NON-EMPLOYEE DIRECTOR AWARDS

- **16.1 Awards to Non-Employee Directors**. The Board or Committee shall determine and approve all Awards to Non-Employee Directors. The terms and conditions of any grant of any Award to a Non-Employee Director shall be set forth in an Award Agreement.
- **16.2 Awards in Lieu of Fees**. The Board or Committee may permit a Non-Employee Director the opportunity to receive an Award in lieu of payment of all or a portion of future director fees (including but not limited to cash retainer fees and meeting fees) or other types of Awards pursuant to such terms and conditions as the Board or Committee may prescribe and set forth in an applicable sub-plan or Award Agreement.

ARTICLE 17. EFFECT OF A CHANGE IN CONTROL

- **17.1 Change in Control**. Subject to Section 23.1, if a Participant has in effect an employment, retention, change in control, severance or similar agreement with the Company or any Affiliate or is subject to a policy or plan that discusses the effect of a Change in Control on a Participant's Awards, then such agreement, plan or policy shall control. In all other cases, unless provided otherwise in an Award Agreement or by the Committee prior to the date of the Change in Control, in the event of a Change in Control the Committee may, but shall not be obligated to:
 - (a) accelerate, vest or cause the restrictions to lapse with respect to all or any portion of any Award;
- (b) cancel any Awards for a cash payment or delivery of other property equal to Fair Market Value (as determined in the sole discretion of the Committee);
- (c) provide for the issuance of substitute Awards that shall substantially preserve the otherwise applicable terms of any affected Awards previously granted hereunder as determined by the Committee in its sole discretion;

- (d) provide that for a period of at least 15 days prior to the Change in Control, Options or SARs shall be exercisable as to all Shares subject thereto and that upon the occurrence of the Change in Control, such Options or SARs shall terminate and be of no further force and effect; or
- (e) provide for the automatic acceleration and vesting with respect to all or any portion of any Award held by a Participant who is involuntarily terminated on or within two years following the applicable Change in Control.

If the value of an Award is based on the Fair Market Value of a Share, Fair Market Value shall be deemed to mean the per share Change in Control price. The Committee shall determine the per-share Change in Control price paid or deemed paid in the Change in Control transaction.

ARTICLE 18. DIVIDENDS AND DIVIDEND EQUIVALENTS

The Committee may provide Participants with the right to receive dividends or payments equivalent to dividends ("**Dividend Equivalents**") or interest with respect to an outstanding Award, which payments can either be paid in cash, either on a current or deferred basis, or deemed to have been reinvested in Shares, or a combination thereof, as the Committee shall determine, in each case, subject to all applicable laws, rules and regulations, including, without limitation, Code Section 409A.

ARTICLE 19. BENEFICIARY DESIGNATION

Each Participant under this Plan may, from time to time, name any beneficiary or beneficiaries (who may be named contingently or successively) to whom any benefit under this Plan is to be paid in case of his death before he receives any or all of such benefit. Each such designation shall revoke all prior designations by the same Participant, shall be in a form prescribed by the Committee and shall be effective only when filed by the Participant in writing with the Company during the Participant's lifetime. In the absence of any such beneficiary designation, benefits remaining unpaid or rights remaining unexercised at the Participant's death shall be paid to or exercised by the Participant's executor, administrator or legal representative.

ARTICLE 20. RIGHTS OF PARTICIPANTS

- **20.1 Employment.** Nothing in this Plan or an Award Agreement shall (a) interfere with or limit in any way the right of the Company or any Affiliate to terminate any Participant's employment with or Service to the Company or any Affiliate at any time or for any reason not prohibited by law or (b) confer upon any Participant any right to continue his employment or Service for any specified period of time. Neither an Award nor any benefits arising under this Plan shall constitute an employment contract or create a statutory right to employment with the Company or any Affiliate and, accordingly, subject to Article 3 and Article 21, this Plan and the benefits hereunder may be amended or terminated at any time in the sole and exclusive discretion of the Board without giving rise to any liability on the part of the Company, any Affiliate, the Committee or the Board.
- **20.2 Participation**. No individual shall have the right to be selected to receive an Award under this Plan, or, having been so selected, to be selected to receive a future Award. The Committee may grant more than one Award to a Participant and may designate an individual as a Participant for overlapping periods of time.
- **20.3 Rights as a Shareholder**. Except as otherwise provided herein, a Participant shall have none of the rights of a shareholder with respect to Shares covered by any Award until the Participant becomes the record holder of such Shares. No adjustment shall be made for dividends or other rights for which the record date is prior to the date on which the Participant becomes the record holder of the Shares.

ARTICLE 21. AMENDMENT AND TERMINATION

- **21.1 Amendment and Termination of this Plan and Awards**. Subject to applicable laws, rules and regulations and Section 21.3 of this Plan, the Board may at any time amend or terminate this Plan or amend or terminate any outstanding Award. Notwithstanding the foregoing, no amendment of this Plan shall be made without shareholder approval if shareholder approval is required pursuant to rules promulgated by any stock exchange or quotation system on which Shares are listed or quoted or by applicable U.S. state corporate laws or regulations, applicable U.S. federal laws or regulations and the applicable laws of any foreign country or jurisdiction where Awards are, or shall be, granted under this Plan.
- **21.2 Awards Previously Granted**. Notwithstanding any other provision of this Plan to the contrary, other than Sections 22.2, 22.4, 23.2 and 23.16, no termination or amendment of this Plan or an Award Agreement shall adversely affect in any material way any Award previously granted under this Plan, without the written consent of the Participant holding such Award.
- **21.3 Amendment to Conform to Law**. Notwithstanding any other provision of this Plan to the contrary, the Committee shall have the broad authority to amend this Plan, an Award or an Award Agreement, to take effect retroactively or otherwise, as deemed necessary or advisable in order to comply with, take into account changes in, or interpretations of, applicable tax laws, securities laws, employment laws, accounting rules and other applicable laws, rules, rulings and regulations promulgated thereunder. By accepting an Award under this Plan, a Participant agrees to any amendment made pursuant to this Section 21.3 to this Plan, an Award or an Award Agreement without further consideration or action.
- 21.4 Repricing of Options and Stock Appreciation Rights. Except in connection with a corporate transaction involving the Company (including, without limitation, any stock dividend, stock split, extraordinary cash dividend, recapitalization, reorganization, merger, consolidation, split-up, spin-off, combination or exchange of Shares), the terms of outstanding Awards may not be amended, without shareholder approval, to reduce the exercise price of outstanding Options or Stock Appreciation Rights, or to cancel outstanding Options or Stock Appreciation Rights with an exercise price that is less than the exercise price of the original Options or Stock Appreciation Rights.

ARTICLE 22. TAX WITHHOLDING

- **22.1 Tax Withholding.** A member of the Company Group may require any individual entitled to receive a payment of an Award to remit to the applicable member of the Company Group prior to payment, an amount sufficient to satisfy any applicable federal, state, local and foreign tax withholding requirements. A member of the Company Group shall also have the right to deduct from all cash payments made to a Participant (whether or not such payment is made in connection with an Award) any applicable taxes required to be withheld with respect to such Award.
- **22.2 Share Withholding.** With respect to withholding required upon the exercise of Options or SARs, upon the lapse of restrictions on Restricted Stock, upon the settlement of Restricted Stock Units or upon the achievement of performance goals related to Performance Shares, or any other taxable event arising as a result of an Award granted hereunder (collectively and individually referred to as a "**Share Payment**"), the Committee may permit or require a Participant to satisfy the withholding requirement, in whole or in part, by having the Company withhold Shares from a Share Payment (or repurchase Shares that were previously issued) having a Fair Market Value on the date the withholding is to be determined equal to the minimum statutory withholding requirement or such other rate as shall not result in any adverse accounting consequences, as determined by the Company in its sole discretion.

ARTICLE 23. GENERAL PROVISIONS

- **23.1 Minimum Vesting.** All Awards shall be subject to a minimum vesting restriction or Performance Period of not less than one year (or, in the case of awards to Non-Employee Directors, the period from one annual meeting of shareholders to the next); provided, however, the requirements set forth in this sentence shall not apply to substituted Awards with time-based vesting restrictions no less than the time-based vesting restrictions of the Awards being replaced and Awards with respect to an aggregate number of Shares not in excess of 5% of the total Shares authorized for issuance under this Plan, as well as Awards accelerated in the event of a Termination of Services or a Change in Control. For purposes of this Section 23.1, a Performance Period shall include any period with respect to which an award is earned.
- **23.2 Forfeiture Events**. The Committee may specify in an Award Agreement that the Participant's rights, payments and benefits with respect to an Award shall be subject to reduction, cancellation, forfeiture or recoupment upon the occurrence of certain specified events as determined by the Committee in its sole discretion.
- **23.3 Legend.** All certificates for Shares delivered under this Plan shall be subject to such stock-transfer orders and other restrictions as the Committee may deem advisable under the rules, regulations and other requirements of the Securities and Exchange Commission, any exchange upon which the Shares are then listed, and any applicable securities law, and the Committee may cause a legend or legends to be put on any such certificates to make appropriate reference to such restrictions.
- 23.4 Data Privacy. As a condition for receiving any Award and to the extent permitted by applicable law, the Committee may require a Participant to explicitly and unambiguously consent to the collection, use and transfer, in electronic or other form, of the Participant's personal data by and among, as applicable, the Company and its Affiliates for the exclusive purpose of implementing, administering and managing the Participant's participation in this Plan. The Company and its Affiliates may hold certain personal information about Participants, including, but not limited to, a Participant's name, address, telephone number, birth date, social security, insurance number or other identification numbers, salary, nationality, job title(s), Shares held in the Company or its affiliates and Award details, to implement, manage and administer this Plan and Awards (the "Data"). The Company and its Affiliates may transfer the Data amongst themselves as necessary to implement, administer and manage a Participant's participation in this Plan, and the Company and its Affiliates may transfer the Data to third parties assisting in implementation, administration and management of the Plan, that these recipients may be located in the Participant's country or elsewhere, and that any recipient's country may have different data privacy laws and protections than the Participant's country. By accepting an Award, the Participant authorizes the recipients to receive, possess, use, retain and transfer Data, in electronic or other form, for the sole purpose of implementing, administering and managing the Participant's participation in this Plan. Furthermore, the Participant acknowledges and understands that the transfer of Data to the Company or to any third parties is necessary for the Participant's participation in this Plan. A Participant may view Data, request information about the storage and processing of Data, request any corrections to Data, or withdraw the consents herein (in any case, without cost to the Participant) by contacting compliance@tpg.com in writing. The withdrawal of any consent by a Participant may affect the Participant's participation in the Plan. The Participant may contact compliance@tpg.com for further information about the consequences of any withdrawal of consents herein.
- **23.5 Gender and Number**. Except where otherwise indicated by the context, any masculine or feminine term used herein also shall include all applicable genders, the plural shall include the singular, and the singular shall include the plural.

- **23.6 Severability**. In the event any provision of this Plan shall be held illegal or invalid for any reason, the illegality or invalidity shall not affect the remaining parts of this Plan, and this Plan shall be construed and enforced as if the illegal or invalid provision had not been included.
- **23.7 Requirements of Law**. The granting of Awards and the issuance of Shares under this Plan shall be subject to all applicable laws, rules and regulations, and to such approvals by any governmental agencies or national securities exchanges as may be required.
 - **23.8 Delivery of Title.** The Company shall have no obligation to issue or deliver evidence of title for Shares issued under this Plan prior to:
 - (a) Obtaining any approvals from governmental agencies that the Company determines are necessary or advisable; and
- (b) Completion of any registration or other qualification of the Shares under any applicable national, state or foreign law or ruling of any governmental body that the Company determines to be necessary or advisable.
- **23.9 Inability to Obtain Authority**. The inability of the Company to obtain authority from any regulatory body having jurisdiction, which authority is deemed by the Company's counsel to be necessary to the lawful issuance and sale of any Shares hereunder, shall relieve the Company of any liability in respect of the failure to issue or sell such Shares as to which such requisite authority shall not have been obtained.
- **23.10 Investment Representations**. The Committee may require any individual receiving Shares pursuant to an Award under this Plan to represent and warrant in writing that the individual is acquiring the Shares for investment and without any present intention to sell or distribute such Shares.
- **23.11 TPG Personnel Based Outside of the United States**. Notwithstanding any provision of this Plan to the contrary, subject to Section 23.1, in order to comply with the laws in other countries in which the Company or any Affiliates operate or have a TPG Person, the Committee, in its sole discretion, shall have the power and authority to:
 - (a) Determine which Affiliates shall be covered by this Plan;
 - (b) Determine which a TPG Person outside the United States is eligible to participate in this Plan;
- (c) Modify the terms and conditions of any Award granted to a TPG Person outside the United States to comply with applicable foreign laws;
- (d) Establish sub-plans and modify exercise procedures and other terms and procedures, to the extent such actions may be necessary or advisable. Any sub-plans and modifications to Plan terms and procedures established under this Section 23.11 by the Committee shall be attached to this Plan document as appendices; and
- (e) Take any action, before or after an Award is made, that it deems advisable to obtain approval or comply with any necessary local government regulatory exemptions or approvals.

Any sub-plans and special provisions may take precedence over other provisions of this Plan, but unless otherwise superseded by the terms of such sub-plans and special provisions, the provisions of this Plan shall govern. Notwithstanding the above, the Committee may not take any actions hereunder, and no Awards shall be granted, that would violate applicable law.

- **23.12 Uncertificated Shares**. To the extent that this Plan provides for the issuance of certificates to reflect the transfer of Shares, the transfer of such Shares may be affected on a noncertificated basis, to the extent not prohibited by applicable law or the rules of any stock exchange.
- 23.13 Unfunded Plan. Participants shall have no right, title or interest whatsoever in or to any investments that the Company or any Affiliates may make to aid it in meeting its obligations under this Plan. Nothing contained in this Plan, and no action taken pursuant to its provisions, shall create or be construed to create a trust of any kind, or a fiduciary relationship between the Company and any Participant, beneficiary, legal representative or any other individual. To the extent that any individual acquires a right to receive payments from the Company or any Affiliate under this Plan, such right shall be no greater than the right of an unsecured general creditor of the Company or any Affiliate, as the case may be. All payments to be made hereunder shall be paid from the general funds of the Company, or any Affiliate, as the case may be, and no special or separate fund shall be established, and no segregation of assets shall be made to assure payment of such amounts except as expressly set forth in this Plan.
- **23.14 No Fractional Shares**. No fractional Shares shall be issued or delivered pursuant to this Plan or any Award. The Committee shall determine whether cash, Awards or other property shall be issued or paid in lieu of fractional Shares or whether such fractional Shares or any rights thereto shall be forfeited or otherwise eliminated.
- **23.15 Retirement and Welfare Plans**. Neither Awards made under this Plan nor Shares or cash paid pursuant to such Awards may be included as "compensation" for purposes of computing the benefits payable to any Participant under the Company's or any Affiliate's retirement plans (both qualified and nonqualified) or welfare benefit plans unless such other plan expressly provides that such compensation shall be taken into account in computing a Participant's benefit.

23.16 Deferrals.

- (a) Notwithstanding any contrary provision in this Plan or an Award Agreement, if any provision of this Plan or an Award Agreement contravenes any regulations or guidance promulgated under Code Section 409A or would cause an Award to be subject to additional taxes, accelerated taxation, interest or penalties under Code Section 409A, such provision of this Plan or Award Agreement may be modified by the Committee without consent of the Participant in any manner the Committee deems reasonable or necessary. In making such modifications, the Committee shall attempt, but shall not be obligated, to maintain, to the maximum extent practicable, the original intent of the applicable provision without contravening the provisions of Code Section 409A. Moreover, any discretionary authority that the Committee may have pursuant to this Plan shall not be applicable to an Award that is subject to Code Section 409A to the extent such discretionary authority would contravene Code Section 409A or the guidance promulgated thereunder.
- (b) If a Participant is a "specified employee" as defined under Code Section 409A and the Participant's Award is to be settled on account of the Participant's separation from service (for reasons other than death) and such Award constitutes "deferred compensation" as defined under Code Section 409A, then any portion of the Participant's Award that would otherwise be settled during the six-month period commencing on the Participant's separation from service shall be settled as soon as practicable following the conclusion of the six-month period (or following the Participant's death if it occurs during such six-month period).
- (c) In accordance with the procedures authorized by, and subject to the approval of, the Committee, Participants may be given the opportunity to defer the payment or settlement of an Award to one or more dates selected by the Participant; <u>provided</u>, <u>however</u>, that the terms of any deferrals must comply with all applicable laws, rules and regulations, including, without limitation, Code Section 409A. No deferral opportunity shall exist with respect to an Award unless explicitly permitted by the Committee on or after the time of grant.

- **23.17 Nonexclusivity of this Plan**. The adoption of this Plan shall not be construed as creating any limitations on the power of the Board or Committee to adopt such other compensation arrangements as it may deem desirable for any Participant.
- **23.18 No Constraint on Corporate Action**. Nothing in this Plan shall be construed to: (a) limit, impair or otherwise affect the Company's or an Affiliate's right or power to make adjustments, reclassifications, reorganizations or changes of its capital or business structure, or to merge or consolidate, or dissolve, liquidate, sell or transfer all or any part of its business or assets or (b) limit the right or power of the Company or any Affiliate to take any action that such entity deems to be necessary or appropriate. The proceeds received by the Company from the sale of Shares pursuant to Awards shall be used for general corporate purposes.
- **23.19 Conflicts**. Except as otherwise provided herein, in the event of any conflict or inconsistency between this Plan and any Award Agreement, this Plan shall govern, and the Award Agreement shall be interpreted to minimize or eliminate any such inconsistency.
- **23.20 Recoupment**. Notwithstanding anything in this Plan to the contrary, all Awards granted under this Plan and any payments made under this Plan shall be subject to clawback or recoupment as permitted or mandated by applicable law, rules, regulations or Company policy as enacted, adopted or modified from time to time. For the avoidance of doubt, this provision shall apply to any gains realized upon exercise or settlement of an Award and Dividend Equivalents paid on unvested Awards.
- **23.21 Delivery and Execution of Electronic Documents**. To the extent permitted by applicable law, the Company may (a) deliver by email or other electronic means (including posting on a website maintained by the Company or by a third party under contract with the Company) all documents relating to this Plan or any Award thereunder (including without limitation, prospectuses and other securities requirements) and all other documents that the Company is required to deliver to its security holders (including without limitation, annual reports and proxy statements) and (b) permit Participants to electronically execute applicable Plan documents (including, but not limited to, Award Agreements) in a manner prescribed to the Committee.
- **23.22 No Representations or Warranties Regarding Tax Effect**. Notwithstanding any provision of this Plan to the contrary, the Company, Affiliates, the Board and the Committee neither represent nor warrant the tax treatment under any federal, state, local or foreign laws and regulations thereunder (individually and collectively referred to as the "**Tax Laws**") of any Award granted or any amounts paid to any Participant under this Plan including, the Tax Laws.
- 23.23 Indemnification. Subject to applicable laws, rules and regulations and the Company's Certificate of Incorporation as it may be amended from time to time, each individual who is or shall have been a member of the Board, or a Committee appointed by the Board, or an officer of the Company to whom authority was delegated in accordance with Article 3, shall be indemnified and held harmless by the Company against and from (a) any loss, cost, liability or expense that may be imposed upon or reasonably incurred by him or her in connection with or resulting from any claim, action, suit or proceeding to which he or she may be a party or in which he or she may be involved by reason of any good faith action taken or failure to act under this Plan and (b) any and all amounts paid by him or her in settlement thereof, with the Company's approval or paid by him or her in satisfaction of any judgment in any such action, suit or proceeding against him or her; provided that he or she shall give the Company an opportunity, at its own expense, to handle and defend the same before he or she undertakes to handle and defend it on his or her own behalf. Notwithstanding the foregoing, no individual shall be entitled to indemnification if such loss,

cost, liability or expense is a result of his/her own willful misconduct. The foregoing right of indemnification shall not be exclusive of any other rights of indemnification to which such individuals may be entitled under the Company's Articles of Incorporation or By-laws, as a matter of law or otherwise, or any power that the Company may have to indemnify them or hold them harmless.

23.24 Successors. Subject to Article 17, all obligations of the Company under this Plan with respect to Awards granted hereunder shall be binding on any successor to the Company (each, a "**Successor**"), whether the existence of such successor is the result of a direct or indirect purchase, merger, consolidation or otherwise, of all or substantially all of the business or assets of the Company.

23.25 Governing Law. This Plan and each Award Agreement shall be governed by the laws of the State of Delaware excluding any conflicts or choice of law rule or principle that might otherwise refer construction or interpretation of this Plan to the substantive law of another jurisdiction.

23.26 Arbitration.

- (a) Any dispute, controversy or claim arising out of or relating to this Plan or with respect to an Award granted hereunder, including, without limitation, any dispute regarding the validity or termination of this Plan or an Award granted hereunder, or the performance or breach hereof or of the terms of any Award Agreement, shall be settled exclusively by arbitration administered by the American Arbitration Association (the "AAA"). The place of arbitration shall be Fort Worth, Texas, and the proceedings shall be conducted in the English language. The arbitration shall be conducted in accordance with the AAA rules governing commercial arbitration in effect at the time of the arbitration, except as modified herein. There shall be three arbitrators, one of whom shall be nominated by the Company or its designee and one who shall be nominated by the Participant within 30 days of receipt by respondent of the demand for arbitration, and the third arbitrator, who shall chair the arbitral tribunal, shall be nominated by the party nominated arbitrators within 30 days of the nomination of the second arbitrator. If any arbitrator is not appointed within the time limit provided herein, upon request of any party to the arbitration, such arbitrator shall be appointed by the AAA within 15 days of receiving such request.
- (b) The arbitration shall commence within 45 days after the appointment of the third arbitrator; the arbitration shall be completed within 60 days of commencement; and the arbitrators' award shall be made within 30 days following such completion. The parties may agree to extend the time limits specified in the foregoing sentence.
- (c) The arbitral tribunal may award any form of relief permitted under this Plan, the relevant Award Agreement, or applicable law, including damages and temporary or permanent injunctive relief, except that the arbitral tribunal is not empowered to award damages in excess of compensatory damages, and each party hereby irrevocably waives any right to recover punitive, exemplary or similar damages with respect to any dispute. The award shall be in writing and shall state the reasons for the award.
- (d) The decision rendered by the arbitral tribunal shall be final and binding on the parties to the relevant Award Agreement. Judgment may be entered in any court of competent jurisdiction. By accepting an Award, the parties to the relevant Award Agreement waive, to the fullest extent permitted by law, any rights to appeal to, or to seek review of such award by, any court. By accepting an Award, the parties to the relevant Award Agreement agree to obtain the arbitral tribunal's agreement to preserve the confidentiality of the arbitration.
- (e) Without limiting the foregoing, each party to an Award Agreement agrees that service of process on such party shall be deemed effective service of process on such party.